SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported)			June 30, 2011		
		PACIFIC ETHAN	NOL, INC.		
	(Exact name of registrant as sp		·)	
	Delaware	000-2146	67	41-2170618	
	(State or other jurisdiction of incorporation)	(Commission File	Number)	(IRS Employer Identification No.)	
	400 Capitol Mall, Suite 2060, Sad	cramento, CA		95814	
	(Address of principal executiv	re offices)		(Zip Code)	
Registra	nt's telephone number, including area o	ode:		(916) 403-2123	
	(Form	er name or former address, if	changed since last i	report)	
	ck the appropriate box below if the Fo llowing provisions (<i>see</i> General Instru		imultaneously satisf	y the filing obligation of the registrant under any	
	Written communications pursuant to	Rule 425 under the Securities	Act (17 CFR 230.4	425)	
	Soliciting material pursuant to Rule	4a-12 under the Exchange A	ct (17 CFR 240.14a	-12)	
	Pre-commencement communications	pursuant to Rule 14d-2(b) ur	nder the Exchange A	act (17 CFR 240.14d-2(b))	
	Pre-commencement communications	pursuant to Rule 13e-4(c) un	der the Exchange A	ct (17 CFR 240.13e-4(c))	

Item 1.01. Entry into a Material Definitive Agreement.

On June 29, 2010, Pacific Ethanol Holding Co. LLC ("PEHC"), Pacific Ethanol Madera LLC ("Madera"), Pacific Ethanol Columbia, LLC ("Boardman"), Pacific Ethanol Stockton, LLC ("Stockton") and Pacific Ethanol Magic Valley, LLC ("Burley" and, together with Madera, Boardman and Stockton, each a "Plant Owner" and, collectively, the "Plant Owners"), previously wholly owned subsidiaries of Pacific Ethanol, Inc. (the "Company"), emerged from Chapter 11 pursuant to the terms of the Amended Joint Plan of Reorganization (the "Plan"), which was filed with the Bankruptcy Court on April 16, 2010 and confirmed by the Bankruptcy Court on June 8, 2010. Pursuant to the terms of the Plan, PEHC and the Plant Owners entered into a credit agreement on June 25, 2010 with WestLB, AG, New York Branch and certain other lenders (as amended to date, the "Credit Agreement"). Under the terms of the Plan, 100% of the ownership interest in PEHC was transferred to a newlyformed limited liability company ("New PE Holdco"), resulting in PEHC and the Plant Owners becoming direct and indirect wholly-owned subsidiaries of New PE Holdco. The Company holds a 20% ownership interest in New PE Holdco.

Asset Management Agreement

Each of the Plant Owners owns a separate ethanol production facility. Under the terms of the Credit Agreement, the Plant Owners engaged the Company to provide certain management services to the Plant Owners. On June 29, 2010, the Company entered into an Asset Management Agreement with PEHC and the Plant Owners (as amended to date, the "Previous Asset Management Agreement"). On June 30, 2011, the Company entered into an Amended and Restated Asset Management Agreement with PEHC and the Plant Owners (the "Asset Management Agreement") that replaces the Previous Asset Management Agreement in its entirety.

Pursuant to the Asset Management Agreement, the Company will provide certain management services to the Plant Owners whereby the Company will effectively continue to operate and maintain the production facilities on behalf of the Plant Owners. These services generally include, but are not limited to, administering, at the direction of PEHC, each Plant Owner's compliance with the Credit Agreement and related financing documents and performing billing, collection, record keeping and other administrative and ministerial responsibilities for each Plant Owner. The Company will supply all labor and personnel required to perform its services under the agreement, including, but not limited to, the labor and personnel required to operate and maintain the production facilities.

The Boardman, Burley and Stockton production facilities are currently operational. The Madera facility is currently idled. If market conditions continue to improve, the Company may resume operations at the Madera facility, subject to the approval of New PE Holdco. For production facilities that are currently or in the future operational, the Company will provide substantially all services necessary for the operation and maintenance of the facilities. For production facilities that are currently or in the future idled, the Company is responsible for the preservation and maintenance of the facilities to facilitate a cost effective return to operational status once determined appropriate by the facility's Plant Owner. PEHC or the applicable Plant Owner may elect to recommence operations of the idled facility or place an operational facility into idle status by giving the Company at least 60 days notice. Notwithstanding the services provided by the Company, each Plant Owner retains the ultimate decision-making authority with all matters concerning the production facility owned by it.

The costs and expenses associated with the Company's provision of services under the Asset Management Agreement will be funded by the Plant Owners pursuant to a preapproved budget. The Company's obligation to provide services is limited to the extent there are insufficient funds advanced by the Plant Owners to cover the associated costs and expenses.

As compensation for providing the services, each Plant Owner will pay the Company \$75,000 per month if its production facility is operational and \$40,000 per month if its production facility is idled. In addition to the monthly fee, if during any six-month period (measured biannually on March 31 and September 30 of each year) a production facility has annualized EBITDA per gallon of operating capacity of \$0.20 or more, the Plant Owner of that facility will pay the Company a performance bonus equal to 3% of the increment by which EBITDA exceeds \$0.20. The aggregate performance bonus collectively payable by all Plant Owners is capped at \$2.2 million for each six-month period. The performance bonus will be reduced by 25% if all production facilities then operating do not operate at a minimum average yield of 2.70 gallons of denatured ethanol per bushel of corn. In addition, no performance bonus will be paid if there is a default or event of default under the Credit Agreement resulting from the borrowers' failure pay any amounts then due and owing.

Upon a sale of all or substantially all of the assets of a facility, or all of the equity interests in a Plant Owner, to a third party, the Company will receive an incentive fee with respect to such sale as follows:

"Sale Price per Gallon"	Incentive Fee		
\$.60 or less	\$0		
Above \$.60 up to and including \$.70	The excess of Sale Price Per Gallon over \$.60, to and including the lesser of the Sale Price Per Gallon and \$.70, multiplied by the operating capacity of the facility (in gallons), multiplied by 0.5%; <i>plus</i>		
Above \$.70 up to and including \$.80	If the Sale Price Per Gallon exceeds \$.70, the excess of the Sale Price Per Gallon over \$.70, to and including the lesser of the Sale Price Per Gallon and \$.80, multiplied by the operating capacity of the facility (in gallons), multiplied by 1.0%; <i>plus</i>		
Above \$.80	If the Sale Price Per Gallon exceeds \$.80, the excess of the Sale Price Per Gallon over \$.80, multiplied by the operating capacity of the facility (in gallons), multiplied by 1.5%.		

The Asset Management Agreement becomes effective on June 30, 2011 and shall terminate on June 30, 2012, provided that the Company or PEHC can extend the Asset Management Agreement for additional one year periods by giving the other party not less than 90 days notice unless the other party rejects such renewal notice within 15 days of receiving the notice. In addition to typical conditions for a party to terminate the agreement prior to its expiration, the Company may terminate the agreement, and any Plant Owner may terminate the agreement with respect to its facility, at any time by providing at least 60 days prior notice of such termination. The agreement also contains other terms and provisions customary for an agreement of this type, including, but not limited to, provisions regarding insurance requirements, record retention, confidentiality, indemnification and force majeure.

The description of the Asset Management Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the Asset Management Agreement, which is filed as Exhibit 10.1 to this Form 8-K and incorporated herein by this reference.

Ethanol Marketing Agreements

As contemplated by the Plan, Kinergy Marketing, LLC, a wholly owned subsidiary of the Company ("Kinergy"), entered into separate Ethanol Marketing Agreements with each of Boardman and Burley and Stockton, which grant Kinergy the exclusive right to market, purchase and sell the ethanol produced at the Plant Owner's facility (each a "Previous Ethanol Marketing Agreement"). On June 30, 2011, Kinergy entered into separate Amended and Restated Ethanol Marketing Agreements with each of Boardman and Burley and Stockton that replace the Previous Ethanol Marketing Agreements in their entirety (each a "Ethanol Marketing Agreement"). If the production facility owned by Madera becomes operational, it is contemplated that Kinergy would enter into a substantially identical Ethanol Marketing Agreement with Madera.

Pursuant to the terms of the Ethanol Marketing Agreements, within ten days after a Plant Owner delivering ethanol to Kinergy, that Plant Owner is paid an amount equal to (i) the estimated purchase price payable by the third-party purchaser of the ethanol, minus (ii) the estimated amount of transportation costs to be incurred by Kinergy, minus (iii) the estimated incentive fee payable to Kinergy, which equals 1% of the aggregate third-party purchase price provided that the incentive fee shall not be less than \$0.015 per gallon or more than \$0.0225 per gallon. Within the first five business days of each calendar month, the parties will reconcile and "true up" the actual purchase price, transportation costs and incentive fees for all transactions entered into since the previous true-up date.

To facilitate Kinergy's ability to pay amounts owing to the Plant Owners, the Ethanol Marketing Agreements require that Kinergy maintain lines of credit of at least \$5.0 million. Upon the request of a Plant Owner, the Company will enter into a guaranty with the Plant Owner guaranteeing the performance of Kinergy's obligations under the Ethanol Marketing Agreement.

Due to the limited storage capacity of the Plant Owners, Kinergy agrees to take delivery of any ethanol requested by Kinergy within seven days of a Plant Owner making the ethanol available to Kinergy. Each Ethanol Marketing Agreement shall terminate on June 30, 2012, provided that Kinergy or the Plant Owner that is party to the agreement can extend a Ethanol Marketing Agreement for additional one year periods by giving the other party not less than 90 days notice unless the other party rejects such renewal notice within 15 days of receiving the notice. The agreements also contain other terms and provisions customary for agreements of this type, including, but not limited to, provisions regarding termination, insurance requirements, confidentiality, indemnification and force majeure.

The description of the Ethanol Marketing Agreements does not purport to be complete and is qualified in its entirety by reference to the full text of the form of Ethanol Marketing Agreement, which is filed as Exhibit 10.2 to this Form 8-K and incorporated herein by this reference.

Corn Procurement and Handling Agreements

As contemplated by the Plan, Pacific Ag. Products, LLC, an indirect wholly owned subsidiary of the Company ("Pacific Ag"), has entered into separate Corn Procurement and Handling Agreements with each of Boardman, Burley and Stockton (each a "Previous Corn Procurement and Handling Agreement"). On June 30, 2011, Pacific Ag entered into separate Amended and Restated Corn Procurement and Handling Agreements with each of Boardman and Burley and Stockton that replace the Previous Corn Procurement and Handling Agreements in their entirety (each a "Corn Procurement and Handling Agreement"). If the production facility owned by Madera becomes operational, it is contemplated that Pacific Ag would enter into a substantially identical Corn Procurement and Handling Agreement with Madera.

Pursuant to the terms of the Corn Procurement and Handling Agreements, each Plant Owner appoints Pacific Ag as its agent to solicit, negotiate, enter into and administer, on behalf of the Plant Owner, corn supply arrangements for the Plant Owner to procure the corn necessary to operate its facility. Pacific Ag will also provide each Plant Owner grain handling services including, but not limited to, receiving, unloading and conveying corn into the Plant Owner's storage facilitates and, in the case of whole corn delivered to the Plant Owner, processing and hammering the whole corn. Pacific Ag receives a fee of \$0.045 per bushel of corn delivered to each Plant Owner as consideration for its procurement and grain handling services, each payable monthly.

Upon the request of a Plant Owner, the Company will enter into a guaranty with the Plant Owner guaranteeing the performance of Pacific Ag's obligations under the Corn Procurement and Handling Agreement. Each Corn Procurement and Handling Agreement shall terminate on June 30, 2012, provided that Pacific Ag or the Plant Owner that is party to the agreement can extend a Corn Procurement and Handling Agreement for additional one year periods by giving the other party not less than 90 days notice unless the other party rejects such renewal notice within 15 days of receiving the notice. The agreements also contain other terms and provisions customary for agreements of this type, including, but not limited to, provisions regarding termination, insurance requirements, confidentiality, indemnification and force majeure.

The description of the Corn Procurement and Handling Agreements does not purport to be complete and is qualified in its entirety by reference to the full text of the form of Corn Procurement and Handling Agreement, which is filed as Exhibit 10.3 to this Form 8-K and incorporated herein by this reference.

Distillers Grains Marketing Agreements

Pursuant to the Plan, Pacific Ag entered into separate Distillers Grains Marketing Agreements with each of Boardman, Burley and Stockton (each a "Previous Distillers Grains Marketing Agreement"), which grant Pacific Ag the right to market, purchase and sell the distillers grain products, including dried distillers grains and wet distillers grains, produced at the Plant Owner's facility. On June 30, 2011, Pacific Ag entered into separate Amended and Restated Distillers Grains Marketing Agreements with each of Boardman and Burley and Stockton that replace the Previous Distillers Grains Marketing Agreements in their entirety (each a "Distillers Grains Marketing Agreement"). If the production facility owned by Madera becomes operational, it is contemplated that Pacific Ag would enter into a substantially identical Distillers Grains Marketing Agreement with Madera.

Pursuant to the terms of the Distillers Grains Marketing Agreements, within ten days after a Plant Owner delivers distillers grain product to Pacific Ag, that Plant Owner is paid an amount equal to (i) the estimated purchase price payable by the third-party purchaser of the distillers grain product, minus (ii) the estimated amount of transportation costs to be incurred by Pacific Ag, minus (iii) the estimated amount of fees and taxes payable to governmental authorities in connection with the tonnage of distillers grain product or corn condensed distiller's soluble produced or marketed, minus (iv) the estimated incentive fee payable to Pacific Ag, which equals 5% of the aggregate third-party purchase price provided that the incentive fee shall not be less than \$0.015 per ton or more than \$0.0225 per ton of distillers grain product sold in the transaction. Within the first five business days of each calendar month, the parties will reconcile and "true up" the actual purchase price, transportation costs, governmental fees and taxes, and incentive fees for all transactions entered into since the previous true-up date.

Upon the request of a Plant Owner, the Company will enter into a guaranty with the Plant Owner guaranteeing the performance of Pacific Ag's obligations under the Distillers Grains Marketing Agreement. Due to the limited storage capacity of the Plant Owners, Pacific Ag agrees to take delivery of any distillers grain product requested by Pacific Ag within two days of a Plant Owner making the product available to Pacific Ag. Each Distillers Grain Marketing Agreement shall terminate on June 30, 2012, provided that Pacific Ag or the Plant Owner that is party to the agreement can extend a Distillers Grains Marketing Agreement for additional one year periods by giving the other party not less than 90 days notice unless the other party rejects such renewal notice within 15 days of receiving the notice. The agreements also contain other terms and provisions customary for agreements of this type, including, but not limited to, provisions regarding termination, insurance requirements, confidentiality, indemnification and force majeure.

The description of the Distillers Grain Marketing Agreements does not purport to be complete and is qualified in its entirety by reference to the full text of the form of Distillers Grains Marketing Agreement, which is filed as Exhibit 10.4 to this Form 8-K and incorporated herein by this reference.

Item 9.01. Financial Statements and Exhibits.

	Exhibit No.	<u>Description</u>
	10.1	Amended and Restated Asset Management Agreement (*)
	10.2	Form of Amended and Restated Ethanol Marketing Agreement (*)
	10.3	Form of Amended and Restated Corn Procurement and Handling Agreement (*)
	10.4	Form of Amended and Restated Distillers Grains Marketing Agreement (*)
*	Filed herewith	

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: July 6, 2011 PACIFIC ETHANOL, INC.

By: /S/ CHRISTOPHER W. WRIGHT

Christopher W. Wright,

Vice President, General Counsel & Secretary

EXHIBITS FILED WITH THIS REPORT

Exhibit No.	<u>Description</u>
10.1	Amended and Restated Asset Management Agreement
10.2	Form of Amended and Restated Ethanol Marketing Agreement
10.3	Form of Amended and Restated Corn Procurement and Handling Agreement
10.4	Form of Amended and Restated Distillers Grains Marketing Agreement

SECOND AMENDED AND RESTATED ASSET MANAGEMENT AGREEMENT

by and between

PACIFIC ETHANOL, INC.,

and

PACIFIC ETHANOL HOLDING CO. LLC, PACIFIC ETHANOL MADERA LLC, PACIFIC ETHANOL COLUMBIA, LLC, PACIFIC ETHANOL STOCKTON, LLC and PACIFIC ETHANOL MAGIC VALLEY, LLC

Dated as of June 30, 2011

TABLE OF CONTENTS

Article I	DEFINITIONS	1
1.1	Definitions	1
1.2	Interpretation	8
Article II	APPOINTMENT	9
Article III	RESPONSIBILITIES OF MANAGER	9
3.1	Scope of Asset Management Services	9
3.2	Scope of Asset Preservation Services	11
3.3	Scope of O&M Services	12
3.4	Subcontracts	12
3.5	Standards for Performance of the Asset Management Services	13
3.6	Standards for Performance of the Asset Preservation Services and the O&M Services	13
3.7	Personnel Standards.	13
3.8	Training	13
3.9	Additional Obligations of Manager	13
3.10	Licenses and Permits	14
3.11	Records and Reports; Other Material Information	14
3.12	No Liens or Encumbrances	14
3.13	Emergency Action	14
3.14	Scope; Manager Authority	14
3.15	Retention of Control by Owners	15
3.16	Limitations on Performance	15
3.17	Deficiency of Funds	15
Article IV	ITEMS TO BE FURNISHED BY OWNERS	15
4.1	General	15
4.2	Information	15
4.3	Corn, Fuel and Other Materials	15
4.4	Cold Shutdown or Start Up of a Facility	15
Article V	DEVELOPMENT PROJECTS; FACILITY SERVICES	16
5.1	Development Projects	16
Article VI	REPORTING AND PERSONNEL	16
6.1	Accounts and Reports	16
6.2	Budget	17
6.3	Manager Representative	17
Article VII	LIMITATIONS ON AUTHORITY	18
Article VIII	COMPENSATION	19
8.1	Asset Management Fee	19
8.2	Sale Incentive Fee	19

	8.3	Lien Waivers	20
	8.4	Netting	20
	8.5	Intentionally Left Blank	20
Article		TERM	20
	9.1	Term	20
	9.2	Owner Defaults and Manager Remedies	21
	9.3	Manager Defaults and Owner Remedies	21
	9.4	Termination for Convenience	22
	9.5	Facility Condition at End of Term; Successor Manager.	22
	9.6	Termination Payment; Manager Payment.	23
Article		INSURANCE	23
	10.1	Manager Insurance	23
	10.1	Owners Insurance	25
	10.2		25
		Manager Insurance Premiums and Deductibles	
	10.4	Subcontractor Insurance	25
Article		INDEMNIFICATION	25
	11.1	Owners' Indemnity	25
	11.2	Manager's Indemnity	25
Article XII	e	LIABILITIES OF THE PARTIES	26
	12.1	Maximum Liability of Manager	26
	12.2	No Consequential or Punitive Damages	26
Article XIII	e	CONFIDENTIALITY	26
	13.1	Owner's Confidential Information	26
	13.2	Manager's Confidential Information	27
Article XIV	e	TITLE, DOCUMENTS AND DATA	27
	14.1	Materials and Equipment	27
	14.2	Documents	27
Article XV	e	FORCE MAJEURE	28
	15.1	Events Constituting Force Majeure	28
	15.2	Effect	28
	15.3	Limitations	28
Article XVI	e	MISCELLANEOUS PROVISIONS	29
	16.1	Assignment	29
	16.2	Sale of Facilities	29
	16.3	Cooperation in Financing	29
	16.4	Access	29
	16.5	Not for Benefit of Third Parties	30
	16.6 16.7	Amendments	30 30
	16.7	Survival No Waiver	30
	10.0	TO WAIVE	30

	16.9	Notices	30
	16.10	Representations and Warranties.	31
	16.11	Counterparts and Execution	32
	16.12	Governing Law	32
	16.13	Entire Agreement	32
	16.14	Severability	33
	16.15	Successors and Assigns	33
	16.16	Owner Agent	33
	16.17	Independent Contractor	33
	16.18	Captions; Appendices	33
	16.19	No Novation	33
	16.20	Several Liability	33

Appendices

Appendix A-1 Scope of Asset Management Services and Asset Preservation Services

Appendix A-2 Scope of O&M Services

Appendix B Personnel Listing

Appendix C Form of Report (Cold Shutdown)
Appendix D Form of Report (Operating)

Appendix E Budget

Exhibits

Exhibit I Example Performance Bonus Calculation Exhibit II Example Sale Incentive Fee Calculation

Exhibit III Example EBITDA Calculation

This **SECOND AMENDED AND RESTATED ASSET MANAGEMENT AGREEMENT** (this "<u>Agreement</u>") is made and entered into as of June 30, 2011 by and between Pacific Ethanol Holding Co. LLC, a Delaware limited liability company ("<u>Pacific Holding</u>"), Pacific Ethanol Madera LLC, a Delaware limited liability company ("<u>Madera</u>"), Pacific Ethanol Columbia, LLC, a Delaware limited liability company ("<u>Boardman</u>"), Pacific Ethanol Stockton, LLC, a Delaware limited liability company ("<u>Stockton</u>"), and Pacific Ethanol Magic Valley, LLC, a Delaware limited liability company ("<u>Burley</u>" and, together with Madera, Boardman and Stockton, an "<u>Owner</u>" and collectively "<u>Owners</u>"), Pacific Holding as Owner Agent ("<u>Owner Agent</u>") and Pacific Ethanol, Inc., a Delaware corporation ("<u>Manager</u>"). Owners and Manager are each individually referred to herein as a "<u>Party</u>" and collectively are referred to herein as the "<u>Parties</u>".

RECITALS

WHEREAS, the Parties hereto were previously party to that certain Amended and Restated Asset Management Agreement, dated as of October 15, 2010 (the "<u>Prior Agreement</u>") pursuant to which Manager provided various services to the Owners;

WHEREAS, Owners have entered into the Credit Agreement (as defined below) to provide funds for the working capital requirements of Owners and for other purposes permitted under the Credit Agreement;

WHEREAS, as a condition precedent to the obligation of the lenders party to the Credit Agreement to make loans or other extensions of credit to Owners, Owners are required to engage Manager to provide certain management services to Owners; and

WHEREAS, Manager desires to provide such management services.

WHEREAS, the parties to the Prior Agreement desire to amend and restate the Prior Agreement in its entirety upon the terms and conditions set forth herein.

NOW, THEREFORE, for good and adequate consideration, the receipt and adequacy of which are hereby acknowledged, the parties hereto, intending to be legally bound, do hereby agree as follows:

ARTICLE I

DEFINITIONS

1.1 <u>Definitions.</u> Capitalized terms used in this Agreement but not defined herein shall have the meanings assigned thereto in the Credit Agreement. The following terms shall have the meanings set forth below when used in this Agreement:

"Act of Insolvency" means, with respect to any Person, any of the following: (a) commencement by such Person of a voluntary proceeding under any jurisdiction's bankruptcy, insolvency or reorganization law; (b) the filing of an involuntary proceeding against such Person under any jurisdiction's bankruptcy, insolvency or reorganization law which is not vacated within 60 days after such filing; (c) the admission by such Person of the material allegations of any petition filed against it in any proceeding under any jurisdiction's bankruptcy, insolvency or reorganization law; (d) the adjudication of such Person as bankrupt or insolvent or the winding up or dissolution of such Person; (e) the making by such Person of a general assignment for the benefit of its creditors (assignments for a solvent financing excluded); (f) the appointment of a receiver or an administrator for all or a substantial portion of such Person's assets, which receiver or administrator, if appointed without the consent of such Person, is not discharged within 60 days after its appointment; or (g) the occurrence of any event analogous to any of the foregoing with respect to such Person occurring in any jurisdiction.

"Affiliate" of a specified Person means any corporation, partnership, sole proprietorship or other Person which directly or indirectly through one or more intermediaries controls, is controlled by or is under common control with the Person specified. The term "control" means the ownership, either direct or indirect, of twenty-five percent (25%) or more of the voting securities (or comparable equity interests) or other ownership interests of a Person, or the possession, either direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or any other means whatsoever.

"Agreement" means this Second Amended and Restated Asset Management Agreement, including all appendices hereto, as the same may be modified, supplemented or amended from time to time in accordance with the provisions hereof.

"Asset Management Fee" has the meaning assigned to such term in Section 8.1.

"Asset Management Services" has the meaning assigned to such term in Section 3.1.

"Asset Preservation Services" has the meaning assigned to such term in Section 3.2.

"Asset Sale Proceeds" in respect of a Facility means the cash proceeds received from the sale of such Facility net of reasonable attorney's fees, investment banking fees, accounting fees and other fees and all taxes applied or estimated (as determined in good faith by Owner of such Facility) required to be paid or that become due within the following 12 months as a result of such sale and any amounts escrowed or deferred (provided that when released such escrowed or deferred amounts shall be Asset Sale Proceeds when received).

"Availability of Funds" means, with respect to any obligation that constitutes a Manager Expense, an Operating Disbursement or a Direct Reimbursement Expense, that sufficient funds have been provided by Owners and are available to Manager to pay such obligation. For sake of clarity, with respect to costs for Manager's obligations constituting Manager Expenses, Availability of Funds means that the Asset Management Fee has been paid, without regard to the actual costs necessary to be incurred by Manager to fulfill such obligations.

"Boardman Corn Supply Agreement" means the Amended and Restated Corn Procurement and Handling Agreement between Pacific AG. Products, LLC and Boardman dated as of June 30, 2011, and any other agreement relating to the procurement of corn for the Boardman Facility.

"<u>Boardman Ethanol Sales and Marketing Agreement</u>" means the Amended and Restated Ethanol Marketing Agreement between Boardman and Kinergy Marketing, LLC, dated as of June 30, 2011, and any other agreement relating to the sale of ethanol from the Boardman Facility.

"Boardman Facility" means the Facility owned by Boardman.

"Boardman Facility Agreements" means this Agreement, the Boardman Corn Supply Agreement, the Technology Licensing Agreement to which Boardman is a party, the Boardman Ethanol Sales and Marketing Agreement, the Boardman WDG Sales and Marketing Agreement, and such other or additional material agreements entered into by Boardman with respect to the operation and maintenance of the Boardman Facility.

"Boardman WDG Sales and Marketing Agreement" means the Amended and Restated Distillers Grains Marketing Agreement between Boardman and Pacific Ag Products, LLC, dated as of June 30, 2011, and any other agreement relating to the sale of distillers grains from the Boardman Facility.

"Budget" means the Budget attached hereto as Appendix E, as amended or modified from time to time in accordance with Section 6.2.

"Burley Corn Supply Agreement" means the Amended and Restated Corn Procurement and Handling Agreement between Pacific AG. Products, LLC and Burley dated as of June 30, 2011, and any other agreement relating to the procurement of corn for the Burley Facility.

"Burley Ethanol Sales and Marketing Agreement" means the Amended and Restated Ethanol Marketing Agreement between Burley and Kinergy Marketing, LLC, dated as of June 30, 2011, and any other agreement relating to the sale of ethanol from the Burley Facility.

"Burley Facility" means the Facility owned by Burley.

"Burley Facility Agreements" means this Agreement, the Burley Corn Supply Agreement, the Technology Licensing Agreement to which Burley is a party, the Burley Ethanol Sales and Marketing Agreement, the Burley WDG Sales and Marketing Agreement, and such other or additional material agreements entered into by Burley with respect to the operation and maintenance of the Burley Facility.

"<u>Burley WDG Sales and Marketing Agreement</u>" means the Amended and Restated Distillers Grains Marketing Agreement between Burley and Pacific Ag Products, LLC, dated as of June 30, 2011, and any other agreement relating to the sale of distillers grains from the Burley Facility.

"Business Day" means any day other than a Saturday, Sunday or a day on which commercial banks in Sacramento, California or New York, New York are required or authorized to be closed.

"Consistent with Past Practices" means consistent with those practices previously utilized by the Manager with respect to a Facility pursuant to the Prior Agreement.

"Consumables" means all items consumed, or needing regular, periodic replacement or replenishment by Manager in the performance of services pursuant to this Agreement, including, but not limited to, chemicals, hand tools, catalysts, lubricants, rags, oils, filter media, ammonia, additives, anti-corrosion devices, gases (CO₂, O₂, halon, etc.), and other expendable materials.

"Credit Agreement" means the Credit Agreement, dated June 25, 2010, among Owners, Owner Agent, each of the Lenders from time to time party thereto, WestLB AG, New York Branch, as Administrative Agent and Collateral Agent, and Amarillo National Bank, as accounts bank, as the same may be amended, restated, modified or otherwise supplemented from time to time.

"<u>Direct Reimbursement Expenses</u>" means those expenses included in the line items of the Budget under the category "Asset Management Agreement-Direct Reimbursement."

"EBITDA" means, with respect to an Owner, the earnings of such Owner before interest, taxes, depreciation and amortization, reorganization adjustments, financing charges and fees, current month lower cost or market inventory adjustment, and other lender related expenses calculated in accordance with Exhibit III.

"EBITDA per Gallon of Operating Capacity" of a Facility means the quotient obtained by dividing EBITDA by Operating Capacity.

"Effective Date" has the meaning assigned to such term in Section 9.1.

"Environmental Law" means any statute, law, regulation, ordinance, rule, judgment, order, decree, legally binding directive or requirement, or any similar form of decision of or determination by, or any interpretation or administration of any of the foregoing by a Governmental Authority, relating to the environment, health or safety as affected by the environment or any Hazardous Material as now or hereinafter in effect.

"Extension Notice" has the meaning assigned to such term in Section 9.1.

"Facility" means an ethanol production facility owned by an Owner including (a) all equipment associated with the operation of such facility and forming part thereof, (b) the storage space for corn, ethanol and wet distillers grains, (c) administrative offices and building structures housing facility equipment, (d) site improvements such as roads, railroad spur lines, barge docks and fencing, and (e) the Loading/Unloading Facilities and (f) piping, structures and equipment for the delivery of ethanol and wet distillers grains and for water, fuel, sewer, waste water discharge and other Consumables required for facility operation and maintenance.

"<u>Facility Agreements</u>" means the Boardman Facility Agreements, the Burley Facility Agreements, the Madera Facility Agreements and the Stockton Facility Agreements.

"<u>Facility Manuals</u>" means Facility equipment manuals, system descriptions, system operating instructions, equipment maintenance instructions and pertinent design documentation as developed by the construction contractor of the Facility and/or Plant Designer.

"Facilities Records" has the meaning assigned to such term in Section 3.11.1.

"Force Majeure Event" has the meaning assigned to such term in Section 15.1.

"Governmental Authority" means any United States federal, state, municipal, local, territorial, or other governmental department, commission, board, bureau, agency, regulatory authority, instrumentality, judicial or administrative body.

"Hazardous Materials" means (a) any petroleum or petroleum products, radioactive materials, asbestos in any form that is or could become friable, urea formaldehyde foam insulation, and transformers or other equipment that contain dielectric fluid containing polychlorinated biphenyls; (b) any chemicals, materials or substances which are now or hereafter become defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous wastes," "toxic substances," "toxic pollutants," "pollution," "pollutants," "regulated substances," or works of similar import, under the Environmental Laws, including but not limited to the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended (42 U.S.C. § 9601 et seq.); the Hazardous Material Transportation Act, as amended (49 U.S.C. § 1801 et seq.); the Resource Conservation and Recovery Act, as amended (42 U.S.C. § 6901 et seq.); the Toxic Substances Control Act, as amended (42 U.S.C. § 7401 et seq.); the Clean Air Act, as amended (42 U.S.C. § 7401 et seq.); the Federal Water Pollution Control Act, as amended (33 U.S.C. § 1251 et seq.); or in the regulations promulgated pursuant to said laws; (c) any Material of Environmental Concern as defined in the Credit Agreement; and (d) any other chemical, material, substance or waste declared to be hazardous, toxic, or polluting material by any Governmental Authority, exposure to which is now or hereafter prohibited, limited or regulated by any Governmental Authority.

"Law" means any law, statute, act, legislation, bill, enactment, policy, treaty, international agreement, ordinance, judgment, injunction, award, decree, rule, regulation, interpretation, determination, requirement, writ or order of any Governmental Authority.

"Liabilities" has the meaning assigned to such term in Section 11.1.

"<u>Loading/Unloading Facilities</u>" means the rail spurs, barge and/or truck docks located at a Facility, and all loading and unloading equipment and facilities with respect thereto located at such Facility, including, but not limited to, all conveyors, lifts and elevators used in connection with movement of materials and products (including grain and grain products) in and out of the Storage Silos and other locations at a Facility.

"Madera Corn Supply Agreement" means any agreement relating to the procurement of corn for the Madera Facility.

"Madera Ethanol Sales and Marketing Agreement" means any agreement relating to the sale of ethanol from the Madera Facility.

"Madera Facility" means the Facility owned by Madera.

"Madera Facility Agreements" means this Agreement, the Madera Corn Supply Agreement, the Technology Licensing Agreement to which Madera is a party, the Madera Ethanol Sales and Marketing Agreement, the Madera WDG Sales and Marketing Agreement, and such other or additional material agreements entered into by Madera with respect to the operation and maintenance of the Madera Facility.

- "Madera WDG Sales and Marketing Agreement" means any agreement relating to the sale of distillers grains from the Madera Facility.
- "Magic Valley Air Permit" means a final, non-appealable air quality permit required to operate the Magic Valley Facility at not less than nameplate capacity.
 - "Magic Valley Facility" means the Facility owned by Burley.
 - "Manager" has the meaning assigned to such term in the Preamble.
- "Manager Account" means a depositary account designated by Manager, from time to time, which account shall be under the sole dominion and control of Manager and located in a banking institution designated by Manager.
- "Manager Expenses" means the expenses incurred by Manager in performing the Services that are to be for the account of Manager, as specified in the Manager Expense portion of the Budget.
 - "Manager Indemnified Person" has the meaning assigned to such term in Section 11.2.
 - "Manager Information" has the meaning assigned to such term in Section 13.2.
 - "Manager Proprietary Property" has the meaning assigned to such term in Section 9.5(a).
- "NewCo" means New PE Holdco LLC, a Delaware limited liability company and the owner on the date hereof, directly or indirectly, of all the equity interests in Owner Agent and each Owner.
- "Operating Capacity" means 57,142,857 un-denatured gallons annually in the case of each of the Magic Valley and the Stockton Facilities and 38,095,238 un-denatured gallons annually in the case of each of the Boardman and Madera Facilities, plus, in each case, the actual gallons of denaturant used annually with regard to any Facility not in Cold Shutdown.
- "Operating Disbursements" means those expenses included in the line items of the Budget under the category "Operating Disbursements".
 - "O&M Services" has the meaning assigned to such term in Section 3.3.
 - "Owner Agent" has the meaning assigned to such term in the Preamble
 - "Owners" has the meaning assigned to such term in the Preamble.
 - "Owner Indemnified Person" has the meaning assigned to such term in Section 11.1.
 - "Party" has the meaning assigned to such term in the Preamble.

"Permits" means all permits, authorizations, registrations, consents, approvals, waivers, exceptions, variances, orders, judgments, written interpretations, decrees, licenses, exemptions, publications, filings, notices to and declarations of or with, or required by, any Governmental Authority, or required by any Law, and shall include all environmental and operating permits and licenses that are required for the full use, occupancy, zoning and operation of each Facility.

"Person" means any individual, partnership, corporation, association, business, trust, government or political subdivision thereof, governmental agency or other entity.

"Plant Designer" means Delta-T Corporation.

"Prior Agreement" has the meaning assigned to such term in the Recitals.

"Prudent Ethanol Practices" means those reasonable practices, methods and acts that (i) are commonly used in the regions where the Facilities are located to manage and maintain ethanol production, distribution, equipment and associated facilities of the size and type that comprise the Facilities safely, reliably, and efficiently and in compliance with applicable Laws, manufacturers' warranties and manufacturers' and licensor's recommendations and guidelines, and (ii) in the exercise of reasonable judgment, skill, diligence, foresight and care are expected of an ethanol plant manager, in order to efficiently accomplish the desired result consistent with safety standards, applicable Laws, manufacturers' warranties and manufacturers' recommendations, in each case taking into account whether a Facility is in operation or is in Cold Shutdown. Prudent Ethanol Practices does not necessarily mean one particular practice, method, equipment specifications or standard in all cases, but is instead intended to encompass a broad range of acceptable practices, methods, equipment specifications and standards.

"Sale Price per Gallon" in respect of a Facility (or the equity interests in Owner of such Facility) means the quotient obtained by dividing (i) the excess of (x) Asset Sale Proceeds in respect of such Facility (or such equity interests) over (y) the pro rata portion of the Loans allocated (on the basis of relative Operating Capacities) to such Facility by (ii) the Operating Capacity of such Facility.

"Services" means, collectively, the Asset Management Services, the Asset Preservation Services and the O&M Services and any other services provided by or on behalf of the Manager with respect to the Facilities in accordance with this Agreement.

"Sponsor Support Agreement" means that certain Amended and Restated Sponsor Support Agreement, dated as of October 22, 2008, among Owner Agent, Manager and WestLB AG, New York Branch, as Administrative Agent.

"Stockton Completion Obligation" means those actions required to be taken to cause the Stockton Facility to achieve Final Completion (as defined in the Pre-Petition Credit Agreement) including receipt of all Permits.

"Stockton Corn Supply Agreement" means the Amended and Restated Corn Procurement and Handling Agreement between Pacific AG. Products, LLC and Stockton dated as of June 30, 2011, and any other agreement relating to the procurement of corn for the Stockton Facility.

"Stockton Ethanol Sales and Marketing Agreement" means the Amended and Restated Ethanol Marketing Agreement between Stockton and Kinergy Marketing, LLC, dated as of June 30, 2011, and any other agreement relating to the sale of ethanol from the Stockton Facility.

"Stockton Facility" means the Facility owned by Stockton.

"Stockton Facility Agreements" means this Agreement, the Stockton Corn Supply Agreement, the Technology Licensing Agreement to which Stockton is a party, the Stockton Ethanol Sales and Marketing Agreement, the Stockton WDG Sales and Marketing Agreement, and such other or additional material agreements entered into by Stockton with respect to the operation and maintenance of the Stockton Facility.

"Stockton WDG Sales and Marketing Agreement" means the Amended and Restated Distillers Grains Marketing Agreement between Stockton and Pacific Ag Products, LLC, dated as of June 30, 2011, and any other agreement relating to the sale of distillers grains from the Stockton Facility.

"Storage Silos" means Owners' grain storage silos located at a Facility.

"Supplemental Termination Payment" means the product obtained by multiplying (i) the difference between (x) sixty (60) and (y) the number of days notice (fewer than sixty (60)) that Manager is given pursuant to clause (y) of Section 9.4 by (ii) the quotient of (x) the Asset Management Fee with respect to the relevant Facility divided by (y) 30.

"<u>Technology Licensing Agreement</u>" means (i) each License for Technology between an Owner of a Facility (other than Madera) and Plant Designer, each dated as of September 6, 2006 and (ii) the License for Technology between Madera and Plant Designer, dated as of September 1, 2005 in each case, as such agreement may from time to time be amended.

"Termination Payment" has the meaning assigned to such term in Section 9.6.1.

- 1.2 <u>Interpretation</u>. The following interpretations and rules of construction shall apply to this Agreement:
- (a) titles and headings are for convenience only and will not be deemed part of this Agreement for purposes of interpretation;
- (b) unless otherwise stated, references in this Agreement to "Sections," "Appendices" or "Articles" refer, respectively, to Sections, Appendices or Articles of this Agreement;
- (c) "including" means "including, but not limited to", and "include" or "includes" means "include, without limitation" or "includes, without limitation";
- (d) "hereunder", "herein", "hereto" and "hereof", when used in this Agreement, refer to this Agreement as a whole and not to a particular Section or clause of this Agreement;

- (e) in the case of defined terms, the singular includes the plural and vice versa;
- (f) unless otherwise indicated, all accounting terms not specifically defined shall be construed in accordance with generally accepted accounting practices in the United States;
- (g) unless otherwise indicated, each reference to a particular Law is a reference to such Law as it may be amended, modified, extended, restated or supplemented from time to time, as well as to any successor Law thereto;
- (h) unless otherwise indicated, references to agreements shall be deemed to include all subsequent amendments, supplements and other modifications thereto; and
- (i) unless otherwise indicated, each reference to any Person shall include such Person's successors and permitted assigns.

ARTICLE II

APPOINTMENT

Owners hereby appoint and retain Manager to provide the Services from the Effective Date on the terms and conditions set forth in this Agreement. Manager hereby accepts such appointment and agrees to perform the Services in accordance with the terms and conditions set forth in this Agreement.

ARTICLE III

RESPONSIBILITIES OF MANAGER

- 3.1 <u>Scope of Asset Management Services</u>. With respect to each Facility, commencing on the Effective Date, subject to the terms of this Agreement and the Availability of Funds, Manager shall perform, or cause to be performed, the following services with respect to such Facility (collectively, the "<u>Asset Management Services</u>"):
- Agent, Manager shall administration of Financing Documents. Subject to the terms of this Agreement and the direction of Owner Agent, Manager shall administer the Owners' compliance with all covenants and obligations set forth in the Financing Documents, including monitoring compliance with the Financing Documents and advising Owners as to their compliance obligations thereunder and providing Owner Agent with Manager's analysis and recommendations regarding the application of the Financing Documents. In furtherance of the foregoing, Appendix A-1 sets forth various time and certification requirements regarding Manager's furnishing to Owner of various reports, budgets and other submissions for review by Owner prior to the submission of same to the Administrative Agent. For convenience, Appendix A-1 also sets forth the time and requirements for reports called for by this Agreement and the Facility Agreements. Manager shall not take any action that would reasonably be expected to cause any Owner to contravene such Owner's obligations under the Financing Documents. Manager shall be authorized to deal directly with the Administrative Agent and the Collateral Agent pursuant to the Financing Documents. Manager shall reasonably cooperate with the Administrative Agent and the Collateral Agent pursuant to the Financing Documents. Manager shall reasonably cooperate with the Administrative Agent, the Collateral Agent and the Consultants. The parties to this Agreement hereby agree that the terms of this Agreement shall not vary or amend the obligations of any Owner under any Financing Document to which such Owner is a party or subject Manager to any obligations under the Financing Documents.

- 3.1.2 <u>Billing and Collection of Revenues.</u> Manager shall implement and maintain billing and collection procedures in respect of all accounts receivable and other amounts due Owners.
- 3.1.3 <u>Bank Accounts and Disbursement of Funds.</u> Manager shall establish and/or maintain on behalf of and in the name of the applicable Owner one or more bank accounts as may be required by the applicable Owner or the Financing Documents. Subject to availability of adequate funds in such accounts, Manager shall withdraw from such accounts such funds as may be necessary to pay such Owner's Operating Disbursements and Direct Reimbursement Expenses in accordance with the Budget.
- 3.1.4 <u>Accounting and Documentation.</u> Manager shall provide full bookkeeping, accounting (including tax accounting), and record keeping services to Owners as required from time to time by Owners and the Financing Documents.
- 3.1.5 <u>Insurance</u>. Manager shall implement Owners' insurance programs, including procuring and maintaining any and all insurance required to be maintained by Owners under the Financing Documents. Manager also shall be responsible for administering all claims, arranging for all payments, and making all collections on behalf of Owners under insurance policies covering Owners.
- 3.1.6 <u>Licenses and Permits</u>. Manager shall monitor and use commercially reasonable efforts to assist each Owner with its respective obligations to maintain compliance with any required permits, licenses and governmental approvals required and obtained by or for an Owner in connection with the ownership or operation of its Facility. Manager shall, upon request by any Owner, prepare or cause the preparation of any application, filing or notice related thereto, shall cause such materials to be submitted to, and shall represent Owner in contacts with, the appropriate governmental agency, and shall perform all ministerial or administrative acts necessary for timely issuance and the continued effectiveness thereof. Copies of all permits, licenses and governmental approvals obtained by or for an Owner pursuant hereto shall be maintained by Manager at its offices and at the respective Facility sites.
- 3.1.7 <u>Public Relations.</u> Manager shall coordinate all public and community relations matters of Owners with respect to the Facilities as directed by Owner Agent. Notwithstanding the foregoing, the Manager shall not issue (nor have any obligation to issue) any press release regarding an Owner or a Facility without the prior written consent of such Owner.
- 3.1.8 <u>Reports and Budgets</u>. Manager shall prepare and distribute, or cause to be prepared and distributed, all financial or other reports, budgets (including the Budget), estimates, tax returns and other information required to be prepared and distributed by Owners pursuant to the Financing Documents.

- 3.1.9 <u>Dispute Resolution.</u> Subject to the directions of Owner Agent, Manager shall manage any litigation, arbitration, or other proceedings involving any Facility (except any litigation, arbitration, or other proceedings involving Manager). Manager shall obtain Owner Agent's written approval prior to commencing any litigation, arbitration or other proceeding on behalf of an Owner and Manager shall periodically advise Owner Agent on the status of each litigation, arbitration and other proceeding involving any Facility. In addition, notwithstanding the foregoing, except as otherwise authorized by Owner Agent, Manager shall not settle any claim brought by or against an Owner without the prior written approval of Owner Agent.
- 3.1.10 <u>Materials</u>. Manager shall provide all materials necessary for the performance of the Asset Management Services and shall provide appropriate office space for its personnel performing the Asset Management Services.
- 3.1.11 <u>Sale of Facilities.</u> Manager shall use commercially reasonable efforts to assist Owners with the sale of the Facilities.
- 3.1.12 <u>Offices</u>. Manager shall maintain the principal office of each Owner and arrange for the provision of cleaning, security and other necessary services with respect to each such office and to each Facility.
- 3.1.13 <u>Consultation</u>. Manager shall consult with Owner Agent on all aspects of the preservation, operation and maintenance of each Facility, at such places and times as Owner Agent may reasonably request.
- 3.1.14 Other. Manager shall use commercially reasonable efforts to provide any other assistance or services reasonably requested by Owners that are consistent with the foregoing services and necessary for, or materially beneficial to, the management or administration of the Facilities including the services set forth in <u>Appendix A-1</u>.
- 3.2 <u>Scope of Asset Preservation Services</u>. With respect to each Facility that is in Cold Shutdown, subject to the terms of this Agreement and the Availability of Funds, Manager shall perform, or cause to be performed, the following services with respect to such Facility (collectively, the "Asset Preservation Services"):
- 3.2.1 <u>Layup Services.</u> Manager will take all commercially reasonable steps to ensure that site security measures, equipment preservation and general site maintenance is conducted in order to support a cost effective return of each Facility to operational status once schedules therefor are established by Owners and communicated to Manager. Manager shall:
 - (a) be in complete charge of, and have care and custody over, each Facility;
 - (b) perform, or cause to be performed on behalf of Owners, all maintenance of each Facility Consistent with
 - (c) perform periodic inspections Consistent with Past Practice.

Past Practice: and

- 3.2.2 <u>Waste Management</u>. Subject to Environmental Laws and any permits maintained with respect to each Facility by its Owner or Manager, Manager shall be responsible for the onsite management of all wastes generated by or used in the preservation of each Facility.
- 3.2.3 Other. Manager shall use commercially reasonable efforts to provide any other assistance or services reasonably requested by Owners that are consistent with the foregoing services, Consistent with Past Practices and necessary for, or materially beneficial to, the preservation of the Facilities including the services set forth in Appendix A-1.
- 3.3 <u>Scope of O&M Services</u>. With respect to each Facility that is not in Cold Shutdown, commencing on the Effective Date and subject to the terms of this Agreement, the Availability of Funds and Consistent with Past Practices, Manager shall perform, or cause to be performed, the following services (collectively, the "<u>O&M Services</u>"):
 - 3.3.1 Operations and Maintenance. Manager shall operate and maintain each such Facility. Manager shall:
 - (a) perform, or cause to be performed on behalf of such Facility, all operations and maintenance of such

maintain such Facility;

Facility;

- (b) supply, or cause to be supplied, all goods and materials, including spare parts, required to operate and
- (c) maintain, control and store such Facility's equipment and spare parts inventory;
- (d) perform periodic inspections; and
- (e) coordinate compliance by Owner with the applicable Facility Agreements.
- 3.3.2 <u>Waste Management.</u> Subject to Environmental Laws and any permits maintained with respect to such Facility by Owner of such Facility or Manager, Manager shall be responsible for the onsite management of all wastes generated by or used in the operation or maintenance of such Facility.
- 3.3.3 Other. Manager shall use commercially reasonable efforts to provide any other assistance or services reasonably requested by Owner of such Facility, that are consistent with the foregoing services and necessary for, or materially beneficial to, the operations and maintenance of such Facility, respectively, including the services set forth in Appendix A-2.
- 3.4 <u>Subcontracts.</u> Manager shall not enter into any subcontract for the services described herein without the prior written consent of Owner Agent; provided that Manager shall be permitted to subcontract certain of the Services relating to the maintenance and operation of the Facilities to Affiliates of Manager (other than Owners) Consistent with Past Practices (it being understood that Manager shall remain primarily liable for any services performed by any subcontractor).

- 3.5 <u>Standards for Performance of the Asset Management Services</u>. Manager shall perform the Asset Management Services in a prudent, businesslike and efficient manner in accordance with (i) all applicable Laws (including Environmental Laws) and Permits, (ii) the applicable terms and conditions of the Financing Documents and (iii) this Agreement.
- 3.6 <u>Standards for Performance of the Asset Preservation Services and the O&M Services.</u> Subject to Availability of Funds and Consistent with Past Practices, Manager shall perform the Asset Preservation Services and the O&M Services in accordance with (i) the Facility Manuals, (ii) all applicable Laws (including Environmental Laws) and Permits, (iii) Prudent Ethanol Practices, (iv) the applicable Facility Agreements and (v) this Agreement. Manager shall obtain and maintain in effect all licenses and permits required to allow Manager to do business or perform its services hereunder in the jurisdictions where such services are to be performed except where the failure to do so shall not adversely affect Manager's ability to perform its obligations under this Agreement.

3.7 Personnel Standards.

- 3.7.1 Manager shall provide and make available as necessary, in accordance with the requirements of this Agreement, all labor and professional, supervisory and managerial personnel as are required to perform its services hereunder in accordance with the terms hereof. Such personnel shall be qualified and experienced in the duties to which they are assigned and shall be the employees of Manager or its Affiliates, and their working hours, rates of compensation and all other matters relating to their employment shall be determined solely by Manager. Manager shall retain sole authority, control and responsibility with respect to its employment policy in connection with the performance of its obligations hereunder. A preliminary listing of personnel that Manager anticipates will be necessary to operate, preserve and/or maintain each Facility in its current state of operation or Cold Shutdown as applicable as such state may change from time to time, including function, the number of such positions and the date by which such personnel should be hired, is set forth in Appendix B, it being understood that such listing is non-binding and shall be revised from time to time (but not less frequently than quarterly, if changes have occurred) by Manager in consultation with Owner Agent.
- 3.7.2 Upon the written request of Owner Agent, Manager shall remove from each site and each Facility workforce, any employee or subcontractor.
- 3.8 <u>Training.</u> Manager shall maintain regular training procedures approved by Owner Agent in the exercise of its reasonable judgment and discretion. Such procedures shall be adequate to keep Manager's personnel informed and knowledgeable regarding the operation and maintenance of each Facility.
- 3.9 <u>Additional Obligations of Manager</u>. Anything contained herein to the contrary notwithstanding and without regard to Availability of Funds, Manager shall (i) take all actions as may be necessary to cause the Stockton Facility to satisfy the Stockton Completion Obligation and (ii) perform its obligations under Section 2.01(b) of the Sponsor Support Agreement to cause the Magic Valley Facility to obtain the Magic Valley Air Permit. This <u>Section 3.9</u> shall not be deemed to make Manager a party to the Facility Agreements (other than this Agreement) or to impose any obligations on Manager under the Facility Agreements (other than this Agreement).

3.10 <u>Licenses and Permits</u>. Manager has reviewed and shall continue to review all Laws and regulations containing or establishing compliance requirements in connection with the operation and maintenance of each Facility and applicable to Manager in connection with its obligations under this Agreement, and assist Owners in securing and complying with, as appropriate, all Permits necessary for the operation and maintenance of each Facility (and renewals or replacements of the same), including those relating to Facility operation, waste water and sewer use and treatment, chemical and other waste including Hazardous Materials; provided, however, that all such permits, licenses and approvals relating to Hazardous Materials shall be in the name of Owners except for any individual licenses or permits required under Section 3.6. Manager shall also initiate and maintain precautions and procedures necessary to comply with applicable provisions of all such Laws (including Environmental Laws), including those related to prevention of injury to persons or damage to property at each Facility.

3.11 Records and Reports; Other Material Information

- 3.11.1 Records and Reports. Manager shall prepare and maintain logs, records and reports documenting the operation and maintenance of each Facility including all information and reports required by applicable Laws or beneficial for proper operation and maintenance of each Facility in accordance with Prudent Ethanol Practices. Manager shall also prepare reports and data related to the maintenance of Hazardous Materials onsite at each Facility in a manner complying with applicable Environmental Laws, and shall maintain current revisions of the drawings, specifications, lists, clarifications and other materials provided to Manager by Owners, construction contractors and/or Plant Designer. Appendix A-1 details timing and certification requirements relating to certain of such reports. Copies of all such reports that may be submitted to any Governmental Authority by Manager shall be transmitted to Owner Agent. All such reports and other documents specified in this Section 3.11.1 are referred to as the "Facilities Records".
- 3.11.2 Other Material Information. Manager shall promptly submit to Owner Agent any material information that it or any of its Affiliates may have concerning new or significant developments relating to any Facility and, upon Owner Agent's reasonable request, shall promptly submit any other information that it or any of its Affiliates may have concerning any Facility or the services performed by Manager hereunder.
- 3.12 <u>No Liens or Encumbrances</u>. Manager shall keep and maintain each Facility free and clear of all liens and encumbrances resulting from the action of Manager or work done at the request of or by Manager.
- 3.13 Emergency Action. In the event of an emergency affecting the safety or protection of Persons or endangering a Facility or property located at a Facility, Manager shall promptly notify Owner Agent and, take prompt action to attempt to prevent any damage, injury or loss resulting from such emergency.
- 3.14 <u>Scope; Manager Authority</u>. On at least 60 days prior written notice, Owner Agent may revoke or rescind all or any material part of the authority granted to Manager or materially reduce or materially restrict the scope of the Services.

- 3.15 Retention of Control by Owners. Notwithstanding anything in this Agreement to the contrary, Owners and Manager expressly acknowledge and agree that: (a) this Agreement does not convey ownership or control over the Facilities from Owners to the Manager; and (b) Owners retain ultimate decision-making authority with respect to the Facilities, including ultimate decision-making authority relating to the operation of, and sale of products produced by, the Facilities.
- 3.16 <u>Limitations on Performance</u>. Notwithstanding anything herein to the contrary, none of the following shall result in a breach of Manager's obligations under this Agreement to the extent caused by: (i) lack of Availability of Funds, (ii) Force Majeure Events or (iii) any failure by Owners to perform their respective obligations under this Agreement.
- 3.17 <u>Deficiency of Funds</u>. If funds shall not be sufficient to make applicable disbursements for Operating Disbursements or Direct Reimbursement Expenses, Manager shall promptly notify Owner Agent and Owner Agent shall promptly provide the required funds, subject to the terms of this Agreement. Manager, in its sole discretion, may elect (but shall not be obligated), to advance any such funds for the account of Owners, and Owner Agent shall promptly reimburse Manager for any such advances properly made by Manager in accordance with the terms of this Agreement. Manager shall use commercially reasonable efforts to keep Owner Agent advised as to projected cash deficits so as to permit the orderly funding thereof by Owner Agent.

ARTICLE IV

ITEMS TO BE FURNISHED BY OWNERS

- 4.1 <u>General.</u> Owners shall furnish to Manager, at Owners' expense, the information, services, materials and other items described in this Article IV. All such items shall be made available at such times and in such manner as may be reasonably required for the expeditious and orderly performance of the Services by Manager.
- 4.2 <u>Information</u>. Owners shall provide copies of the Facility Agreements and the Facility Manuals to Manager as well as technical, operational and other Facility information reasonably available to Owners or in Owners' possession and necessary for the performance of the Services. Subject to the standards of performance set forth in Article III, Manager shall be entitled to rely upon such information in the performance of the Services.
- 4.3 <u>Corn, Fuel and Other Materials.</u> During the period in which Manager is required to provide O&M Services to a Facility, Owner of such Facility shall be responsible for furnishing and delivering to such Facility: (a) corn, (b) water, (c) natural gas, (d) electricity, (e) denaturant, and (f) any other materials necessary for the operation of such Facility, in each case, in sufficient quantities to produce ethanol and wet distillers grains in accordance with the applicable Facility Agreements.
- 4.4 <u>Cold Shutdown or Start Up of a Facility</u>. From time to time, upon not less than sixty (60) days notice to Manager, Owner Agent or Owner of a Facility may elect to recommence operations of such Facility that is then in Cold Shutdown or to place such Facility in Cold Shutdown. Within ten (10) days of receipt of any such notice, Manager shall provide Owner's Agent and Owner of such Facility with a budget for any costs to be incurred with such recommencing operation or Cold Shutdown. Upon agreement of Manager, Owner Agent and Owner of such Facility with respect to such budget, and the concurrence therewith by the Administrative Agent, Manager shall recommence operations or place such Facility in Cold Shutdown, as applicable.

ARTICLE V

DEVELOPMENT PROJECTS; FACILITY SERVICES

Development Projects. If, during the term of this Agreement, Manager intends to develop or pursue any business opportunity that involves or promotes any Facility assets, properties located adjacent to the Facilities or products produced at the Facilities, then Manager shall notify Owners of such opportunity, including providing reasonable detail thereof (following Owners' execution of a reasonable and customary confidentiality and non-intervention agreement, if required by Manager), and Manager agrees to negotiate, in good faith, Owners participating in such business opportunity on terms mutually acceptable to the parties. Notwithstanding the foregoing, Manager (i) shall not be under any obligation to include any Owner in any transaction regarding any such opportunities and Owners shall have no right to participate in any such transaction, (ii) shall be free to pursue and engage in such opportunity with any or no party as Manager shall in its sole and absolute discretion deem appropriate, (iii) shall be free to negotiate concurrently with third parties regarding such opportunity and (iv) may cease discussions with Owners regarding any such opportunity at any time and in Manager's sole discretion.

ARTICLE VI

REPORTING AND PERSONNEL

- 6.1 <u>Accounts and Reports.</u> Manager shall furnish or cause to be furnished to Owner Agent (i) the reports required to be delivered to the Administrative Agent pursuant to the Financing Documents and (ii) the following reports and information:
- 6.1.1 CS Reports. (a) With respect to each Facility that is in Cold Shutdown, as soon as available and in any event within 25 days following the end of each calendar month, Manager shall submit to Owner Agent a summary report in the form attached hereto as Appendix C, which report shall include, with respect to each Facility, a numerical and narrative assessment in respect of such month of (i) the Facility's compliance with each category in the Budget, (ii) plant availability, (iii) cash receipts and disbursements including balances in the Accounts, (iv) major maintenance activity, (v) material casualty losses (whether or not covered by insurance), (vi) disputes with any contractor, materialman, supplier or other Person and any related claims against any Owner, (vii) compliance with governmental permits, and (viii) a comparison of figures to corresponding figures provided in the prior month.
- (b) With respect to each Facility that is in Cold Shutdown, as soon as practicable and in any event within 25 days following the end of each calendar quarter, Manager shall submit to Owner Agent a summary report containing the information required to be provided pursuant to Section 6.1.1(a) for the quarter then ended.

- 6.1.2 Operating Reports. With respect to each Facility that is not in Cold Shutdown, as soon as available and in any event within 25 days following the end of each calendar month, Manager shall submit to Owner Agent a summary report in the form attached hereto as Appendix D, which report shall include, with respect to each such Facility, a numerical and narrative assessment in respect of such month of (i) each such Facility's compliance with each category in the Budget, (ii) ethanol and WDG production and delivery, (iii) corn deliveries and use, (iv) plant availability, (v) cash receipts and disbursements including balances in the Accounts, (vi) major maintenance activity, (vii) material casualty losses (whether or not covered by insurance), (viii) disputes with any contractor, materialman, supplier or other Person and any related claims against the Owner of each such Facility, (ix) compliance with governmental permits, and (x) a comparison of figures to corresponding figures provided in the prior month.
- 6.1.3 <u>Manager Report.</u> As soon as available and in any event within ten (10) days after the filing thereof, Manager shall submit to Owner Agent and the Administrative Agent copies of all reports filed by Manager or any Affiliate with the Securities and Exchange Commission, any communications or information provided by Manager to its shareholders, and the monthly financial statements of Kinergy Marketing LLC as provided to its working capital lenders. Each report shall be certified as complete and correct by an Authorized Officer of Manager.
- 6.1.4 Other Information. Any other information concerning the Services, Owners or the Facilities or reasonably requested by Owner Agent regarding any Facility or the Services.
- Budget. The Budget sets forth the budgeted amounts for all Operating Disbursements and Direct Reimbursement Expenses and for the Asset Management Fee. Manager shall submit a revised Budget each quarter and each year in conjunction with the quarterly and annual budgets provided to the Lenders pursuant to the Credit Agreement. If accepted by the Owners and the Lenders, each such quarterly budget shall serve as the Budget hereunder. Manager shall promptly notify Owner Agent of any actual or anticipated variance from the amounts budgeted for Operating Disbursements and Direct Reimbursement Expenses, the reasons therefor and Manager's recommendations with respect thereto. Each Owner shall be responsible for (but shall not be obligated to fund) all Operating Disbursements and Direct Reimbursement Expenses in respect of its Facility; provided that any failure by an Owner to provide such funding shall relieve Manager of any obligation hereunder for which there is no Availability of Funds. Each Owner agrees to pay Manager the Asset Management Fee in respect of its Facility; provided that, except as expressly set forth herein, any such failure to fund shall relieve Manager of its obligations hereunder. Manager agrees that the amount of the Asset Management Fee is fixed and that any increase or decrease in the Manager Expenses shall be solely for the account of the Manager and shall not result in any increase or decrease in the Asset Management Fee.
- 6.3 <u>Manager Representative</u>. Manager has appointed a representative (a "<u>Manager Representative</u>") authorized and empowered to act for and on behalf of Manager on all matters concerning this Agreement and the Services with respect to a Facility. The appointment of any Manager Representative shall be subject to the reasonable approval of Owner Agent. Such appointment shall remain in full force and effect until such Manager Representative is replaced by Manager with the reasonable approval of Owner Agent. At any time, a Manager Representative may act through or be represented by one or more individuals appointed by Manager.

ARTICLE VII

LIMITATIONS ON AUTHORITY

Notwithstanding any provision in this Agreement to the contrary, unless otherwise approved in writing in advance by Owner Agent, Manager shall not (and shall not permit any of its agents or representatives to):

- (a) sell, lease, pledge, mortgage, encumber, convey, or make any license, exchange or other transfer or disposition of any property or assets of an Owner (other than products produced by an Owner for sale in the ordinary course of business), including any property or assets purchased by Manager hereunder;
- (b) make, enter into, execute, amend, terminate, modify or supplement any contract or agreement (including any labor or collective bargaining agreement) on behalf of or in the name of an Owner;
- (c) make any expenditure or acquire any equipment, materials, assets or other items, except in substantial conformity with the Budget, or consent or agree to do any of the foregoing; <u>provided</u>, that in the event of an emergency affecting the safety or protection of Persons or endangering a Facility or property located at a Facility, Manager, without approval from Owner Agent, shall be authorized to take all reasonable actions to prevent damage, injury or loss;
- (d) settle, compromise, assign, pledge, transfer, release or consent to the compromise, assignment, pledge, transfer or release of, any claim, suit, debt, demand or judgment against or due by, Owners, or initiate, submit or respond to any such claim, dispute or controversy or arbitration or judicial process, or stipulate in respect thereof to a judgment, or consent to the same; or
 - (e) INTENTIONALLY LEFT BLANK
 - (f) describe itself as an employee, agent or representative of any Owner;
 - (g) make any warranty or representation relating to or on behalf of any Owner;
 - (h) engage in any other transaction on behalf of Owners not permitted under this Agreement; or
 - (i) commit to or enter into any agreement or arrangement with respect to any of the foregoing.

ARTICLE VIII

COMPENSATION

8.1 <u>Asset Management Fee.</u> As compensation to Manager for the performance of the Services in respect of a Facility, Owner of such Facility shall pay Manager a monthly management fee equal to (i) \$75,000 for each calendar month during which a Facility is not in Cold Shutdown, and (ii) \$40,000 for each calendar month during which a Facility is in Cold Shutdown (the "<u>Asset Management Fee</u>"), payable in advance in equal semi-monthly installments on the 1st and 11th Business Days of each month by deposit into the Manager Account (and pro rated for partial calendar months) commencing on the Effective Date and continuing for the term of this Agreement with respect to such Facility.

In addition to the foregoing compensation, during any six-month period (measured on September 30 and March 31 of each year commencing March 31, 2011) in which any Facility shall have an annualized EBITDA Per Gallon of Operating Capacity of \$.20 or greater, Owner of such Facility shall pay to Manager a performance bonus equal to the product of 3% of the amount by which annualized EBITDA Per Gallon of Operating Capacity at such Facility for such six-month period exceeds \$.20 multiplied by the number of gallons of ethanol produced at such Facility during such period; provided that (i) no performance bonus shall be paid at any time when a Default or an Event of Default under the Credit Agreement as a result of Borrower's failure to pay any amounts then due and owing shall exist and be continuing; (ii) such performance bonus shall be capped at \$2.2 million for each six-month period; and (iii) such performance bonus shall be reduced by 25% if all Facilities then operating do not operate at a minimum average yield (in the aggregate across all operating Facilities) of 2.70 gallons of denatured ethanol per bushel of corn during such six-month period.

An example of this computation is set forth in Exhibit I.

8.2 <u>Sale Incentive Fee.</u> Upon the sale of all or substantially all the assets of a Facility or all of the equity interests in an Owner of a Facility, in each case to a third party (for sake of clarity, a sale to Manager or any Affiliate of Manager shall not be considered a sale to a third party for this purpose), Manager shall receive an Incentive Fee with respect to such sale, as set forth below:

"Tier"	Sale Price per Gallon	Incentive Fee
Tier I	\$.60 or less	0
Tier II	Above \$.60 up to and including \$.70	The excess of Sale Price Per Gallon over \$.60, to and including the lesser of the Sale Price Per Gallon and \$.70, multiplied by the Operating Capacity of the Facility (in gallons), multiplied by 0.5%; <i>plus</i>
m; III		
Tier III	Above \$.70 up to and including \$.80	Off the Sale Price Per Gallon exceeds \$.70, the excess of the Sale Price Per Gallon over \$.70, to and including the lesser of the Sale Price Per Gallon and \$.80, multiplied by the Operating Capacity of the Facility (in gallons), multiplied by 1.0%; plus
Tier IV	Above \$.80	If the Sale Price Per Gallon exceeds \$.80, the excess of the Sale Price Per Gallon over \$.80, multiplied by the Operating Capacity of the Facility (in gallons), multiplied by 1.5%.

An example of this computation is set forth in Exhibit II.

- 8.3 <u>Lien Waivers</u>. In connection with any payment of the Asset Management Fee, Manager shall provide such releases or waivers of mechanics or other liens as any Owner may require.
- 8.4 <u>Netting.</u> If the Parties each owe an amount to the other for obligations incurred under this Agreement, the undisputed portion(s) of such amounts shall be aggregated, and the Parties shall satisfy their payment obligations through netting, in which case the Party owing the greater aggregate undisputed amount shall pay to the other Party the difference between the undisputed amounts owed.

8.5 INTENTIONALLY LEFT BLANK

ARTICLE IX

TERM

9.1 Term. This Agreement shall be effective as of June 30, 2011 (the "Effective Date") and, unless earlier terminated in accordance with its terms, shall continue in effect until and including June 30, 2012; provided, that either Owner Agent or Manager may extend this Agreement for additional one year periods, in each case by written notice to the other (an "Extension Notice") delivered not less than ninety (90) days prior to the end of the original or renewal term, provided further that this Agreement shall nonetheless terminate if the recipient of any such Extension Notice rejects such extension in a written notice delivered to the party that issued such Extension Notice not more than fifteen (15) days after receipt of the Extension Notice. Notwithstanding anything to the contrary contained herein, the obligations set forth in Section 3.9 hereof shall survive the termination or expiration of this Agreement; provided, that (i) following the termination of this Agreement with respect to the Magic Valley Facility by Burley pursuant to Section 9.4, the total liability of Manager with respect to such obligations relating to the Magic Valley Facility hereunder and under Section 2.01(b) of the Sponsor Support Agreement shall not exceed the reasonable cost of performing such obligations as estimated by the Independent Engineer plus fifteen percent (15%) of such estimate and (ii) such obligations relating to the Stockton Facility shall survive only to the extent that Stockton pays to Manager any amounts required to perform such obligations constituting Operating Disbursements or Direct Reimbursement Expenses.

9.2	Owner Defaults and Manager Remedies. Upon the occurrence of any of the following events, Manager may exercise such
rights and remedies	as may be available to it at law or in equity, including the right to terminate this Agreement with respect to any Facility or
Facilities, by written	notice to the Owners:

- (a) the failure by an Owner to pay the Asset Management Fee required to be paid to Manager hereunder when due, and such failure continues for ten (10) days after receipt of written notice from Manager of such failure, with respect to the first such failure and within five (5) days after receipt of written notice from Manager of such failure with respect to any subsequent such failure;
- (b) the failure by an Owner to make any other payment, deposit or transfer required to be paid to Manager hereunder when due and such failure continues for fifteen (15) days after receipt of written notice from Manager of such failure;
- (c) the failure of any statement, representation or warranty made by such Owner in this Agreement to have been correct in any material respect when made if such failure could reasonably be expected to have a material adverse effect on such Owner's ability to perform its obligations under this Agreement; or
- (d) the failure of such Owner to perform any of its material obligations under this Agreement (other than with respect to the payment of money) and such failure continues for thirty (30) days after receipt of written notice from Manager of such failure; <u>provided</u>, that such thirty (30) day period shall be extended for up to an aggregate of ninety (90) days so long as such Owner is diligently attempting to cure such failure.
- 9.3 <u>Manager Defaults and Owner Remedies</u>. Upon the occurrence of any of the following events, each Owner may exercise such rights and remedies as may be available to it at law or in equity, including the right to terminate this Agreement with respect to such Owner's Facility or the right of the Owners to terminate this Agreement with respect to all Facilities, by written notice to the Manager, provided, that no such notice shall be required for a termination pursuant to clause (c) of this <u>Section 9.3</u>:

- (a) the failure by Manager to make any payment, deposit or transfer required hereunder when due and such failure continues for fifteen (15) days after receipt of written notice from such Owner of such failure;
- (b) the failure of any statement, representation or warranty made by Manager in this Agreement to have been correct in any material respect when made if such failure could reasonably be expected to have a material adverse effect on Manager's ability to perform its obligations under this Agreement;
 - (c) the occurrence of an Act of Insolvency with respect to Manager; or
- (d) the failure of Manager to perform any of its material obligations under this Agreement and such failure continues for thirty (30) days after receipt of written notice from such Owner of such failure; <u>provided</u>, that such thirty (30) day period shall be extended for up to an aggregate of ninety (90) days so long as Manager is diligently attempting to cure such failure.
- 9.4 <u>Termination for Convenience</u>. Each Owner may terminate this Agreement with respect to its Facility for any reason upon (x) sixty (60) days prior written notice to Manager or (y) less than sixty (60) days prior written notice to Manager if such Owner shall pay the Supplemental Termination Payment. Manager may terminate this Agreement with respect to the Facilities for any reason upon sixty (60) days prior written notice to Owner Agent.

9.5 Facility Condition at End of Term; Successor Manager.

Upon expiration or termination of this Agreement with respect to a Facility, Manager shall remove its personnel from such Facility. Manager shall leave such Facility in as good condition as at the date hereof subject to (i) normal wear and tear, (ii) if a Facility has suffered damages covered by insurance, the availability to Manager of the proceeds of such insurance and (iii) changes in condition resulting from Force Majeure Events; provided that Manager is otherwise in compliance with its obligations under this Agreement. All special tools, improvements, inventory of supplies, spare parts, safety equipment (in each case as obtained by or provided by Manager and paid for by Owners during the term of this Agreement), Facilities Records, and any other items furnished under this Agreement will be left at the Facility and will remain the property of Owner without additional charge. Owners shall also have the right, in their sole discretion, to directly assume and become liable for any contracts or obligations that Manager may have undertaken with third parties principally in connection with the Services. Manager shall execute all documents and take all other reasonable steps requested by Owner that may be required to assign to and vest in Owner all rights and obligations, benefits, interests and title in connection with such contracts or obligations. Notwithstanding the foregoing, Manager, at all times, shall be deemed the owner of and shall be permitted to retain or remove all trademarks, logos, trade names and similar proprietary rights of Manager and its Affiliates ("Manager Proprietary Property") and shall disable all connections to Manager's ERP System servicing the Facilities (provided that Manager shall provide hard copies of financial, operational and other information relating principally to the Facilities as reasonably requested by Owners).

- (b) Upon expiration or termination of this Agreement with respect to a Facility, Manager shall, at the request and expense of Owner of such Facility, assist such Owner in arranging for the future performance of the Services by such Owner or the successor to Manager (the "Successor Manager"), provided that Manager shall provide such assistance at no charge to such Owner if the Agreement is terminated pursuant to Section 9.3. Manager shall provide such Owner and the Successor Manager full access to such Facility and to all relevant information, data and records relating thereto reasonably required for taking over the performance of Services for such Facility.
- (c) Promptly after expiration or termination of this Agreement with respect to a Facility, Manager shall deliver to Owner of such Facility or (if so required by such Owner by notice to Manager) to the Successor Manager all property in its possession or under its control owned by such Owner or leased or licensed to such Owner.
- (d) The parties agree that Manager has granted each Owner and Owner's Agent a license to use the name "Pacific Ethanol". Upon expiration or termination of this Agreement, each Owner and Owner Agent as promptly as practicable shall cease using the name Pacific Ethanol and shall change its name to delete any reference to Pacific Ethanol, it being understood that Owner and Owner's Agent may use existing letterhead and similar supplies for a period not to exceed 30 days.

9.6 Termination Payment; Manager Payment.

- 9.6.1 <u>Termination Payment.</u> Promptly after the date of any termination, Manager shall be paid on a pro rata basis (i) the Asset Management Fee earned through the date of termination but not paid (the "<u>Termination Payment</u>"), (ii) if such termination is pursuant clause (y) of Section 9.4 the Supplemental Termination Payment and (iii) all other payments that it is entitled to under this Agreement for the period through the date of termination. Except for the Termination Payment, the Supplemental Termination Payment, if applicable, and such other payments, Owners shall not be liable for any costs incident to termination.
- 9.6.2 <u>Manager Payment</u>. Subject to Article XII, in the event of a termination of this Agreement by an Owner under <u>Section 9.3</u>, such Owner shall be entitled to recover from Manager any damages, costs, fines or penalties such Owner suffers or incurs as a result of any such termination, including the reasonable costs of mobilizing and training a successor Manager, and such Owner hereby releases Manager from any liability in excess thereof. In calculating such reasonable costs, any savings achieved due to the retention by the successor Manager of any or part of the Facility workforce shall be taken into account.

ARTICLE X

INSURANCE

Manager Insurance. Without limiting any of the other obligations or liabilities of Manager under this Agreement, Manager shall at all times carry and maintain or cause to be carried and maintained, the minimum insurance coverage set forth in this Section 10.1:

- (a) Manager shall maintain (i) Workers' Compensation insurance in compliance with the workers' compensation laws of the jurisdictions in which each Facility is located as extended by the Broad Form All States Endorsements, the United States Longshoreman's and Harbor Workers' Coverage Endorsements on an if-any-exposure basis and the Voluntary Compensation Coverage Endorsement, and (ii) Employer's Liability (including Occupational Disease) coverage with limits of not less than \$1,000,000, which shall cover all of Manager's employees engaged in providing services hereunder.
- (b) Manager shall maintain automobile liability insurance for owned (if any), non-owned and hired vehicles with combined single limits for bodily injury/property damage not less than \$1,000,000 per occurrence and containing appropriate no-fault insurance provisions wherever applicable.
- (c) Manager will maintain commercial general liability insurance with a limit for bodily injury/property damage of not less than \$1,000,000 per occurrence and \$2,000,000 in the annual aggregate. Such coverage shall include premises/operations, explosion, collapse and underground property damage, broad form contractual, independent contractors, products/completed operations (including Manager errors and omissions), broad form property damage, sudden and accidental pollution, personal injury and incidental professional liability (if not covered under product/completed operations and if commercially available).
- (d) Manager shall maintain or cause to be maintained umbrella liability insurance providing coverage limits in excess of those set forth in Section (a), (b) and (c) above. The limits of this umbrella coverage shall not be less than \$10,000,000 per occurrence and in the annual aggregate.

The terms and conditions of all insurance policies (including the amount, scope of coverage, deductibles, and self-insured retentions) shall be acceptable in all respects to Owner Agent. All insurance carried pursuant to this Section shall conform to the relevant provisions of this Agreement and be with insurance companies which are rated "A, IX" or better by Best's Insurance Guide and Key Ratings, or other insurance companies of recognized responsibility satisfactory to Owner Agent. Owner Agent shall be furnished with satisfactory evidence that the foregoing insurance is in effect and shall be notified 30 calendar days prior to the cancellation or material change of any such coverage. Coverage for the insurance under Section (c) and (d) above shall be written on a claims made basis provided that if the policy is not renewed, Manager shall obtain for the benefit of Owners an extended reporting period coverage or "tail" of at least three years past the final day of coverage of such policy. Manager shall provide Owner Agent with evidence that such extended reporting period coverage or "tail" has been obtained. Manager agrees to ensure that the insurance policies outlined in this Section require the insurer to waive subrogation against Owners, the Senior Secured Parties and their respective Affiliates together with their respective officers, directors, Affiliates and employees and all such Persons shall be an additional insured as their interests may appear with respect to all policies procured by Manager.

- Owners Insurance. Owners shall ensure, to the extent commercially available, that Manager is named as an additional insured on each insurance policy relating to ownership, operation and maintenance of each Facility which Owners are required to take out and maintain pursuant to the Financing Documents and which are not included within the insurance coverages procured by Manager pursuant to Section 10.1. Owners shall provide Manager with a certified copy of each such insurance policy and shall notify Manager in writing of any changes to such policy from time to time or, before doing so, of the cancellation of any such policy or policies. Any obligation or liability for premiums, commissions, assessments or calls in connection with any insurance policy required under this Section shall be the sole responsibility of Owners except for claims arising from losses due to Manager error and omissions in which case Manager shall be responsible for all deductibles. All policies procured by Owners shall require the insurer to waive subrogation against Manager and its officers, directors, Affiliates and employees and all such Persons shall be named as additional insureds to the extent of their interests; provided, that Manager shall have no interest with respect to business interruption coverage.
- Manager Insurance Premiums and Deductibles. All premiums for insurance coverage procured by Manager pursuant to Section 10.1 shall be for the account of Manager. Manager shall be liable for the payment of deductibles on insurance policies obtained pursuant to Section 10.1.
- Subcontractor Insurance. Before permitting any Manager's subcontractor to perform any services at a Facility, each such subcontractor must provide proof of insurance satisfactory to Owner Agent and obtain a certificate of insurance evidencing that such subcontractor has obtained insurance in such amounts and such risks as is customarily carried by Persons engaged in similar businesses in the same geographic area and from such carriers as are licensed to do business in the jurisdiction in which such Facility is located.

ARTICLE XI

INDEMNIFICATION

- Owners' Indemnity. Each Owner shall defend, indemnify and hold harmless Manager and its Affiliates (and each officer, director, employee, shareholder, partner, member or agent of Manager and its Affiliates) (each, an "Owner Indemnified Person") from and against (i) any and all claims, actions, damages, expenses (including reasonable and documented attorneys' fees and expenses), losses, settlements or liabilities (collectively, "Liabilities") incurred or asserted against any Owner Indemnified Person (a) as a result of any failure on the part of such Owner to perform its obligations under this Agreement or (b) arising out of or in any way connected with the grossly negligent acts or omissions of such Owner and (ii) Liabilities incurred by any Owner Indemnified Person to a third party or asserted against any Owner Indemnified Person by a third party as a result of Manager performing the Services in accordance with the terms of this Agreement.
- Manager's Indemnity. Manager shall defend, indemnify and hold harmless each Owner and its Affiliates (and each officer, director, employee, shareholder, partner, member or agent of each Owner and its Affiliates) (each, a "Manager Indemnified Person") from and against any and all Liabilities incurred or asserted against any Manager Indemnified Person (a) as a result of any failure on the part of Manager to perform its obligations under this Agreement, or (b) arising out of or in any way connected with the grossly negligent acts or omissions of Manager or its subcontractors and Affiliates (other than Owners) in connection with this Agreement.

ARTICLE XII

LIABILITIES OF THE PARTIES

- Maximum Liability of Manager. The total aggregate liability of Manager to the Owners, or of the Owners to the Manager, under this Agreement during the term of this Agreement shall not exceed two million dollars (\$2,000,000). Notwithstanding the foregoing, such limitations on liability shall not apply with respect to (i) the wrongful withholding of payments due hereunder, and (ii) any net loss, damage or liability resulting from or arising out of the gross negligence or willful misconduct of Manager or the Owners, as applicable.
- No Consequential or Punitive Damages. In no event shall either Party be liable to any other Party by way of indemnity or by reason of any breach of contract or of statutory duty or by reason of tort (including negligence or strict liability) or otherwise for any loss of profits, loss of revenue, loss of use, loss of production, loss of contracts or for any incidental, indirect, special or consequential or punitive damages of any other kind or nature whatsoever that may be suffered by such other Party.

ARTICLE XIII

CONFIDENTIALITY

Owner's Confidential Information. Manager agrees to maintain the confidentiality of the Owner Information (as defined 13 1 below), except that Owner Information may be disclosed (a) to Manager's Affiliates and its and their officers, directors, managers, employees, agents and representatives, including accountants, legal counsel and other advisors, but only to the extent such Persons have a need to know such information for purposes of permitting Manager to perform its obligations hereunder (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Owner Information and instructed to keep such Owner Information confidential), (b) to the extent requested by any regulatory authority having jurisdiction over Manager, (c) to the extent required by applicable laws or regulations or by any subpoena or similar legal process, (d) in accordance with any Financing Document, (e) in connection with any suit, action or proceeding relating to this Agreement or the enforcement of rights hereunder, (f) with the consent of Owner Agent, (g) to any potential equity investors in PEI, provided that such potential equity investors execute confidentiality agreements with Owner Agent that are at least as restrictive as the provisions set forth herein or (h) to the extent such Owner Information (1) becomes publicly available other than as a result of a breach of this Article XIII or (2) becomes available to Manager on a non-confidential basis from a source other than an Owner. "Owner Information" means all information received from any Owner, or developed by Manager in the course of performing its obligations under this Agreement, relating to the business and operations of any Owner. Manager shall be considered to have complied with its obligation to maintain the confidentiality of Owner Information if Manager has exercised the same degree of care with respect to the Owner Information as Manager would accord to its own confidential information.

13.2 Manager's Confidential Information. Each Owner agrees to maintain the confidentiality of the Manager Information (as defined below), except that Manager Information may be disclosed (a) to Owner's Affiliates and its and their officers, directors, managers, employees, agents and representatives, including accountants, legal counsel and other advisors, but only to the extent such Persons have a need to know such information (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Owner Information and have agreed to abide and be bound by the provisions hereof), (b) to the extent requested by any regulatory authority having jurisdiction over Owner, (c) to the extent required by applicable laws or regulations or by any subpoena or similar legal process, (d) in accordance with any Financing Document, (e) in connection with any suit, action or proceeding relating to this Agreement or the enforcement of rights hereunder, (f) with the consent of Manager Agent, or (g) to any potential equity investors in, or acquirers or potential acquirers of, NewCo, Owner Agent, any Owner or any Facility, provided that such potential equity investors, acquirers or potential acquirers execute confidentiality agreements with Manager that are at least as restrictive as the provisions set forth herein or (h) to the extent such Manager Information (1) becomes publicly available other than as a result of a breach of this Article XIII or (2) becomes available to Owner on a non confidential basis from a source other than Manager. "Manager Information" means all information received from Manager relating to the business and operations of Manager or any of its Affiliates (other than the Owners, the Owner Agent or NewCo), including without limitation information disclosed pursuant to Section 6.1.3. An Owner shall be considered to have complied with its obligation to maintain the confidentiality of Manager Information if Owner has exercised the same degree of care with respect to the Manager Information as Owner would accord to its own confidential information.

ARTICLE XIV

TITLE, DOCUMENTS AND DATA

- Materials and Equipment. Title to all materials, equipment, supplies, Consumables, spare parts, and other items purchased or obtained by Manager shall pass immediately to and vest in Owners upon the passage of title from the vendor or supplier thereof, provided, however, that such transfer of title shall in no way affect Manager's obligations as set forth in the other provisions of this Agreement.
- Documents. All materials and documents prepared or developed by Manager or its employees, representatives or contractors in connection with a Facility or the performance of the Services hereunder, including all manuals, data, designs, drawings, plans, specifications, reports and accounts but expressly excluding any Manager Proprietary Property, shall become or remain the property of Owners when prepared. All such materials and documents, together with any materials and documents furnished to Manager or to its contractors by Owners, and any and all copies thereof shall be delivered to Owners upon expiration or termination of this Agreement. In addition, all such materials and documents shall be available for review by Owners at all reasonable times during development and promptly upon completion. All such materials and documents required to be submitted for the approval of Owners shall be prepared and processed in accordance in all material respects with the requirements and specifications set forth in the Facility Manuals. However, Owners' approval of materials and documents submitted by Manager shall not relieve Manager of its responsibility to perform its obligations under this Agreement.

ARTICLE XV

FORCE MAJEURE

- 15.1 Events Constituting Force Majeure. As used herein, "Force Majeure Event" means any cause(s) which render(s) a Party wholly or partly unable to perform its obligations under this Agreement (other than obligations to make payments when due), and which are neither reasonably within the control of such Party nor the result of the fault or negligence of such Party, and which occur despite all reasonable attempts to avoid, mitigate or remedy. Force Majeure Events may include acts of God, war, riots, civil insurrections, cyclones, hurricanes, floods, fires, explosions, earthquakes, lightning, storms, chemical contamination, epidemics or plagues, acts or campaigns of terrorism or sabotage, blockades, embargoes, accidents or interruptions to transportation, trade restrictions, acts of any Governmental Authority after the date hereof, strikes and other labor difficulties, mechanical breakdowns, and other events or circumstances beyond the reasonable control of the claiming Party and which otherwise satisfy the criteria set forth above for a Force Majeure Event.
- Effect. A Party claiming relief as a result of a Force Majeure Event shall give the other Parties written notice within five Business Days of becoming aware of the occurrence of the Force Majeure Event, or as soon thereafter as practicable, describing the particulars of the Force Majeure Event, and will use reasonable efforts to remedy its inability to perform as soon as possible. If the Force Majeure Event (including the effects thereof) continues for fifteen consecutive days, the affected Party shall report to the other Parties the status of its efforts to resume performance and the estimated date thereof. If the affected Party was not able to resume performance prior to or at the time of the report to the other Parties of the onset of the Force Majeure Event, then it will report in writing to the other Parties when it is again able to perform. If a Party fails to give timely notice, the excuse for its non-performance shall not begin until notice is given.
- 15.3 <u>Limitations</u>. Any obligation(s) of a Party (other than an obligation to make payments when due) may be temporarily suspended during any period such Party is unable to perform such obligation(s) by reason of the occurrence of a Force Majeure Event, but only to the extent of such inability to perform, provided, that:
- (a) the suspension of performance is of no greater scope and of no longer duration than is reasonably required by the Force Majeure Event; and
 - (b) the Party claiming the occurrence of the Force Majeure Event bears the burden of proof.

ARTICLE XVI

MISCELLANEOUS PROVISIONS

- Assignment. No Party shall assign this Agreement or any of its rights or obligations hereunder without first obtaining the prior written consent of each other Party and the Administrative Agent; provided, that each Owner shall be entitled to assign its rights hereunder (as collateral security or otherwise) for financing purposes (including a collateral assignment) without the consent of Manager.
- Sale of Facilities. The Parties acknowledge that Owners may consider a sale of one or more of the Facilities to one or more purchasers. In connection with any sales process, if requested by Owners, Manager will make all records and other documents under its control regarding the Facilities available to potential purchasers and will provide such other services and assistance as Owners may reasonably request. Manager will not restrict or limit its personnel who are, or have been, working on-site at any Facility from joining any company or Affiliate of a company that acquires ownership of a Facility.
- 16.3 <u>Cooperation in Financing</u>. Manager shall use its reasonable efforts to execute, acknowledge and deliver any and all further documents and instruments, and to take any other actions, which may be necessary to satisfy the reasonable requests of any Senior Secured Party or prospective Senior Secured Party in connection with the financing of each Facility, including delivering to the Collateral Agent a customary consent to the assignment by Owners of its rights under this Agreement to the Collateral Agent.

16.4 Access

- 16.4.1 Owners. Each Owner and its agents and representatives shall have access at all times to its Facility, all Facility operations and any documents, materials and records and accounts relating to the Facility operations. Upon request of such Owner, and its agents and representatives, Manager shall make available to such Persons on site and provide them with access to any data and all logs relating to such Facility.
- 16.4.2 <u>Senior Secured Parties</u>. As and to the extent required under the Financing Documents, the Senior Secured Parties and their agents and representatives, at all reasonable times (and upon reasonable prior notice), shall have access to each Facility, all Facility operations and any documents, materials, records and accounts relating to Facility operations for purposes of inspection and review. Manager shall have the right to have its personnel accompany such Senior Secured Parties and such agents and representatives during such access.
- 16.4.3 <u>Cooperation.</u> During any such inspection or review of any Facility, Owners, the Senior Secured Parties and their agents and representatives shall comply with all of Manager's safety and security procedures, and Owners, the Senior Secured Parties and their agents and representatives shall conduct such inspection and reviews in such a manner as to cause minimum interference with Manager's activities. Manager also shall cooperate with Owners in allowing other visitors access to each Facility under conditions as Owners shall designate.

- 16.5 <u>Not for Benefit of Third Parties.</u> Except as otherwise expressly provided in Articles XI and XVI, this Agreement and each and every provision hereof are for the exclusive benefit of the Parties hereto and is not for the benefit of any third party.
- Amendments. No Party hereto shall be bound by any amendment, supplement, waiver or modification of any term hereof unless such Party and the Administrative Agent shall have consented thereto in writing.
- 16.7 <u>Survival</u>. Cancellation, expiration or earlier termination of this Agreement shall not relieve the Parties of obligations that by their nature should survive such cancellation, expiration or termination, including remedies, limitations on liability, promises of payment, indemnity and confidentiality. Without limiting the generality of the foregoing, the following provisions of this Agreement shall survive cancellation, expiration or earlier termination of this Agreement: Articles V, VIII, IX, XI, XIII and XVI and the limitations on liabilities set forth in Article XII.
- 16.8 <u>No Waiver</u>. No delay or failure on the part of any Party in exercising any rights hereunder, and no partial or single exercise thereof, shall constitute a waiver of such rights or of any other rights hereunder.
- Notices. All notices or other communications which are required or permitted hereunder shall be in writing and shall be deemed sufficiently given (a) upon delivery, if delivered personally, (b) the day the notice is received, if it is delivered by overnight courier or certified or registered mail, postage prepaid, or (c) upon the effective receipt of electronic transmission, facsimile, telex or telegram (with effective receipt being deemed to occur upon the sender's receipt of confirmation of successful transmission of such notice or communication), to the addresses set forth below or such other address as the addressee may have specified in a notice duly given to sender as provided herein:

If to Manager: Pacific Ethanol, Inc.

400 Capitol Mall, Suite 2060 Sacramento, CA 95814 Attention: Neil Koehler

Telephone: (916) 403-2123 Facsimile: (916) 446-3936

With a copy to: Pacific Ethanol, Inc.

400 Capitol Mall, Suite 2060 Sacramento, CA 95814 Attention: General Counsel Telephone: (916) 403-2130 Facsimile: (916) 403-2785 If to Owners: Pacific Ethanol Holding Co. LLC

c/o JT Miller Group LLC

777 Campus Commons Road #200

Sacramento, CA 95825 Attention: John Miller

Telephone: (916) 565-7422 Facsimile: (916) 565-7423

16.10 Representations and Warranties.

16.10.1 Manager's Representations and Warranties. Manager represents and warrants to Owners, as of the date hereof, as follows:

16.10.1.1 <u>Due Formation.</u> Manager (a) is a corporation duly formed and validly existing under the laws of the State of Delaware, (b) has the requisite power and authority to own its properties and carry on its business as now being conducted and currently proposed to be conducted and to execute, deliver and perform its obligations under this Agreement, and (c) is qualified to do business in every jurisdiction in which failure so to qualify could be reasonably be expected to have a material adverse effect on Manager's ability to perform its obligations hereunder.

16.10.1.2 <u>Authorization; Enforceability.</u> Manager has taken all action necessary to authorize it to execute, deliver and perform its obligations under this Agreement. This Agreement constitutes a legal, valid and binding obligation of Manager enforceable in accordance with its terms.

16.10.1.3 No Conflict. The execution, delivery and performance by Manager of this Agreement does not and will not (a) violate any Law applicable to Manager, (b) result in any breach of Manager's constituent documents or (c) conflict with, violate or result in a breach of or constitute a default under any agreement or instrument to which Manager or any of its properties or assets is bound or result in the imposition or creation of any lien or security interest in or with respect to any of Manager's property or assets, other than in each case any such violations, conflicts, breaches or impositions which could not be reasonably be expected to have a material adverse effect on Manager's ability to perform its obligations hereunder.

16.10.1.4 <u>No Authorization</u>. No authorization or approval or other action by, and no notice to or filing with, any Governmental Authority (other than those which have been obtained) is required for the due execution, delivery and performance by Manager of this Agreement, other than any such authorizations, approvals or actions the failure of which to obtain could not be reasonably be expected to have a material adverse effect on Manager's ability to perform its obligations hereunder.

16.10.1.5 <u>Litigation</u>. Manager is not a party to any legal, administrative, arbitration or other proceeding, and, to Manager's knowledge, no such proceeding is threatened, which could be reasonably be expected to have a material adverse effect on Manager's ability to perform its obligations hereunder.

16.10.2 <u>Owners' Representations and Warranties.</u> Each Owner represents and warrants to Manager, as of the date hereof, as follows:

16.10.2.1 <u>Due Formation</u>. Such Owner (a) is a limited liability company duly formed and validly existing under the laws of the State of Delaware, (b) has the requisite power and authority to own its properties and carry on its business as now being conducted and currently proposed to be conducted and to execute, deliver and perform its obligations under this Agreement, and (c) is qualified to do business in every jurisdiction in which failure so to qualify could be reasonably be expected to have a material adverse effect on such Owners' ability to perform its obligations hereunder.

Authorization; Enforceability. Such Owner has taken all action necessary to authorize it to execute, deliver and perform its obligations under this Agreement. This Agreement constitutes a legal, valid and binding obligation of such Owner enforceable in accordance with its terms.

16.10.2.3 No Conflict. The execution, delivery and performance by such Owner of this Agreement does not and will not (a) violate any Law applicable to such Owner, (b) result in any breach of such Owner's constituent documents or (c) conflict with, violate or result in a breach of or constitute a default under any agreement or instrument to which such Owner or any of its properties or assets is bound or result in the imposition or creation of any lien or security interest in or with respect to any of such Owner's property or assets, other than in each case any such violations, conflicts, breaches or impositions which could not be reasonably be expected to have a material adverse effect on such Owner's ability to perform its obligations hereunder.

16.10.2.4 No Authorization. No authorization or approval or other action by, and no notice to or filing with, any Governmental Authority (other than those which have been obtained) is required for the due execution, delivery and performance by such Owner of this Agreement, other than any such authorizations, approvals or actions the failure of which to obtain could not be reasonably be expected to have a material adverse effect on such Owner's ability to perform its obligations hereunder.

16.10.2.5 <u>Litigation</u>. Such Owner is not a party to any legal, administrative, arbitration or other proceeding, and, to such Owner's knowledge, no such proceeding is threatened, which could be reasonably be expected to have a material adverse effect on such Owner's ability to perform its obligations hereunder.

- 16.11 <u>Counterparts and Execution</u>. This Agreement may be executed in any number of counterparts and each such counterpart shall be deemed to be an original instrument, but all such counterparts together shall constitute one and the same agreement.
- 16.12 <u>Governing Law.</u> This Agreement shall be governed by and construed in accordance with the laws of the State of New York without regard to the principles of conflicts of laws thereof.
- 16.13 <u>Entire Agreement</u>. This Agreement contains the entire agreement among the Parties with respect to the subject matter hereof and supersedes all prior agreements, negotiations and understandings among the Parties with respect to such subject matter. Nothing in this Agreement shall be construed as creating a partnership or joint venture among the Parties.

- Severability. In the event any one or more of the provisions contained in this Agreement should be held invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein shall not in any way be affected or impaired thereby. The Parties shall endeavor in good-faith negotiations to replace the invalid, illegal or unenforceable provisions with valid provisions, the economic and practical effect of which comes as close as possible to that of the invalid, illegal or unenforceable provisions.
- 16.15 <u>Successors and Assigns</u>. This Agreement shall be binding upon and shall inure to the benefit of the Parties hereto and their respective successors and permitted assigns.
- Owner Agent. Each Owner hereby appoints and authorizes Pacific Holding, and Pacific Holding hereby accepts such appointment, as such Owner's Owner Agent to act as agent on such Owner's behalf and to make any representations or certifications, deliver and receive any notices or other communications, and otherwise represent and act on behalf of such Owner under this Agreement, and to comply with all covenants, conditions and other provisions of this Agreement required to be satisfied by Owner Agent. Each Owner hereby acknowledges and agrees that it will be bound by any action or inaction taken by Owner Agent as if such action or inaction had been taken by such Owner.
- 16.17 <u>Independent Contractor</u>. Manager shall be an independent contractor with respect to the performance of the Services and its other obligations hereunder. Neither Manager nor its employees or other agents employed in the Services shall be deemed to be agents of Owners or to have assumed any other obligations with respect to any Person, except to the extent of the obligations expressly created hereunder pursuant to the authority granted to Manager under this Agreement.
- 16.18 <u>Captions; Appendices.</u> Titles or captions of sections contained in this Agreement are inserted only as a matter of convenience and for reference, and in no way define, limit, extend, describe or otherwise affect the scope of meaning of this Agreement or the intent of any provision hereof. All appendices attached hereto shall be considered a part hereof as though fully set forth herein.
- 16.19 <u>No Novation</u>. The terms and conditions of the Prior Agreement are amended as set forth in, and restated in their entirety and superseded by, this Agreement. Nothing in this Agreement shall be deemed to work a novation of any obligation under the Prior Agreement.
- 16.20 <u>Several Liability</u>. For the avoidance of any doubt, the obligations of each Owner and the Owner Agent hereunder, shall be several not joint.

[THE REMAINDER OF THIS PAGE HAS BEEN LEFT BLANK INTENTIONALLY]

IN WITNESS WHEREOF, this Second Amended and Restated Asset Management Agreement has been duly executed by the parties hereto as of the date first written above.

PACIFIC ETHANOL, INC.

By: /s/ Neil M. Koehler
Name: Neil M. Koehler
Title: Chief Executive Officer
PACIFIC ETHANOL COLUMBIA, LLC
By:
Name:
Title:
PACIFIC ETHANOL MADERA LLC
TACIFIC ETHANOE MADERA EEC
Ву:
Name:
Title:
PACIFIC ETHANOL STOCKTON, LLC
D
By:
Name: Title:
Title:
PACIFIC ETHANOL MAGIC VALLEY, LLC
,
By:
Name:
Title:
DI CIVILO DIVILI VICI VICI DIVICI CO LI C
PACIFIC ETHANOL HOLDING CO. LLC
D
By:
Name: Title:
THE:

Signature Page to Asset Management Agreement

APPENDIX A-1

SCOPE OF ASSET MANAGEMENT SERVICES AND ASSET PRESERVATION SERVICES

A. General Approach.

With respect to each Facility, Manager shall perform, or cause to be performed, all tasks necessary to manage and preserve such Facility in accordance with the Agreement, including:

- Staffing Headcount, Task assignment, Scheduling
- Management Methods, Procedures, Safety, Consumables/Materials Procurement
- Maintenance Management Methods, Procedures, Safety
- Subcontract Management
- Administrative Manager Payroll
- Training
- Security
- Permit Compliance

B. <u>Facility Monitoring</u>.

Manager will provide usual and customary monitoring including the following:

- (a) Perform and record periodic operational checks and tests of equipment in accordance with the equipment manufacturer's specifications.
- (b) Maintain logs, records and reports for operation of the Facility.
- (c) Inform Owners as far in advance as possible when the facility does not have the capability to receive corn or store Ethanol or wet distillers grains.

Each Facility will be monitored on a continuing basis so that data are distributed quickly and accurately to the Facility staff and to Owners.

C. <u>Layup Preservation Services</u>.

Manager will provide usual and customary services including the following:

- (a) Adhere to all regulatory requirements including management of all applicable permits including the Facility Storm Water Pollution Plan and the Facility air permits to include leak detection and repair inspections.
- (b) Clean and lay-up equipment to protect from environmental factors including rust and mold.
- (c) Maintain a regular program for equipment preservation to include equipment rotation and inspection.

- (d) Monitor plant fire protection system and alarms.
- (e) Maintain plant readiness for potential re-start or sale.
- (f) Maintain site to include lawn and garden maintenance.
- (g) Perform site security services to prevent theft, vandalism, etc.

D. Facility Restart.

If a Facility restart is requested by an Owner, Manager will provide usual and customary services including the following:

- (a) Interview, select, hire and train necessary operating personnel.
- (b) Make ready all equipment.
- (c) Order necessary start-up supplies.
- (d) Communicate re-start activities to appropriate parties including city and county officials, suppliers, vendors, etc.
- (e) Obtain all necessary permits and conduct all testing necessary to satisfy the terms and conditions of any such permit.

E. <u>Administrative Support.</u>

Manager will provide usual and customary support including the following:

1. General.

Manager will develop and implement a comprehensive and specific administrative procedures including:

- (a) Maintain an effective Facility work force through proper hiring, training, administration and compensation.
- (b) Procure labor, materials and services.
- (c) All replacement spare parts for those taken out of inventory with either an identical part or replacement part(s) manufactured by the original equipment manufacturers, or where practical and economical, refurbish (or have refurbished) spare parts to allow their reuse.
- (d) Provide information and other reasonable assistance at Owners' request, regarding operation and maintenance of the Facility.

- (e) Maintain accurate Facility cost ledger and accounting records regarding the materials and services provided in accordance with generally accepted accounting principles, consistently applied.
- (f) Cooperate with Owners' independent certified public accountant to perform annual financial audits for all cost reimbursable services provided for the Facility.
- (g) Communicate with: (i) the corn suppliers or transporters, as the case may be, under each Corn Supply Agreement with respect to the amount of corn Owners procures from such corn suppliers or transporters; and, (ii) Owners with respect to all other matters.
- (h) Procure Consumables.

2. Facility Safety Program.

To ensure the safety of all employees and personnel working in or near the Facility, Manager will establish and implement a safety plan which conforms to federal and any state or local regulations. Key components of the plan will include:

- (a) <u>Lock Out/Tag Out</u>. Only trained and authorized employees shall perform work on equipment that requires Lock out/tag out (LOTO). Employees shall follow written LOTO procedures when performing such work.
- (b) <u>Confined Spaced Entry</u>. Only trained and authorized employees shall perform work on any equipment that requires the employee to enter a permit required confined space. Employees shall follow written confined space entry procedures.
- (c) <u>Hot Work</u>. Only trained and authorized employees shall perform work on any equipment that requires the employee to obtain a hot work permit. Employees shall follow written hot work procedures.
- (d) <u>Responsibility</u>. Operations and duties shall be performed only by duly authorized employees, who shall be held responsible for their actions.
- (e) <u>Facility Safety Inspections</u>. Employees shall inspect on a routine basis; operating conditions, equipment and the general workplace to prevent a potential hazard to personnel and equipment.
- (f) <u>Records</u>. Employees required to keep logs and records shall keep them current and accurately. Abnormal or special conditions shall be called promptly to the attention of the proper supervisor and logged. Shift employees shall familiarize themselves with all activities within their jurisdiction that have taken place during the preceding shift.

3. Facility Security.

Manager will adhere to the written Facility Security Plan and Facility staff will be trained in its requirements. Facility staff and all visitors will be required to adhere to the plan to ensure security of the Facility in normal as well as emergency situations.

F. <u>Maintenance Management Program.</u>

Manager will provide usual and customary services including the following:

The prime objective of the maintenance management program will be to maximize equipment reliability and availability and minimize maintenance costs. The ultimate goal is to maximize Facility profitability and to maintain, as closely as possible, given normal mechanical degradation, to the original design performance of all major pieces of equipment.

The key elements of the maintenance program are:

- Preventive maintenance
- Predictive maintenance
- Corrective maintenance
- Operational downtime management
- Spare parts inventory control

1. <u>Preventive Maintenance</u>.

Preventive maintenance will be done by the Facility staff, consisting of periodic inspection and adjustment of equipment, to avoid deterioration to a point at which the equipment will not start, run, or operate efficiently or reliably. Some preventive maintenance can be accomplished while the unit is shutdown or running.

Preventive maintenance schedules will be included in the computerized program and calibrated to an overall Facility schedule. This schedule will alert maintenance crews, providing daily, weekly, monthly and annual forecasting of preventive maintenance activities. Spare parts considerations, such as lead time, can also be correlated.

Preventive maintenance intervals are directed by two measures, calendar time and unit operating parameters (vibration measurements, operating temperatures, amp loading, etc). Items scheduled by calendar time will be reviewed on a daily, weekly, monthly or yearly basis. As a rule, preventive maintenance work will be scheduled to result in minimal interference with Facility operations.

2. Predictive Maintenance.

A predictive maintenance program will center on Manager's ability to track vital trend information. Observing critical variables over time will yield valuable information regarding equipment deterioration. The preventive maintenance frequency may be increased or decreased, depending on the condition of the equipment found during a predictive maintenance inspection.

Use of the preventive and predictive maintenance schedules, in conjunction with the monitoring of operational data that reflect equipment condition, will result in high reliability and reduced maintenance costs.

3. <u>Corrective Maintenance</u>.

The corrective maintenance activities will be aimed at avoiding repeat failures. Discussion meetings will be held to review failures which caused major repairs or shutdowns. Facility staff will review the conditions preceding the failure and, if possible, determine the exact cause, and make findings available to maintenance and operating personnel.

4. Shutdown Management.

Shutdowns for equipment repair or replacement will be aggressively managed to minimize downtime. Advanced planning, work packages, shutdown schedules and other project management methods will be used to allocate Facility resources to produce efficient shutdowns.

During the pre-shutdown phase, the Facility staff and the major equipment manufacturers will cooperate closely to conduct the planned inspections in a minimum amount of time at a reasonable cost. This advance planning might begin anywhere from six months to a year before the shutdown, depending on the need for and the availability of major equipment components.

All pre-shutdown and shutdown related activities will be broken down into work packages or discrete elements which are more easily tracked over the course of the shutdown and provide a better means to manage the shutdown schedule.

A scheduling program, such as Microsoft Project, using the critical path method will itemize various work packages, organize them and calculate the effect any work package has on the overall shutdown length. This program will provide a reporting tool that allows the Facility staff to create easy-to-understand shutdown schedules and reports showing manpower and equipment resources, usage profiles and problems leading to schedule slippage.

5. Spare Parts Inventory Control.

The Facility staff will implement a spare parts inventory control system designed to minimize tied-up capital, yet not jeopardize Facility contractual commitments by risking extended equipment downtime.

The Facility staff will establish an adequate spare parts inventory by studying manufacturers' recommendations, reviewing past experience, and evaluating parts availability and delivery time from the various vendors, taking into account acceptable equipment downtime. The spares inventory will be managed so that parts are replaced when used, maintenance records are kept current, and the inventory is updated regularly to reflect changing Facility requirements.

Prior to outages and inspections, the Facility staff will order parts prone to normal wear and tear that are identified as needing to be replaced through the preventive/predictive maintenance programs. Furthermore, by maintaining close contact with the original equipment suppliers, the Facility staff will ensure that they purchase the latest revision parts.

G. Corporate Support.

Manager will provide usual and customary support including the following:

Accounting & Finance

- Invoice/AR/Collection
- Purchasing/AP
- Inventory Tracking
- Hedge Accounting
- Utilize Software for Reporting
- Quarterly and Year End Reports Audited if required by an Owner

Treasury & Cash Management Loan Compliance Insurance Property Tax Preparation & Appeals Corporate Tax Preparation

Administration Services

Human Resources

- Payroll Administration (ADP)
- Benefits Administration
- Employee Relations
- Recruit/Hire/Terminate Personnel
- Training Obligations

IT

- Operate ERP System
- Implement and Operate Communications Systems
- Maintain Desktop Equipment and Services

Facilities Management

- Lease Administration
- Office Supplies
- Administrative Support

Operations Services

Plan, monitor and report plant activities.

- Activity planning for lay-up and preservation to minimize asset risk and cost.
- Monitoring/auditing of activities to ensure compliance with regulations/loan provisions.
- Manage personnel resources during activities.

Plan, monitor and report SH&E activities.

- Numerous regulatory issues to manage during plant layup and preservation.
- Title V activities for California plants.
- CA low NOx requirement being instituted.
- Regulatory reports management.

Licensing and Reporting

- EPA RFS program registration.
- DOT hazardous materials registration.
- DOE monthly reports.
- USDA grain warehouse operator reports.
- Others as may be required by any Permit condition or applicable law.

Legal Services

Litigation/Dispute Management

- Claims in process

Contract Management

- Documentation tracking systems

Compliance Oversight

- Operating and environmental regulatory oversight

H. Reporting Obligations and Administration of Financing Documents.

Without limiting the Manager's obligations under Sections 3.1.1 and 3.11, the Manager shall provide Owner with the various certifications and reports, budgets and other submissions required to be provided to Owner pursuant to this Agreement or to the Administrative Agent or Lenders pursuant to the Credit Agreement as set forth below. In all cases, Manager shall furnish drafts of the various reports, budgets and other submissions as soon as reasonably available in advance of delivery of the final report, budget or other submission, so as to keep Owner informed as to the progress and substance of the work in question. Where certifications are required to be provided by the Owner Agent to the Administrative Agent, the Manager shall make the same such certification to the Owner, provided that with respect to items involving potential Events of Default, Environmental Claims or Pension matters, such certification need only be provided if so requested by Owner Agent. Capitalized terms used below have their respective meanings as set forth in the Credit Agreement.

	Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification/ Approval	Date Due to Administrative Agent	Reference to Reporting Format	Reference to Credit Agreement	Date Due to Pacific Holding
1		Forecast	Prepare a rolling cash flow forecast ending on the earlier of 13 weeks after the cash flow is delivered and the Maturity Date with the same level of detail as the then current budget.	No	2nd business day each week			As soon as reasonably possible
2		Expenses	Create a report setting forth, in a form and in sufficient detail satisfactory to the Administrative Agent, a comparison of actual receipts and expenses to budgeted receipts and expenses for the preceding week, within the current Budget.		2nd business day each week			As soon as reasonably possible

	Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification/ Approval	Date Due to Administrative Agent	Reference to Reporting Format	Reference to Credit Agreement	<u>Pacific</u>
3	Monthly	Financial Statements	Prepare a report setting forth, in each case in a form and in sufficient detail satisfactory to the Administrative Agent, balance sheets for each borrower for the current month and income statements, cash flows and profit and loss statements for such month and year-to-date. The report is to be certified by an Authorized Officer of the Borrowing Agent for each financial covenant set forth in Section 7.02(w) that the Borrowers are in full compliance with each such covenant (or, if any of such certifications cannot be given, state in reasonable detail the necessary qualifications to such certifications)	be certified as correct	When available and within 25 days after the end of each month		7.03 (f)	As soon as reasonably possible, and in any event no later than within 24 days after the end of each month

	Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification/ Approval	Date Due to Administrative Agent	Reference to Reporting Format	Reference to Credit Agreement	Date Due to Pacific Holding
4	Monthly	Operating Statements	Prepare and deliver to the Administrative Agent, an Operating	,		Exhibit 7.03 (o)	()	As soon as reasonably
				•	end of each	7.03 (0)		possible, and
					month			in any event
			•	and				no later than
				complete by				within 24 days
			f	an				after the end
			•	Authorized				of each month
			showing in detail all actual expenses	Officer of				
			relating to the operation and	the				
			maintenance of each Plant compared	Borrowers'				
			to the budgeted expenses for the	Agent				
			period; information showing the					
			amounts of ethanol and other					
			products produced in the period;					
			ethanol and Distillers Grain sold per					
			Agreements; amount of all other					
			sales including ethanol and/or					
			Distillers Grains, with the amount,					
			an explanation of the sale and					
			identification of the purchaser for					
			each Plant.					

	Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification/ Approval	Date Due to Administrative Agent	Reference to Reporting Format	Reference to Credit Agreement	<u>Pacific</u>
5	Quarterly	Budgets	Prepare a budget containing a rolling cash flow forecast that projects cash flows for each Plant in detail on an aggregate basis for the Project for the period starting on the current Period Start Date and ending on the earlier of 13 weeks after the current Period Start Date and the Maturity Date. Each forecast is effective upon approval of the Administrative Agent and the Required Lenders of the Revolving Loan Class. Also include a worksheet detailing all changes in material assumptions used while preparing the Budget with a line item for each expense category requested with amount as requested.		7 days before the first day of the Fiscal Quarter		(k) (i) (ii)	As soon as reasonably possible, and in any event no later than 21 days before the first day of the Fiscal Quarter, with any requested revisions thereto provided no later than 14 days before the first day of the Fiscal Ouarter
6	Quarterly	Financial Statements		executed by an	45 days after the end of the first three quarters of the year		7.03 (a) 7.03 (c)	As soon as reasonably possible, and in any event no later than 44 days after the end of the first three quarters of the year

	Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification/ Approval	Date Due to Administrative Agent	Reference to Reporting Format	Reference to Credit Agreement	Date Due to Pacific Holding
7	Annually	Ü	Prepare an Initial Annual Forecast detailing projected requirements for Operation and all Maintenance and Maintenance Capital Expenses for the year on a monthly basis for each Plant.	No	60 days before each year-end			As soon as reasonably possible, and in any event no later than 30 days before such Budget is due to the Administrative Agent, with any requested revisions thereto provided no later than 15 days before such Budget is due to the Administrative Agent
8	Annually	Statements	and cash flows of Pacific Holding with an unqualified opinion of the Auditors for the periods in conformity with GAAP. The statements are to be certified by an		90 days after the end of each year			As soon as reasonably possible, and in any event no later than within 88 days after the end of each year
9	Trigger Event		A statement of an Authorized Officer of the Borrowers' Agent is to be submitted that includes details of the event and the actions Borrowers have taken and propose to take.	an Authorized	As soon as possible or within 5 days after Event of Default		7.03 (h)	As soon as reasonably possible

Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification/ Approval	Date Due to Administrative Agent	Reference to Reporting Format	Reference to Credit Agreement	Date Due to Pacific Holding
Trigger Event		Agent and is to include details of any litigation or proceeding pending against any Borrower, the	Authorized Officer of the Borrowers'	As soon as possible or within 5 days after Borrower obtains knowledge of Event		7.03 (i)	As soon as reasonably possible
Trigger Event		Following receipt of a statement of an Authorized Officer of the Borrowers' Agent relating to any event of default, copies of all material notices or documents received by a Borrower relating to the Project Documents including allegations of any breach or default and any notice regarding or requesting consent to, any assignment, termination, modification, waiver or variation thereof.		Promptly after receipt		-	Promptly after receipt
00	of Material Risk	related to the pension plan. A copy of any notice or filing that is	executed certificate of an Authorized Officer of the Borrowers'	business days of a Borrower		7.03 (k)	As soon as reasonably possible

Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification/ Approval	Date Due to Administrative Agent	Reference to Reporting Format	Reference to Credit Agreement	Date Due to Pacific Holding
Event	Subsequent to an Event	notification of its final decision finding liability and date the liability must be paid, then a copy of the letter along with a certificate	executed certificate by the President or CFO of the Borrower	Borrower receiving a			As soon as reasonably possible
22	Claims	must also be issued after any Borrower is aware of the following conditions: receipt of any written communication from the Government or any other Person alleging that an Environmental Affiliate is not in	executed certificate of an Authorized Officer of the Borrowers' Agent	days after any Borrower is			As soon as reasonably possible

	Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification/ Approval	Date Due to Administrative Agent	Reference to Reporting Format	Reference to Credit Agreement	Date Due to Pacific Holding
15		Inspection of Records	Keep complete records and make available for inspection upon notice; all non-privileged correspondence, investigations, studies, sampling and testing conducted, and all remedial actions taken by any party relating to Environmental Concerns in order for Consultants, Lenders and their employees (in an Event of Default), and the Administrative Agent to review.	No	Upon request of inspection		7.03 (n)	Upon request of inspection
16	At Request of Lender		0	No	Consistently		7.01 (i)	Consistently
17	Trigger Event	Auditor Correspondence	Submit upon receipt, copies of any detailed audit reports, management letters or recommendations that have been submitted to any Borrower by the Auditors.		Submit upon receipt		7.03 (g)	Submit upon receipt
18	Trigger Event	Documents	Provide complete copies of each Necessary Project Approval requested and all other information requested by the Administrative Agent or any Lender through the Administrative Agent.		Submit upon request		7.03 (p) 7.03 (q)	Submit 3 days prior to submission to Administrative Agent

	Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification/ Approval	A dministrative	Reporting	to Credit	<u>Pacific</u>
19	Trigger	Insurance	Obtain and maintain complete	No	Annually	Schedule	7.01 (h)	Annually
	Event		copies of each Plant's executed insurance policies (including		upon renewal	7.01 (h)		upon renewal
			insurance required pursuant to					
			the Transaction Document), and					
			include amounts of insurance					
			listed and described on Schedule					
			7.01(h) of the Credit Agreement					
			upon renewal on an annual basis.					

AMA Reporting Requirements

	Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification/ Approval	Date Due to Administrative Agent	Reference to Reporting Format	Reference to Credit Agreement	Date Due to Pacific Holding
1	Monthly		● With respect to each Facility that is in Cold Shutdown, as soon as available and in any event within 25 days following the end of each calendar month, Manager shall submit to Owner Agent a summary report in the form attached hereto as Appendix C, which report shall include, with respect to each Facility, a numerical and narrative assessment in respect of such month of (i) the Facility's compliance with each category in the Budget, (ii) plant availability, (iii) cash receipts and disbursements including balances in the Accounts, (iv) major maintenance activity, (v) material casualty losses (whether or not covered by insurance), (vi) disputes with any contractor, materialman, supplier or other Person and any related claims against any Owner, (vii) compliance with governmental permits, and (viii) a comparison of figures to corresponding figures provided in the prior month.		N/A		of AMA	25 days following calendar month
2	Quarterly	Report	4	No	N/A			25 days following calendar quarter

	Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification/ Approval	Date Due to Administrative Agent	Panartina	Reference to Credit Agreement	<u>Pacific</u>
3	Monthly	Operating	With respect to each Facility that is not	No	N/A		6.1.2 of	25 days
		Report	in Cold Shutdown, as soon as available				AMA	following
			and in any event within 25 days				Agreement	calendar month
			following the end of each calendar					
			month, Manager shall submit to Owner					
			Agent a summary report in the form					
			attached hereto as Appendix D, which					
			report shall include, with respect to					
			each such Facility, a numerical and					
			narrative assessment in respect of such					
			month of (i) each such Facility's					
			compliance with each category in the					
			Budget, (ii) ethanol and WDG					
			production and delivery, (iii) corn					
			deliveries and use, (iv) plant					
			availability, (v) cash receipts and					
			disbursements including balances in					
			the Accounts, (vi) major maintenance					
			activity, (vii) material casualty losses					
			(whether or not covered by insurance),					
			(viii) disputes with any contractor,					
			materialman, supplier or other Person					
			and any related claims against the					
			Owner of each such Facility,					
			(ix) compliance with governmental					
			permits, and (x) a comparison of					
			figures to corresponding figures					
			provided in the prior month.					

Corn Procurement, Ethanol and Distillers Grains Marketing Agreements' Reporting Requirements

	Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification/ Approval	Date Due to Administrative Agent	Reporting	to Credit	Date Due to Pacific Holding
1	Monthly	Corn Transaction Report	Within 30 days after each Monthly Date occurring after the Effective Date, PAP shall deliver to the Owner a written summary of the Bilateral Transactions which were entered into or performed, in whole or part, during the month ending on such Monthly Date.	No	N/A		3.4 of Corn Procure- ment Agreement	
2	Monthly – 15 days prior to month end	Projected Corn Requirements	On or before the date that is fifteen days prior to the end of a calendar month, Owner shall provide PAP with a forecast of its projected monthly corn requirements for the following month.	No	N/A		3.5 (d) of Corn Procure- ment Agreement	
3	Monthly	Benchmarking	PAP shall furnish Owner on a monthly basis a report benchmarking its performance in accordance with Exhibit D to the Corn Procurement Agreement.	No	N/A		3.6 of Corn Procure- ment Agreement	

	Frequency	Requirement	Requirement Detail of Report to Administrative Agent	Additional Certification Approval	Date Due to Administrative Agent	Reference to Reporting Format	Reterence	Date Due to Pacific Holding
4	Monthly		Kinergy shall furnish Owner on a monthly basis a report benchmarking its performance in accordance with Exhibit D to the Ethanol Marketing Agreement.	No	N/A		2.7 of Ethanol Marketing Agreement	
5	Monthly		PAP shall furnish Owner on a monthly basis a report benchmarking its performance in accordance with Exhibit C to the Distillers Grains Marketing Agreement.	No	N/A		2.7 of Distillers Grains Marketing Agreement	

APPENDIX A-2

SCOPE OF O&M SERVICES

A. General Approach.

Manager shall perform all tasks necessary to operate and maintain the Facilities not in Cold Shutdown in accordance with the Agreement, including:

- Staffing Headcount, Task assignment, Scheduling
- Operations Management Methods, Procedures, Safety, Consumables/Materials Procurement
- Maintenance Management Methods, Procedures, Safety
- Subcontract Management
- Administrative Manager Payroll
- Training
- Security
- Permit Compliance

B. Operations Management.

1. <u>Develop Schedules, Shift Routines, Manloadings.</u>

In developing the staffing plan for each Facility, the continuous operational and maintenance requirements of such Facility will be analyzed. All periodic testing, inspections and maintenance activities will be identified as well as those operational and maintenance requirements that require specialized and additional assistance at specific times during the maintenance cycle of such Facility.

The staffing plan will provide for a permanent Facility staff that will be fully responsive to all ethanol and wet distillers grains production demands and will be responsible for the performance of all preventive maintenance and routine repairs.

Each Facility will be manned on a 24-hours per day, 7 days a week basis. Most operating personnel will work a rotating shift schedule. This schedule will coincide with the times of the peak production hours and provide for minimal shift changes and possible interruptions during the high production periods. Selected off-shift personnel will be used for performing maintenance tasks and when necessary, fill in for sickness, absence and vacation relief.

The remaining Facility staff will work a normal five-day, 40-hour week but may be rescheduled to respond to events being carried out at the Facility. Each Facility staff will be autonomous and under the direction of the Manager Representative. In the absence of the Manager Representative, he/she will appoint a senior person to act in his/her behalf.

The onsite operations and maintenance staff will be supported for non-routine functions by Manager's home office, parent, Affiliate, and available companies. Specialized vendor associated technical support will be subcontracted as needed during outage planning, inspections and overhauls.

During periods of high maintenance activity, qualified maintenance personnel will be made available to each Facility staff through contract support.

2. <u>Facilities Operations and Performance Monitoring.</u>

Manager's management duties will include the following:

- (a) Perform and record periodic operational checks and tests of equipment in accordance with the equipment manufacturer's specifications.
- (b) Maintain operating logs, records and reports for operation of each Facility.
- (c) Perform all Grain Handling Services at each Facility.
- (d) For each Operating Facility, inform the applicable Owner as far in advance as possible when such Owner's Facility does not have the capability to receive corn or store ethanol or wet distillers grains.

Performance of each Facility will be monitored on a continuing basis so that performance data are distributed quickly and accurately to the Facility staff and to Owners. Important parameters to be monitored include Ethanol, wet distillers grains and, if applicable, CO₂ output. These parameters are assumed to be available through the distributive control system.

C. <u>Layup and Preservation Services</u>.

- (a) Adhere to all regulatory requirements including management of all applicable permits including each Facility Storm Water Pollution Prevention Plan and the Facility air permits to include leak detection and repair inspections.
- (b) Clean and lay-up equipment to protect from environmental factors including rust and mold.
- (c) Maintain a regular program for equipment preservation to include equipment rotation and inspection.
- (d) Monitor plant fire protection system and alarms.
- (e) Maintain plant readiness for potential re-start or sale.
- (f) Maintain site to include lawn and garden maintenance.
- (g) Perform site security services to prevent theft, vandalism, etc.

D. Facility Restart.

If a Facility restart is requested by an Owner, Manager will provide usual and customary services including the following:

- (a) Interview, select, hire and train necessary operating personnel.
- (b) Make ready all equipment.
- (c) Order necessary start-up supplies.
- (d) Communicate re-start activities to appropriate parties including city and county officials, suppliers, vendors, etc.
- (e) Obtain all necessary permits and conduct all testing necessary to satisfy the terms and conditions of any such permit.

E. Administrative Support.

1. General.

Manager will develop and implement a comprehensive and specific administrative procedures including:

- (a) Maintain an effective Facility work force through proper hiring, training, administration and compensation.
- (b) Procure labor, materials and services exclusive of items which are designated the responsibility of an Owner.
- (c) All replacement spare parts for those taken out of inventory with either an identical part or replacement part(s) manufactured by the original equipment manufacturers, or where practical and economical, refurbish (or have refurbished), spare parts to allow their reuse.
- (d) Provide information and other reasonable assistance at any Owner's request regarding operation and maintenance of the Facilities.
- (e) Maintain accurate Facility cost ledger and accounting records regarding the materials and services provided in accordance with generally accepted accounting principles, consistently applied.
- (f) Cooperate with the respective Owners' independent certified public accountant to perform annual financial audits for all cost reimbursable services provided for the Facilities.
- (g) Communicate with: (i) the corn suppliers or transporters, as the case may be, under each Corn Supply Agreement with respect to the amount of corn Owners procures from such corn suppliers or transporters; and, (ii) Owners with respect to all other matters including the amount of Ethanol, wet distillers grains and, if applicable, CO₂ produced per Owners' schedule and optimize the use of the corn.
- (h) Procure Consumables.

2. Facility Safety Program.

To ensure the safety of all employees and personnel working in or near each Facility, Manager will establish and implement a safety plan which conforms to federal and any state or local regulations. Key components of the plan will include:

- (a) <u>Lock Out/Tag Out</u>. Only trained and authorized employees shall perform work on equipment that requires Lock out/tag out (LOTO). Employees shall follow written LOTO procedures when performing such work.
- (b) <u>Confined Spaced Entry</u>. Only trained and authorized employees shall perform work on any equipment that requires the employee to enter a permit required confined space. Employees shall follow written confined space entry procedures.
- (c) <u>Hot Work.</u> Only trained and authorized employees shall perform work on any equipment that requires the employee to obtain a hot work permit. Employees shall follow written hot work procedures.
- (d) <u>Responsibility</u>. Operations and duties shall be performed only by duly authorized employees, who shall be held responsible for their actions.
- (e) <u>Facility Safety Inspections</u>. Employees shall inspect on a routine basis; operating conditions, equipment and the general workplace to prevent a potential hazard to personnel and equipment.
- (f) <u>Records</u>. Employees required to keep logs and records shall keep them current and accurately. Abnormal or special conditions shall be called promptly to the attention of the proper supervisor and logged. Shift employees shall familiarize themselves with all activities within their jurisdiction that have taken place during the preceding shift.

3. <u>Facility Security</u>.

Manager will adhere to each written Facility Security Plan and each Facility staff will be trained in its requirements. Each Facility staff and all visitors will be required to adhere to the plan to ensure security of such Facility in normal as well as emergency situations.

F. Maintenance Management Program.

The prime objective of the maintenance management program will be to maximize equipment reliability and availability and minimize maintenance costs. The ultimate goal is to maximize Facility profitability and to maintain, as closely as possible, given normal mechanical degradation, to the original design performance of all major pieces of equipment.

The key elements of the maintenance program are:

- Preventive maintenance
- Predictive maintenance
- Corrective maintenance
- Operational downtime management
- Spare parts inventory control

1. <u>Preventive Maintenance</u>.

Preventive maintenance will be done by each Facility staff, consisting of periodic inspection and adjustment of equipment, to avoid deterioration to a point at which the equipment will not start, run, or operate efficiently or reliably. Some preventive maintenance can be accomplished while the unit is shutdown or running.

Preventive maintenance schedules will be included in the computerized program and calibrated to an overall Facility schedule. This schedule will alert maintenance crews, providing daily, weekly, monthly and annual forecasting of preventive maintenance activities. Spare parts considerations, such as lead time, can also be correlated.

Preventive maintenance intervals are directed by two measures, calendar time and unit operating parameters (vibration measurements, operating temperatures, amp loading, etc). Items scheduled by calendar time will be reviewed on a daily, weekly, monthly or yearly basis. As a rule, preventive maintenance work will be scheduled to result in minimal interference with Facility operations.

2. Predictive Maintenance.

A predictive maintenance program will center on Manager's ability to track vital trend information. Observing critical variables over time will yield valuable information regarding equipment deterioration. The preventive maintenance frequency may be increased or decreased, depending on the condition of the equipment found during a predictive maintenance inspection.

Use of the preventive and predictive maintenance schedules, in conjunction with the monitoring of operational data that reflect equipment condition, will result in high reliability and reduced maintenance costs.

3. Corrective Maintenance

The corrective maintenance activities will be aimed at avoiding repeat failures. Discussion meetings will be held to review failures which caused major repairs or shutdowns. Each Facility staff will review the conditions preceding the failure and, if possible, determine the exact cause, and make findings available to maintenance and operating personnel.

4. Shutdown Management.

Shutdowns for equipment repair or replacement will be aggressively managed to minimize downtime. Advanced planning, work packages, shutdown schedules and other project management methods will be used to allocate Facility resources to produce efficient shutdowns.

During the pre-shutdown phase, each Facility staff and the major equipment manufacturers will cooperate closely to conduct the planned inspections in a minimum amount of time at a reasonable cost. This advance planning might begin anywhere from six months to a year before the shutdown, depending on the need for and the availability of major equipment components.

All pre-shutdown and shutdown related activities will be broken down into work packages or discrete elements which are more easily tracked over the course of the shutdown and provide a better means to manage the shutdown schedule.

A scheduling program, such as Microsoft Project, using the critical path method will itemize various work packages, organize them and calculate the effect any work package has on the overall shutdown length. This program will provide a reporting tool that allows each Facility staff to create easy-to-understand shutdown schedules and reports showing manpower and equipment resources, usage profiles and problems leading to schedule slippage.

5. <u>Spare Parts Inventory Control.</u>

Each Facility staff will implement a spare parts inventory control system designed to minimize tied-up capital, yet not jeopardize Facility contractual commitments by risking extended equipment downtime.

Each Facility staff will establish an adequate spare parts inventory by studying manufacturers' recommendations, reviewing past experience, and evaluating parts availability and delivery time from the various vendors, taking into account acceptable equipment downtime. The spares inventory will be managed so that parts are replaced when used, maintenance records are kept current, and the inventory is updated regularly to reflect changing Facility requirements.

Prior to outages and inspections, each Facility staff will order parts prone to normal wear and tear that are identified as needing to be replaced through the preventive/predictive maintenance programs. Furthermore, by maintaining close contact with the original equipment suppliers, each Facility staff will ensure that they purchase the latest revision parts.

G. Corporate Support.

Manager will provide usual and customary support including the following:

Accounting & Finance

- Invoice/AR/Collection
- Purchasing/AP
- Inventory Tracking
- Hedge Accounting
- Utilize Software for Reporting
- Quarterly and Year End Reports Audited if required by an Owner

Treasury & Cash Management

Loan Compliance

Insurance

Property Tax Preparation & Appeals

Corporate Tax Preparation

Administration Services

Human Resources

- Payroll Administration (ADP)
- Benefits Administration
- Employee Relations
- Recruit/Hire/Terminate Personnel
- Training Obligations

IT

- Operate ERP System
- Implement and Operate Communications Systems
- Maintain Desktop Equipment and Services

Facilities Management

- Lease Administration
- Office Supplies
- Administrative Support

Operations Services

Plan, monitor and report plant activities.

- Activity planning for lay-up and preservation to minimize asset risk and cost.
- Monitoring/auditing of activities to ensure compliance with regulations/loan provisions.
- Manage personnel resources during activities.

Plan, monitor and report SH&E activities.

- Numerous regulatory issues to manage during plant layup and preservation.
- Title V activities for California plants.
- CA low NOx requirement being instituted.
- Regulatory reports management.

Licensing and Reporting

- EPA RFS program registration.
- DOT hazardous materials registration.
- DOE monthly reports.
- USDA grain warehouse operator reports.
- Others as may be required by any Permit condition or applicable law.

Legal Services

Litigation/Dispute Management

- Claims in process

Contract Management

- Documentation tracking systems

Compliance Oversight

- Operating and environmental regulatory oversight

APPENDIX B

Last Name	First Name	Division	Home Department	Job title	Level
Williamson	Douglas	PECA	740070 Operations Management	General MGR of Ops	DIR
Pagard	Cheryl	PECA	740070 Operations Management	Permitting & Compliance, Director	DIR
Jones	Lyndon	PECOL	690010 Plant Labor - General	Plant, Manager	DIR
Jundt	James	PECOL	690010 Plant Labor - General	Commodity/Ag Prod, Manager Environmental Health & Safety	MGR
Lambert	David	PECOL	690010 Plant Labor - General	Manager	MGR
Folmar	Matthew	PECOL	690010 Plant Labor - General	Lab Manager	MGR
Stark	William	PECOL	690010 Plant Labor - General	Maintenance, Manager	MGR
Trim	Steven	PECOL	690010 Plant Labor - General	Production, Manager	MGR
Powell	Joey	PECOL	690010 Plant Labor - General	Comm Operator, Lead	Non-ex
Allardin	Jeanette	PECOL	690010 Plant Labor - General	Commodity Operator	Non-ex
Hernandez	Arturo	PECOL	690010 Plant Labor - General	Commodity Operator	Non-ex
Imus	Jason	PECOL	690010 Plant Labor - General	Commodity Operator	Non-ex
Mikaelsen	Chance	PECOL	690010 Plant Labor - General	Commodity Operator	Non-ex
Sebelien	John	PECOL	690010 Plant Labor - General	Commodity Operator	Non-ex
Timpy	Michael	PECOL	690010 Plant Labor - General	Commodity Operator	Non-ex
Partlow	Stephen	PECOL	690010 Plant Labor - General	Electrician	Non-ex
Abercrombie	Mafe	PECOL	690010 Plant Labor - General	Lab Technician I	Non-ex
Putman	Jackie	PECOL	690010 Plant Labor - General	Lab Technician I	Non-ex
Braden	Larry	PECOL	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Drayton	Donald	PECOL	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Guenther	Marvin	PECOL	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Kestler	Randy	PECOL	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Richardson	David	PECOL	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Spinney	Peter	PECOL	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Richards	David	PECOL	690010 Plant Labor - General	Maintenance Supervisor	Non-ex
Bittinger	Wanliya	PECOL	690010 Plant Labor - General	Office, Manager	Non-ex
Bieren	John	PECOL	690010 Plant Labor - General	Production Operator	Non-ex
Givens	Frank	PECOL	690010 Plant Labor - General	Production Operator	Non-ex
Magallanes	Hector	PECOL	690010 Plant Labor - General	Production Operator	Non-ex
McBride	Ronnie	PECOL	690010 Plant Labor - General	Production Operator	Non-ex
Orcutt	Johnny	PECOL	690010 Plant Labor - General	Production Operator	Non-ex
Pena	Jose	PECOL	690010 Plant Labor - General	Production Operator	Non-ex
Savage	Cory	PECOL	690010 Plant Labor - General	Production Operator	Non-ex
Town	Mitchell	PECOL	690010 Plant Labor - General	Production Operator	Non-ex
White	Michael	PECOL	690010 Plant Labor - General	Production Operator	Non-ex
Reeves	Amanda	PECOL	690010 Plant Labor - General	Receptionist	Non-ex
Hedgpeth	Robert	PECOL	690010 Plant Labor - General	Utility Operator	Non-ex
Shuler	Michael	PECOL	690010 Plant Labor - General	Utility Operator	Non-ex

OPEN		PEM	690010 Plant Labor - General	Plant, Manager	DIR
Lovett	Michael	PEM	690010 Plant Labor - General	Commodity/Ag Prod, Manager	MGR
Taito	Talaonuupo	PEM	690010 Plant Labor - General	Commodity Operator	Non-ex
Ochoa	Bacilio	PEM	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Abarca	Jose	PEM	690010 Plant Labor - General	Production Operator	Non-ex
Barboza	Bruno	PEM	690010 Plant Labor - General	Production Operator	Non-ex
Valenzuela	Luis	PEM	690010 Plant Labor - General	Utility Operator	Non-ex
Wilson	Thomas	PEMV	690010 Plant Labor - General	Plant, Manager	DIR
Garza	Juan	PEMV	690010 Plant Labor - General	Commodity/Ag Prod, Manager	MGR
				Environmental Health & Safety	
Rutherford	William	PEMV	690010 Plant Labor - General	Manager	MGR
Jensen	Zackery	PEMV	690010 Plant Labor - General	Lab Manager	MGR
Lowe	Alvin	PEMV	690010 Plant Labor - General	Maintenance, Manager	MGR
Teubner	Joseph	PEMV	690010 Plant Labor - General	Production, Manager	MGR
Cantu	Freddy	PEMV	690010 Plant Labor - General	Commodity Operator	Non-ex
Cardona	Crispin	PEMV	690010 Plant Labor - General	Commodity Operator	Non-ex
Green	Cecil	PEMV	690010 Plant Labor - General	Commodity Operator	Non-ex
Torres	Patrisio	PEMV	690010 Plant Labor - General	Commodity Operator	Non-ex
Zamora	Julio	PEMV	690010 Plant Labor - General	Commodity Operator	Non-ex
OPEN		PEMV	690010 Plant Labor - General	Commodity Operator	Non-ex
OPEN		PEMV	690010 Plant Labor - General	Commodity Operator	Non-ex
Heward	Scott	PEMV	690010 Plant Labor - General	Electrician	Non-ex
Bair	Monica	PEMV	690010 Plant Labor - General	Lab Technician I	Non-ex
Graetzer	Halina	PEMV	690010 Plant Labor - General	Lab Technician I	Non-ex
Richins	Paul	PEMV	690010 Plant Labor - General	Maint Mech, Lead	Non-ex
Crystal	William	PEMV	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Gummow	Eric	PEMV	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Hess	Robert	PEMV	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Jimenez	Manuel	PEMV	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Sibbett	Jim	PEMV	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Vela	Joseph	PEMV	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Day	Paula	PEMV	690010 Plant Labor - General	Office, Manager	Non-ex
Anderson	Kerry	PEMV	690010 Plant Labor - General	Production Operator	Non-ex
Baer	Mark	PEMV	690010 Plant Labor - General	Production Operator	Non-ex
Carper	Heather	PEMV	690010 Plant Labor - General	Production Operator	Non-ex
Davidson	Brett	PEMV	690010 Plant Labor - General	Production Operator	Non-ex
Halford	Dennis	PEMV	690010 Plant Labor - General	Production Operator	Non-ex
Kossman	Justin	PEMV	690010 Plant Labor - General	Production Operator	Non-ex
Rementeria	Lori	PEMV	690010 Plant Labor - General	Production Operator	Non-ex
Rios	Raul	PEMV	690010 Plant Labor - General	Production Operator	Non-ex
Satterwhite	Adam	PEMV	690010 Plant Labor - General	Production Operator	Non-ex
Thompson	Robert	PEMV	690010 Plant Labor - General	Production Operator	Non-ex
Craig	Elizabeth	PEMV	690010 Plant Labor - General	Receptionist	Non-ex
Koyle	Jared	PEMV	690010 Plant Labor - General	Utility Operator	Non-ex

Nelson	Mark	PEMV	690010 Plant Labor - General	Utility Operator	Non-ex
Todd	Royce	PES	690010 Plant Labor - General	Plant, Manager	DIR
Cayuela	Antonio	PES	690010 Plant Labor - General	Commodity/Ag Prod, Manager	MGR
				Environmental Health & Safety	
Lund	Kirk	PES	690010 Plant Labor - General	Manager	MGR
Huynh	Thao	PES	690010 Plant Labor - General	Lab Manager	MGR
Lee	Patrick	PES	690010 Plant Labor - General	Maintenance, Manager	MGR
Shippy	Skyler	PES	690010 Plant Labor - General	Production, Manager	MGR
Brown	Paul	PES	690010 Plant Labor - General	Commodity Operator	Non-ex
De Los Santos	Juan	PES	690010 Plant Labor - General	Commodity Operator	Non-ex
Gallo	Jose	PES	690010 Plant Labor - General	Commodity Operator	Non-ex
Gutierrez	Jose	PES	690010 Plant Labor - General	Commodity Operator	Non-ex
Lahr	Dennis	PES	690010 Plant Labor - General	Commodity Operator	Non-ex
Machado	Johnny	PES	690010 Plant Labor - General	Commodity Operator	Non-ex
Martin	Bento	PES	690010 Plant Labor - General	Commodity Operator	Non-ex
Orosco	Rudy	PES	690010 Plant Labor - General	Commodity Operator	Non-ex
Wolford	Stephen	PES	690010 Plant Labor - General	Commodity Operator	Non-ex
Torre	Christopher	PES	690010 Plant Labor - General	Electrician	Non-ex
Kaur	Manpreet	PES	690010 Plant Labor - General	Lab Technician I	Non-ex
Teame	Mekonen	PES	690010 Plant Labor - General	Lab Technician I	Non-ex
Davis	John	PES	690010 Plant Labor - General	Maint Mech, Lead	Non-ex
Bowen	Michael	PES	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Newport	Brian	PES	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Parkinson	Phillip	PES	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Spradling	Lee	PES	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
OPEN		PES	690010 Plant Labor - General	Maintenance Mechanic	Non-ex
Riker	Rhonda	PES	690010 Plant Labor - General	Office, Manager	Non-ex
Cepriano	Benjamin	PES	690010 Plant Labor - General	Production Operator	Non-ex
Deleon	Cheryl	PES	690010 Plant Labor - General	Production Operator	Non-ex
Eckerdt	Douglas	PES	690010 Plant Labor - General	Production Operator	Non-ex
Gauba	Sukrat	PES	690010 Plant Labor - General	Production Operator	Non-ex
Lombrana	Benjamin	PES	690010 Plant Labor - General	Production Operator	Non-ex
Mccullough	Edker	PES	690010 Plant Labor - General	Production Operator	Non-ex
Pagnucci	Nicholas	PES	690010 Plant Labor - General	Production Operator	Non-ex
Wells	Raymond	PES	690010 Plant Labor - General	Production Operator	Non-ex
OPEN	•	PES	690010 Plant Labor - General	Production Operator	Non-ex
Maxey	Christina	PES	690010 Plant Labor - General	Receptionist	Non-ex
Esteves	Gilbert	PES	690010 Plant Labor - General	Utility Operator	Non-ex
Koch	Kurt	PES	690010 Plant Labor - General	Utility Operator	Non-ex
Montez	Ralph	PES	690010 Plant Labor - General	Utility Operator	Non-ex

CS REPORT



Pacific Ethanol Stockton LLC Monthly Cold Shutdown Report

Pacific Ethanol, Inc.

May 2009

Contents:

Tab	Title	Report Description
ï	Act vs Budg	Budget review - actual vs. budget
ii	Avail	Plant Availability
iii	Cash	Cash receipts and disbursements including balances
iv	Major Maint	Major Maintenance Activities
v	Losses	Material Casualty Losses
vi	Disputes	Review of Disputes & Claims
vii	Permits	Permit and Compliance Activities
viii	Comparison	Comparison of figures for the current month vs last month

Prepared by Plant Manger Lyndon Jones 6/10/2009

Reviewed by Manager Representative 6/12/2009



Pacific Ethanol Columbia LLC Monthly Operatins Report

Pacific Ethanol, Inc.

May 2009

Contents:

Tab	Title	Report Description
i	Act vs Budg	Budget review - actual vs. budget
ii	Production	Ethanol and WDG Production and Delivery
iii	Com	Com Deliveries and use
iv	Avail	Plant Availability
٧	Cash	Cash receipts and disbursements including balances
vi	Major Maint	Major Maintenance Activities
vii	Losses	Material Casulty Losses
viii	Disputes	Review of Disputes & Claims
ix	Permits	Permit and Compliance Activities
X	Comparison	Comparison of figures for the current month vs last month

Prepared by Plant Manger Jay Holthus 6/10/2009

Reviewed by Manager Representative 6/12/2009

APPENDIX E

Pacific Ethanol Holding Co. LLC Quarterly Budget. GD 2011 As of March 21, 2011 Format Revis of May 25, 2011

Formal ricers of May 23, 2011	_		700				- 107	1007			1,000			
		M# 2	WR 2 8611.8615	Vel. 1 86/19.64/22	WR 6 94:25:94:29	wk5 05-02-05-06	MA 8 05/09/05/13	WA 7 85/16-05/00	wk 8 05/23/85/27	WA 5 05/00 06/00	WR 10 8696 86/18	wk 11 96/13/86/17	96 28 96 24 96 28 96 24	JULY wk 13 06/27/07/01
Consendity Margin Signi	1	0.40	\$ 0.40	\$ 0.40	\$ 0.40	\$ 0.40	\$ 0.40	\$ 0.40 E	T40 S	0.66 1	056	s 050 s	0.56	\$ 06
Revence, Operating & Flux Admin Accts														
Beginning Cash Belonce	*	2,529,106	1 2,097,599	1 2,074,222	1 2,087,442	\$ 296,603	\$ 266,603	1 296,603 1	493,747 8	732,100 1	16,603	5 200 pth 5	741,520	8 2,297,19
Cash inflows														
Revenue - Ethanol (Kinergy)		7,806,997	7,532,498	7 734 549	7,300,720	7,590,568	6,482,636	8,552,984	8,552,964	7,583,949	8,537,147	8 820,426	8,800,438	7,600,39
Reneros - WD 0 (PAP) and others Reneros - CEPIP		1,794,583	1,717,436	1 (889,466)	1,855,918	1,808,166	1,866,306	1,883,251	1,778,300	1,730,832	1,816,453	1,864,551	1,779,544	1,704.29
Others			- 1											
Total Cash Inflows		9,901,581	9,249,994	9,424,015	8,956,638	9,699,064	10,398,962	10,416,235	10,391,284	9,294,780	10,363,909	10.695,019	10,589,972	9,934,89
Dicharcoments														
Operating Dishursements														
Corn Natural Gas		(7 (883,000) (202,492)	(7,662,960) (282,492)	(E) (14,661) (273,587)	(8,177,236)	(8)(534,663) (262,492)	(0.634,663) (262,462)	(9,634,663)	(7,401,140) (083,167)	(8 568 ,474)	(0.660,474)	(II)(600,674) (273,587)	(7,490,130) (253,967)	8,705,61 (201.34)
Bacinoty		(300,246)	(2027,692)	(109.128)	(123,026)	(340.415)	(Jacu, Mez.)	(F21 B53)	(017.643)	(379.396)	(202,902)	(273/397)	(122,788)	(111.36
Insurance						(#3,#91)		2000	20000	(43,491)				(43,49
Property Tax and other Taxes		1000	(360,000)				20-510-513					0.00	17.57	(323,13
Et kanel fleight		(149,149)	(151,966)	(1.43.990)	(153,162)	(168.177)	(158,177)	(158.177)	#36,500s	(168,177)	(168.177)	(198,177)	(136,580)	(159.17
Co-product fireight Lease Pointents		(83,921)	(61,409)	(61,688)	(52,705)	(61,409)	(61,409)	(61,409)	(52,706)	(61,409)	(61,409)	(61,489)	(54,299)	(61,400
Brain presurement and bandling w/PAP			6205,7300		(24,230)		(195,143)		Depart.			(201.872)	property.	
Plant Bygglies & Maintenance		(597,108)	(597,108)	(807,108)	(697,108)	(601,896)	(597,108)	(897,198)	(697,100)	(981,896)	(997,109)	(597,308)	(597,108)	(901),894
Capital Expenditures		(91,055)	(123,867)	(120,007)	(123,007)	(123,867)	(142,630)	(142 (530)	(142,630)	(142,630)	[29,010]	(29 (8 10)	(29,810)	(29,810
Other PEHC casts: Professional Fee														
Bank Fee		100		- 0	- 51		- 8		35	- 5			- 2	
Cordingency		50,000	(50,000)	(50,000)	50,000	50,000	(50,000)	(90,000)	50,000	50,000	(50,000)	50,000	50,000	80,000
Total Operating Disbussements		(9,296,003)	(9,370,359)	(9,372,711)	[9,594,519]	(10,296,508)	(10,121,702)	(10 p19,507)	(8)(8) 4 (202)	(10,388,043)	(9,647,549)	(10,040,517)	(6,756,235)	(10,368,110
Assot Monogement Agreement														
Direct Reindoursable terms under AMA.														
Payoti & Sevetits - Plants & Plant Operations			(397,366)	7	(933,368)	0	(333,366)	*	(939,366)	7	(333.368)		(333,366)	
Other Direct Expenses: Insurance														
Professional fees		(9,302)	(9,302)	(9,302)	(6,977)	(9,302)	5,302)	(9,302)	(6,977)	(9.302)	(9,302)	(9,302)	(6,977)	9,300
Total Street Reimburs able Borns under AMA		(9.302)	(342,670)	(9.302)	(340,346)	(9.302)	(342,670)	9.302)	(340,345)	(9.302)	(342.670)	8.302)	(340,345)	9.300
Ass of Management Fox		-	(132,500)	-	-	(132,500)	-	(1.32,500)		(132,500)	-	(132,500)	-	(132,500
Total Asset Management Agreement		(9,302)	(475,170)	(9,302)	(340,345)	(141,802)	(362,670)	(141 (902)	(240,346)	(141,802)	(342,670)	(141,002)	(340,345)	(141,002
Professional Fee		27.781)	(27,781)	(27.781)	(27.791)	(27.781)	(27,781)	(27,781)	(27.701)	(27,781)	(27.781)	(27.781)	£7781)	(27.78)
Interest & Fees					(874,190)				910,519					(003,570
Total Dishuserouts		(9,433,087)	9,973,310)	(9,410,796)	(41.836.819)	(10,466,092)	(10,492,153)	(10,189,091)	(10,092,943)	(10,957,927)	(10,218,001)	(10,210,101)	(9.104,361)	(11,427,964
Exit Facility Funding		-			59347	767,036	153,202			547,261	114,401		-	- 4
Ending Cash Balance		2 997 509	\$ 2,074,222	\$ 2,007,642	\$ 206,600	\$ 200,000	\$ 266,600	F 401747 S	732/60 \$	16.003 .1	266,603	s 741 520 s	2 207 (3)	\$ 214,900
Curvatative funding	5	29,312,800	\$ 29,312,900	\$ 23,312,900	\$ 29,372,247			\$ 24,292,486 B		24,839,747	24,964,147	8 24,964,147 8	24,964,147	\$ 24,954,147
Net funding	*	20,615,301		\$ 21,225,658		\$ 23,072,002	\$ 24,025,004			28,822,166	24,907,545	26,212,627	22,717,017	24,639,967
Reviews & Operating Account	-													
Beginning Cash Balance	-,	2512502	\$ 2,680,996	\$ 2,057,620	\$ 2,070,840	\$ 250,000	\$ 250,000	1 250,000 5	477.545 5	715.586 5	(0)	\$ 250,000 \$	724.918	\$ 2,220,520
Cash Inflows		9,601,681	9,249,994	9,434,015	8,966,638	9,699,054	10.338,962	10,416,236	10.391.294	9,294,790	10.363,600	10,666,019	10.699.973	9-504-591
													-	
Operating Disbursements Asset Management Agreement		(9,302)	(875,170)	Ø 373,711) Ø 300)	(9,594,519) (340,346)	(10,296,508)	(10,121,702)	(10 (19,507)	(9,814,202)	(10,386,043)	(342,670)	(10,040,517)	(8,736,235) (340,345)	(10,388,110 (141,800
Professional fee		(27,781)	(27,781)	(27.781)	(27.791)	(27.781)	(27,791)	(27,781)	(27.781)	(27,781)	(27.781)	(27.781)	67,7810	Q7.781
Interest & Fees		Meses a	- ALLEY I	100.000	(874,100)	40.000	2000		\$10,515		40.000	300000	40,000	(889,970
Total Bishussements		(9,433,087)	9,973,310)	(9.410.796)	(10,836,809)	(10,496,092)	(10,492,157)	(10,199,091)	(10,092,943)	(10,557,627)	(10,218,001)	(10,210,101)	(9,104,361)	(11,427,964
Exit Facility Funding	_	-			59.347	767,038	163,202			547,261	114,401			
Ending Cash Balance	4	2880.996	\$ 2,067,620	1 2,070,940	\$ 250,000	\$ 250,000	¥ 250,000	4 477.345 4	715.596 6	100 1	250,000	\$ 724.918 B	2,220,539	\$ 297.550
Section 2 and Description	*	2200200	2001,900	4 4000040	1 23/000		-				270,000	124210 6	2,220,000	200

Pacific Ethanol Holding Co. LLC **Quarterly Budget Chart of Accounts (Annotated)**

Line item Credit agreements Notes

Beginning Cash Balance

Cash Inflows

Revenue - Ethanol (Kinergy) Revenue from ethanol sales

Revenue from co product sales, other income (such as rail incentive, Revenue - WDG (PAP) and others

refunds)

Revenue - CEPIP Revenue from CEPIP Others Other special revenue

Total Cash Inflows

Disbursements

Operating Disbursements

As approved by approved by the Owners through the budget process Corn

or by separate line item approval.

Natural Gas Electricity

Insurance Insurance, surety bond paid directly from PEHC

Property Tax and other Taxes Ethanol Freight Co-product freight

Varilease Lease Payments

Grain procurement and handling w/

PAP

Plant Supplies, Maintenance & Catch-all category that includes chemicals, denaturant, and other SG&A

supplies, spare and replacement parts, repairs, janitorial, security and fire services, permitting fees, lab tests, water treatment and disposal, real estate and equipment leases, and SG&A (travel, training,

seminars, meals, entertainment, etc.)

Capital Expenditures Other PEHC costs: Professional Fees

Professionals engaged directly by PEHC (i.e., Hein, Protiviti, KMPG,

legal), all to the extent approved in advance by the Owners through the

budget process or a separate approval process.

Bank fees Operating accounts under exit facility

Contingency

10% compliance

Asset Management Agreement

Total Operating Disbursements

Items can be contracted by Manager or the Plants and in either case **Direct Reimbursable Items under** must be approved in advance by the the Owners through the budget

AMA process or a separate approval process

Payroll & Benefits - Plants & Plant

Operations

Total Asset Management Agreement

Other Direct Expenses Insurance

Insurance paid by PE-DE, with allocations to PEHC plants Professional Fees Professionals engaged by Manager for services to PEHC, all to the

extent approved in advance by the Owners through the budget process

or a separate approval process.

Total Direct Reimbursable Items 10% compliance under AMA

Asset Management Fee Fixed amount per AMA; currently \$265K/mth

New PE Holdco board fees and costs, plus fees and expenses for

Professional Fee officer and advisors of PEHC

Interest & Fees Exit facility related interest and fees **Total Disbursements**

Exit Facility Funding

Ending Cash Balance

Pacific Ethanol, Inc. (Delaware)

Detail of Selling, General & Admin Expenses

Payroll and Benefits by Department All items in this chart of accounts are "Manager Expenses" except as noted below:

Accounting

Corporate Development Executive Finance

Financial Planning & Analysis

Human Resources

Legal Operations Excludes HQ personnel dedicated to Plant management (2).

Office Expense

Corporate Development

M&A transactions

Public policy consultants

FRA membership dues

Donations and contributions - policy Donations and contributions - general

Marketing, promotional items, Web

design

Investor relations advisory

Investor relations other

Press releases

WCB development

Trade shows and seminars

Continuing education

Travel airfare
Travel lodging
Travel car rentla
Meals/entertainment

Excludes budgeted travel expenses for the Plants under approved development project budgets. Excludes budgeted travel expenses for the Plants under approved development project budgets. Excludes budgeted travel expenses for the Plants under approved development project budgets. Excludes budgeted travel expenses for the Plants under approved development project budgets.

Includes computers, software and systems located at the Plants to the extent they comprise part of the wide area and telephone networks. Other hardware items at Plants, such personal computers and servers belonging solely to the local area network, are excluded.

IT Department

Internet Services T1 Lines

Mobile Communications

_Telephones

Conference Calling

Consulting Services - Navision Consulting Services - Network Consulting Services - Phone System Consulting Services - Servers

Continuing Education

Hardware New Users

Maintenance - Environmental

Maintenance - Peripherals

_Maintenance - Phone System

Maintenance - Server

Maintenance - Software Licensing

Maintenance - Workstation

Remote Support

Software New Users

SOX Review/Compliance

Travel

Human Resources Department

Employee recruiting

Background checks and drug screens

Training - corporate

Travel airfare

Travel lodging

Travel car rental

Meals/entertainment

Transfer and relocation

Consulting - general

Consulting - compensation and benefits

Placement fees temp to perm convert

Temporary staffing

Dues and subscriptions

Employee events

Payroll processing fees

Executive Department

Trade shows and seminars

Continuing education

Travel airfare

Travel lodging

Travel car rental

Meals/entertainment

Finance / FPBA Department

Trade shows and seminars

Annual shareholder meeting

Continuing education

Travel airfare

Travel lodging

Travel car rental

Meals/entertainment

Rent and Facilities - Sacramento Office Rent

Lease Terminal fee

Furniture, space planning, art, plants

Security expense

Storage fees

Sacramento parking (reserved)

Sacramento parking (validation/guests)

Office repairs and maintenance

Janitorial services

Operations

 $Consulting \ services-engineering$

Consulting services - biochemistry

Consulting services – general

Trade shows and seminars

Continuing education

Travel - airfare Excludes expense for the Plants.

Travel - lodging Excludes expense for the Plants.

Travel - car rental Excludes expense for the Plants.

Meals/entertainement Excludes expense for the Plants.

Professional Fees - Legal

SEC compliance Corporate maintenance

Lender counsel

Borrower counsel

Project contracting

Business development support

Permitting and regulatory compliance

Human resources

Litigation

Trade shows and seminars

Continuing education

Travel airfare

Travel lodging

Travel car rental

Meals/entertainment

Excludes fees and expenses of outside counsel to the extent they relate soley to legal affairs of the Plants, such as legal fees incurred in connection with loans secured by the Plants.

Excludes fees and expenses of outside professionals to the extent they relate soley to business of the Plants, such as accounting fees incurred for audits and preparation of tax returns of the Plant Owners.

Professional Fees - Accounting

Quarterly reviews

Annual audit - financial

Annual audit - SOX

SOX advisory

401K

Temporary staffing

Derivative accounting

Property tax accounting

Corporate tax prep and advisory

General advisory/special projects

Trade shows and seminars

Continuing education

Travel airfare

Travel lodging

Travel car rental

Meals/entertainment

Director Fees and Expenses

Director fees

Director expenses

Continuing education

Non-cash director compensation

Insurance

D&O liability insurance Property casualty insurance

General liability insurance

Licenses and Fees

Licenses and Fees

PEI DE Asset and Share Tax

Total

PEI DE Asset and Share Tax

Bank Fees

Bank fee Total

Parsons note

Total other interest expense

Depreciation and Amortization

Depreciation expense

Amortization expense intangibles

Excludes premiums for insurance on Plant assets. Excludes premiums for insurance on Plant assets.

Excludes fees for accounts called for by loans secured by the Plants.

Not applicable - not a cash item.

Exhibit I

Sample Performance Incentive Fee Calculation

If six-month EBITDA for plant with a 40 million gallon Operating Capacity is \$5 million and the plant produced 20 million gallons of ethanol during such period, the incentive fee would be calculated as follows:

- The six-month EBITDA is \$5 million. Therefore, the annualized EBITDA is \$10 million (2 X \$5 million).
- The annualized EBITDA per Gallon of Operating Capacity is \$0.25 (\$10 million / 40 million).
- Therefore, the performance bonus is $0.03 \times (\$0.25 \$0.20) \times 20 = \$30,000$.

Exhibit II

Sample Sale Incentive Fee Calculation

Assume (i) a Facility with an Operating Capacity of 60 million gallons is sold generating Asset Sale Proceeds of \$80 million, (ii) the aggregate Operating Capacity of all of the Facilities is 200 million gallons, and (iii) the aggregate outstanding Loans are \$80 million.

Step 1. Compute Sale Price per Gallon of 0.9333333 as follows: Asset Sale Proceeds of 0.9333333 million, less 0.9333333 as follows: Asset Sale Proceeds of 0.9333333 million, less 0.9333333 million. Divide 0.9333333 million. Divide 0.9333333 million.

Step 2. Calculate Incentive Fee for each "Tier" as the product of (i) the excess of the Sale Price per Gallon over applicable tier floor (not to exceed the Tier ceiling for Tiers II and III), (ii) the Operating Capacity of the Facility, and (iii) the applicable percentage for such Tier set forth in Section 8.2, as follows:

	Excess over Tier	Operating Capacity	Applicable	Incentive			
<u>Tier</u>	<u>Floor</u>	(Gallons)	Percentage	<u>Fee</u>			
II - over \$.60-\$.70	\$0.10	60,000,000	0.50%	\$30,000			
III - over \$.70-\$.80	\$0.10	60,000,000	1.00%	\$60,000			
IV - over \$.80	\$0.13333	60,000,000	1.50%	<u>\$120,000</u>			
Step 3. Add fee for each tier to obtain total Incentive Fee: \$210,000							

II-1

Exhibit III

Sample EBITDA Calculation

PROPOSED EBITDA PECOL 4/1/10 - 4/30/10

Description	Current Period Actual	Notes
Description	Actual	Notes
Calca		Deflects financials as autwently reported in monthly AMA report
Sales Ethanol Sales	\$ 5,456,493	Reflects financials as currently reported in monthly AMA report
Biodiesel Sales	Ψ 5,430,473	
E85 Sales		
Denaturant Sales		
RIN Sales		
WDG Sales	1, 101,804	
Syrup Sales	270,456	
Kornplex Sales	270,100	
Grain Sales		
Freight Handling Fees		
Service Revenues		
Freight Revenues	8.204	
Other Sales		
Intercompany Sales	(189,855)	
Total Sales	6,647,103	
Cost of Goods Sold	-,-,	
Ethanol	5,812,345	
Biodiesel	-,- ,-	
E85		
Denaturant		
RINS		
WDG COGS	2,524	
Syrup COGS		
Kornplex Incremental Costs		
Grain		
Labor	192,806	
Freight	490,577	
Demurrage	30	
Supplies	31,184	
Repairs and Maint. Plant	68,020	
Storage		
Equipment Rent and Lease Expense	11,142	
Rail Car Rental		
Insurance, Property Tax and Other	25,775	
Depreciation	178,852	
Derivative (Gains) Losses		
Rail Incentives	(30,300)	
Inventory Adjustments		
Intercompany		
Total Cost of Goods Sold	6,782,954	
Gross Profit (Loss)	(135,851)	
Operating Expenses		
Payroll, Bonus and Benefits		
Hiring and Training	4,500	
Rent or Lease Expense		
Taxes, Permits and Fees	86	
Insurance		
Professional Fees	3,586	
Travel and Auto Expenses	4,265	
Office and Supplies	741	

Advertising and Promotion		
Network, Telephone and Computer		
Broker Commissions		
Bad Debt Expense Business Development		
Non-Cash Compensation Expense		
Depreciation and Amortization	1,140	
Plant Layup & Startup Costs	1,110	
Donations and Contributions		
Intercompany	90,286	
Total operating expenses	104,603	
Net operating income [loss)	(240,455)	
Reorganization Costs	172,107	
Other income (expense):		
Interest Income		
Interest Expense	(11,140)	
Interest Derivatives Gain (Loss)		
Bank Fees	(625)	
Income Tax Expense		
Other Income (Expense) Amortization of Deferred Finance Fees	(2.046)	
_	(3,046) (14,812)	
Total other Income (expense)	(14,812)	
Non-Controlling Interest VIE		
Net Income (Loss)	\$ (427,373)	
Net Income per GL	427,373	
Net Income does not equal gl		
ADJUSTED EBITDA		
CALCULATION		
Net Income (Loss)	\$ (427,373)	
Interest Expense	11,140	
Interest Derivatives Gain (Loss)	-	
Income Tax Expense	170.052	
Depreciation and Amortization COGS Depreciation and Amortization SG&A	178,852 1,140	
Amortization of Deferred Finance Fees	3,046	Capitalized bank tees, etc.
Non-amortizing finance charges	625	Bank fees, etc.
Other lender related expenses	-	Professional and advisory fees, plan administration fees, etc.
LCM	-	and advisory roos, plan administration roos, our
Reorganization Adjustments & Other	172,107	pre-petition, pre-confirmation expenses, other material non-cash (to be
EDIEDA AIOE E E	φ ((0.452)	itemized), etc.
EBITDA (NOT For External Reporting)	\$ (60,463)	



[FORM OF]

AMENDED AND RESTATED ETHANOL MARKETING AGREEMENT ([____] PROJECT)

by and between

PACIFIC ETHANOL [_____], LLC

and

KINERGY MARKETING, LLC

Dated as of June 30, 2011

TABLE OF CONTENTS

			<u>Page</u>
Article I		DEFINITIONS; INTERPRETATION	1
	1.1	Definitions	1
	1.2	Interpretation	5
Article II		MARKETING ACTIVITIES	5
	2.1	Bilateral Transactions.	5
	2.2	Storage	6
	2.3	Obligations of Project Company.	6
	2.4	Back-to-Back Transactions	7
	2.5	Netting	7
	2.6	Title; Delivery Point; Nominations; Measurement.	7
	2.7	Benchmarking	8
Article III		PAYMENTS	8
	3.1	Fees and Payments.	8
	3.2	Overdue Payments; Billing Dispute	9
	3.3	Audit	10
Article IV		TERM; TERMINATION	10
	4.1	Term	10
	4.2	Project Company Defaults and Kinergy Remedies	10
	4.3	Kinergy Defaults and Project Company Remedies	11
	4.4	Change of Control.	11
	4.5	Effect of Termination	11
Article V		INSURANCE	12
	5.1	Kinergy Insurance	12
	5.2	Kinergy Insurance Premiums and Deductibles	13
Article VI		LIMITATIONS ON LIABILITY	13
	6.1	No Consequential or Punitive Damages	13
Article VII		INDEMNIFICATION	13
	7.1	Project Company's Indemnity	13
	7.2	Kinergy's Indemnity	14
Article VIII		REPRESENTATIONS AND WARRANTIES	14

Article IX		FORCE MAJEURE	14
	9.1	Definition	14
	9.2	Effect	15
	9.3	Limitations	15
Article X		DISPUTE RESOLUTION	15
	10.1	Attempts to Settle	15
		Resolution by Expert	15
		Arbitration	16
	10.4	Consequential and Punitive Damages	16
	10.5	Finality and Enforcement of Decision	16
	10.6	Costs	16
	10.7	Continuing Performance Obligations	16
Article XI		CONFIDENTIALITY	16
Article XII		ASSIGNMENT AND TRANSFER	17
Article XIII		MISCELLANEOUS	17
	13.1	Entire Agreement	17
	13.2		17
	13.3	Survival	13
	13.4	Severability	17
	13.5	Governing Law	18
	13.6	Binding Effect	18
		Notices	18
	13.8	Amendment	19
	13.9	No Implied Waiver	19
	13.10	Lines of Credit	19
Exhibit A		Form of PEI Guaranty	
Exhibit B		Operating Protocol	
Exhibit C		Certain Customers	
Exhibit D		Benchmarking	

This AMENDED AND RESTATED ETHANOL MARKETING AGREEMENT (as amended, amended and restated, supplemented or otherwise modified from time to time, this " <u>Agreement</u> ") is entered into by and between PACIFIC ETHANOL [], LLC, a Delaware limited liability company (" <u>Project Company</u> "), and KINERGY MARKETING, LLC, an Oregon limited liability company (" <u>Kinergy</u> "), as of June 30, 2011. Project Company and Kinergy are each individually referred to herein as a " <u>Party</u> ", and collectively are referred to herein as the " <u>Parties</u> ".		
RECITALS		
A. The Parties hereto were previously party to that certain Ethanol Marketing Agreement, dated as of [], 2010 (the "Prior Agreement"), pursuant to which Kinergy provided certain services to Project Company.		
B. Kinergy provides marketing services for denatured fuel ethanol production facilities.		
C. Project Company is the owner of an approximately [] million gallons-per-year denatured fuel ethanol production facility in [], [] (the "Facility") and Project Company has requested that Kinergy provide denatured fuel ethanol marketing services for the Facility and maintain lines of credit available of not less than \$5,000,000.		
D. The Parties desire to amend and restate in its entirety the Prior Agreement and enter into this Amended and Restated Ethanol Marketing Agreement pursuant to which Kinergy will provide such marketing services.		
AGREEMENT		
NOW, THEREFORE, in consideration of the agreements and covenants hereinafter set forth, and intending to be legally bound, the Parties hereto covenant and agree as follows:		
ARTICLE I		
DEFINITIONS; INTERPRETATION		
1.1 <u>Definitions</u> . The following terms shall have the meanings set forth below when used in this Agreement:		
"Act of Insolvency" means, with respect to any Person, any of the following: (a) commencement by such Person of a voluntary proceeding under any jurisdiction's bankruptcy, insolvency or reorganization law; (b) the filing of an involuntary proceeding against such Person under any jurisdiction's bankruptcy, insolvency or reorganization law which is not vacated within 60 days after such filing; (c) the admission by such Person of the material allegations of any petition filed against it in any proceeding under any jurisdiction's bankruptcy, insolvency or reorganization law; (d) the adjudication of such Person as bankrupt or insolvent or the winding up or dissolution of such Person; (e) the making by such Person of a general assignment for the benefit of its creditors (assignments for a solvent financing excluded); (f) such Person fails or admits in writing its inability to pay its debts generally as they become due; (g) the appointment of a receiver or an administrator for all or a substantial portion of such Person's assets, which receiver or administrator, if appointed without the consent of such Person, is not discharged within 60 days after its appointment; or (h) the occurrence of any event analogous to any of the foregoing with respect to such		

Person occurring in any jurisdiction.

"Affiliate" of a specified Person means any corporation, partnership, sole proprietorship or other Person which directly or indirectly through one or more intermediaries controls, is controlled by or is under common control with the Person specified. The term "control" means the ownership, either direct or indirect, of twenty-five percent (25%) or more of the voting securities (or comparable equity interests) or other ownership interests of a Person, or the possession, either direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or any other means whatsoever.

"Agreement" has the meaning given to such term in the <u>preamble</u> hereto.

"Asset Management Agreement" means that certain Second Amended and Restated Asset Management Agreement by and among PEI, as Manager, Pacific Ethanol Stockton, LLC, Pacific Ethanol Holding Co. LLC, Pacific Ethanol Madera LLC, Pacific Ethanol Columbia, LLC and Pacific Ethanol Magic Valley, LLC, dated as of June 30, 2011, as the same may be amended, supplemented or otherwise modified from time to time.

"Bilateral Transaction" means a transaction entered into by Kinergy with one or more Third Parties consisting of one or more forward sales of Ethanol, with respect to which Ethanol produced at the Facility by Project Company is sold to such Third Party or Third Parties.

"Business Day" means any day other than a Saturday, Sunday or a day on which commercial banks in Sacramento, California or New York, New York are required or authorized to be closed.

"Change of Control" has the meaning ascribed thereto in the Credit Agreement.

"Credit Agreement" means the Credit Agreement, dated as of June 25, 2010, by and among Pacific Ethanol Holding Co. LLC, Pacific Ethanol Madera LLC, Pacific Ethanol Stockton, LLC, Pacific Ethanol Magic Valley, LLC, and Pacific Ethanol Columbia, LLC, as Borrowers, Pacific Ethanol Holding Co. LLC, as Borrowers' Agent, WestLB AG, New York Branch, as the administrative agent and the collateral agent, and the lenders parties thereto from time to time, as the same may be amended, supplemented or otherwise modified from time to time.

"Dispute" means a dispute, controversy or claim.

"Ethanol" means denatured fuel ethanol produced by the Facility satisfying the American Society for Testing and Materials (ASTM) D4806 specifications for denatured fuel ethanol.

"Expert" means an expert having sufficient technical expertise to address the matter subject to a Dispute.

"Extension Notice" has the meaning assigned to such term in Section 4.1.

"Facility" has the meaning given to such term in the recitals hereto.

"Financing Documents" means any and all loan agreements, credit agreements (including the Credit Agreement), reimbursement agreements, notes, indentures, bonds, security agreements, pledge agreements, mortgages, guarantee documents, intercreditor agreements, subscription agreements, equity contribution agreements and other agreements and instruments relating to the financing (or refinancing) of the ownership, operation and maintenance of the Facility.

"Financing Parties" means the banks, lenders, noteholders and/or other financial institutions (or an agent or trustee thereof) party to the Financing Documents.

"Force Majeure Event" has the meaning set forth in Section 9.1.

"Good Industry Practice" means any of the practices, methods and acts engaged in or approved by a significant portion of the ethanol production or marketing (as the case may be) industry during the relevant time period, or any of the practices, methods and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision was made, could have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety and expedition. "Good Industry Practice" is not limited to a single, optimum practice, method or act to the exclusion of others, but rather is intended to include acceptable practices, methods or acts generally accepted in the region.

"Governmental Authority" means any United States federal, state, municipal, local, territorial, or other governmental department, commission, board, bureau, agency, regulatory authority, instrumentality, judicial or administrative body.

"Incentive Fee" means, for each Bilateral Transaction, the product of 1.0% multiplied by the aggregate amount of the Purchase Price for such Bilateral Transaction; provided, however, that in no event shall the Incentive Fee for any Bilateral Transaction be less than \$.015 per gallon nor greater than \$.0225 per gallon.

"Incentive Fee (Estimated)" means, for each Bilateral Transaction, the product of 1.0% *multiplied by* the aggregate amount of the Purchase Price (Estimated) for such Bilateral Transaction; *provided, however*, that in no event shall the Incentive Fee (Estimated) for any Bilateral Transaction be less than \$.015 per gallon nor greater than \$.0225 per gallon.

"Kinergy" has the meaning given to such term in the preamble hereto.

"Kinergy Indemnified Person" has the meaning given to such term in Section 7.2.

"Law" means any law, statute, act, legislation, bill, enactment, policy, treaty, international agreement, ordinance, judgment, injunction, award, decree, rule, regulation, interpretation, determination, requirement, writ or order of any Governmental Authority.

"Liabilities" has the meaning given to such term in Section 7.1.

"Minimum Rating Criteria", with respect to any Person, means such Person has a senior unsecured debt rating of at least "Baa2" by Moody's or "BBB" by S&P or an equivalent rating from another nationally recognized rating agency, *provided* that each Person set forth on Exhibit C hereto shall be deemed to satisfy the Minimum Rating Criteria.

"Monthly Date" means the last Business Day of each calendar month.

"Moody's" means Moody's Investors Service Inc., and any successor thereto that is a nationally recognized rating agency.

"NewCo" means New PE Holdco LLC, a Delaware limited liability company and the indirect owner on the date hereof of all the equity interests in Project Company.

"Party" or "Parties" has the meaning given to such term in the preamble hereto.

"Payment Adjustment Date" has the meaning given to such term in Section 3.1(b).

"PEI" means Pacific Ethanol, Inc., a Delaware corporation.

"<u>Person</u>" means and includes natural persons, corporations, limited liability companies, limited partnerships, general partnerships, joint stock companies, joint ventures, associations, companies, trusts, banks, trust companies and other organizations, whether or not legal entities, Governmental Authorities and any other entity.

"Prime Rate" means the rate per annum listed as the "Prime Rate" in the "Money Rates" section of The Wall Street Journal from time to time.

"Prior Agreement" has the meaning assigned to such term in the Recitals.

"Project Company" has the meaning given to such term in the preamble hereto.

"Project Company Indemnified Person" has the meaning given to such term in Section 7.1.

"Purchase Price" means, subject to Section 3.1(a). (a) for each Bilateral Transaction that is part of a "pooling" arrangement by Kinergy among Affiliate Persons that are "Borrowers" under and as defined in the Credit Agreement, the product of (i) the weighted average delivered price of denatured fuel ethanol attributable to sales of denatured fuel ethanol which were brokered by Kinergy for such Persons during the applicable calendar month *multiplied by* (ii) the amount (expressed in gallons) of Ethanol delivered by Project Company to Kinergy hereunder during such calendar month in respect of such Bilateral Transactions or (b) for each Bilateral Transaction that is not part of any such arrangement, the aggregate gross payments received by Kinergy (or, if the applicable Third Party defaults in its payment obligations to Kinergy in respect of such Bilateral Transaction, the aggregate amount of gross payments which Kinergy was entitled to receive) for such Bilateral Transaction from the applicable Third Party; *provided* that, in any event, Kinergy shall elect, in its sole discretion at the time of entering into any Bilateral Transaction (and by notice to Project Company), whether such Bilateral Transaction is part of any such "pooling" arrangement.

"<u>Purchase Price (Estimated)</u>" means, with respect to each Bilateral Transaction, the anticipated aggregate payments of Purchase Price with respect to such Bilateral Transaction (as reasonably determined by Kinergy).

"S&P" means Standard & Poor's Ratings Services, a division of The McGraw-Hill Companies, Inc., and any successor thereto that is a nationally recognized rating agency.

"Third Party" means any Person (other than PEI or a subsidiary thereof) that enters into a Bilateral Transaction with Kinergy.

"Transportation Costs" means, for each Bilateral Transaction, all actual, out-of-pocket and documented costs and other expenses incurred by or on behalf of Kinergy in connection with the transportation of Ethanol to the applicable Third Party, including truck, rail, barge and/or terminal costs.

"<u>Transportation Costs (Estimated)</u>" means, for each Bilateral Transaction, the aggregate amount of Transportation Costs anticipated to be incurred by Kinergy in connection with such Bilateral Transaction (as reasonably determined by Kinergy).

1.2 <u>Interpretation.</u> The following interpretations and rules of construction shall apply to this Agreement: (a) titles and headings are for convenience only and will not be deemed part of this Agreement for purposes of interpretation; (b) unless otherwise stated, references in this Agreement to "Sections" or "Articles" refer, respectively, to Sections or Articles of this Agreement; (c) "including" means "including, but not limited to", and "include" or "includes" means "include, without limitation" or "includes, without limitation"; (d) "hereunder", "herein", "hereto" and "hereof", when used in this Agreement, refer to this Agreement as a whole and not to a particular Section or clause of this Agreement; (e) in the case of defined terms, the singular includes the plural and vice versa; (f) unless otherwise indicated, each reference to a particular Law is a reference to such Law as it may be amended, modified, extended, restated or supplemented from time to time, as well as to any successor Law thereto; (g) unless otherwise indicated, references to agreements shall be deemed to include all subsequent amendments, supplements and other modifications thereto; and (h) unless otherwise indicated, each reference to any Person shall include such Person's successors and permitted assigns.

ARTICLE II MARKETING ACTIVITIES

2.1 <u>Bilateral Transactions</u>.

Subject to the terms hereof, Project Company hereby grants Kinergy the exclusive right to market, purchase and sell all of Project Company's Ethanol commencing on June 30, 2011 and continuing through the expiration or early termination of this Agreement, provided, that during the continuance of any default by Kinergy that would allow Project Company to terminate this Agreement pursuant to Section 4.3 or during the 30-day cure period provided in Section 4.3(c) (notwithstanding such cure period), if Kinergy is not performing its obligations with respect to marketing Project Company's Ethanol or during the continuance of any Force Majeure Event (including the effects thereof) that renders Kinergy unable to perform its obligations under this Agreement, then Project Company shall have the right to engage any other Person to market, purchase and sell Project Company's Ethanol and Kinergy shall not be entitled to any compensation (including Incentive Fees) with respect to any replacement services provided by such Person. Kinergy shall use its reasonable commercial efforts to solicit, negotiate and enter into, and Kinergy shall perform, Bilateral Transactions with Third Parties. Kinergy shall have absolute discretion in the solicitation, negotiation, administration (including the collection of payments), enforcement and execution of Bilateral Transactions and all sales of Ethanol produced by the Facility shall be effectuated by Bilateral Transactions. Kinergy shall not enter into any transaction in respect of Project Company's Ethanol that (i) is not a Bilateral Transaction; (ii) requires deliveries of ethanol more than one hundred eighty (180) days after the date of execution of such transaction or (iii) is with a counterparty that does not satisfy the Minimum Rating Criteria, in each case without the prior written consent of Project Company, which consent may be withheld by Project Company in its discretion. Project Company hereby grants Kinergy the power and authority necessary to perform its obligations and exercis

(b)	As further described in Sections 2.3, 2.4 and 2.6 below and except as otherwise provided herein, Project Compa	ny
shall provide Ethanol to K	Kinergy free and clear of all liens and encumbrances.	

- (c) Kinergy shall perform its obligations hereunder and under Bilateral Transactions in accordance with this Agreement, applicable Laws and Good Industry Practice and shall use commercially reasonable efforts to maximize the proceeds generated from the sale of Ethanol.
 - (d) Kinergy shall at all times maintain available lines of credit of not less than \$5,000,000.
- 2.2 <u>Storage</u>. Kinergy acknowledges that Project Company has only limited storage capacity and Kinergy agrees that it shall take any Ethanol requested by Kinergy within seven days (or such longer period of time as may reasonably be agreed by Project Company) of the time that Project Company has made such Ethanol available to Kinergy.

2.3 Obligations of Project Company.

- (a) Project Company shall provide Kinergy with all information reasonably requested by Kinergy, and Project Company shall assist Kinergy as reasonably requested in the solicitation, negotiation and performance of Bilateral Transactions.
- (b) Notwithstanding anything to the contrary herein, Project Company shall not be responsible for the delivery of any Ethanol to Kinergy during any periods of scheduled Facility maintenance (unless and to the extent the applicable Ethanol is available to be delivered to Kinergy from Project Company's storage facilities); *provided*, that, at any time that PEI or one of its Affiliates is not the asset manager pursuant to the Asset Management Agreement (or any successor agreement), Kinergy shall have received at least ten Business Days prior notice of such scheduled maintenance (it being acknowledged and agreed that if Kinergy does not receive at least ten Business Days prior notice, then such maintenance activity shall be deemed to be a mechanical breakdown and covered by clause (c) below for purposes hereof).

- (c) If on any day, Project Company is unable to perform its obligations to deliver Ethanol under this Agreement due to a mechanical breakdown (including a forced outage of the Facility) that is not a Force Majeure Event and such mechanical breakdown has continued for more than three consecutive days, Kinergy shall, at Project Company's option and at Project Company's expense, and provided that, at any time that PEI or one of its Affiliates is not the asset manager pursuant to the Asset Management Agreement (or any successor agreement), Project Company provides Kinergy with prompt notice of its intent to exercise such option, use commercially reasonable efforts to identify and procure replacement Ethanol to be delivered to the Third Party under the applicable Bilateral Transaction. In such event, if and only if the Parties reach agreement as to an alternative delivery point, Kinergy shall acquire and deliver replacement Ethanol in a quantity sufficient to meet the contract quantity of such Bilateral Transaction at such alternate point (and Project Company shall be responsible for all transportation costs associated therewith). In all other instances, Project Company shall be responsible for any damages incurred by Kinergy in connection with Kinergy's failure to perform under the applicable Bilateral Transaction as a result of such mechanical breakdown (it being acknowledged and agreed that Kinergy shall use commercially reasonable efforts to mitigate the effects of any such mechanical breakdown and Project Company's resulting inability to deliver Ethanol).
- (d) At the request of Project Company, Kinergy will cause PEI to execute and deliver and maintain in full force and effect a guaranty in the form of Exhibit A hereto.
- 2.4 <u>Back-to-Back Transactions</u>. Each Bilateral Transaction undertaken by Kinergy shall immediately and automatically, without necessity of further documentation or any action whatsoever by any of the Parties, create and cause to be undertaken according to the terms of this Agreement an equivalent transaction in terms of the obligation to deliver Ethanol, the quantity of Ethanol sold and the timing for the delivery of such Ethanol by Project Company with Kinergy (as if Kinergy were the Third Party).
- 2.5 Netting. Netting of amounts due in respect of Bilateral Transactions between Kinergy and a Third Party may arise in circumstances in which Kinergy owes amounts to such Third Party in respect of Bilateral Transactions and, at the same time, such Third Party owes amounts to Kinergy in respect of Bilateral Transactions. In such circumstances, the party owing the greater amount may pay such amount to the other party as reduced by the amount owed to it and both parties will be deemed to have satisfied their obligations thereby. When such netting occurs, for purposes of this Agreement, for all Bilateral Transactions that have been subject to such netting arrangements, Kinergy shall be deemed to have paid amounts owed by it and to have received amounts owed to it.

2.6 <u>Title; Delivery Point; Nominations; Measurement.</u>

(a) Project Company shall deliver Ethanol to Kinergy in respect of Bilateral Transactions (or corresponding back-to-back transactions under Section 2.4) at the inlet flange of the applicable receiving truck, barge or railcar that will remove such Ethanol from the Facility. Title to, risk of loss with respect to and the obligation to transport such Ethanol shall pass from Project Company to Kinergy at such delivery point. The Parties acknowledge that the quality and quantity of Ethanol may degrade or shrink after such Ethanol is delivered by Project Company to Kinergy at such delivery point, and the Parties acknowledge that the risk of such degradation or shrinkage and all other risk of loss shall be borne by Kinergy.

- (b) Kinergy and Project Company shall utilize the previously agreed upon operating protocol, with respect to the mechanics, timing and process for (i) Kinergy to communicate to Project Company its Ethanol requirements on a monthly, weekly and daily basis, (ii) determining the quantity of Ethanol to be stored by Project Company in its storage facilities, and (iii) implementing the Ethanol sales contemplated by this Agreement. A copy of such operating protocol is attached hereto as Exhibit B. By mutual agreement, such operating protocol shall be updated from time to time thereafter.
- (c) Project Company agrees to collect samples of each shipment of Ethanol it delivers to Kinergy hereunder and keep such samples for 30 days. Project Company shall label each sample to include the customer order number and any other information reasonably necessary to identify such Ethanol and the applicable shipment. Kinergy shall have the right, upon reasonable notice and at reasonable times and at its expense, to test such samples to confirm that the Ethanol delivered to it hereunder meets the requirements of this Agreement. The Parties agree that the amount of Ethanol delivered hereunder (whether measured as net gallons, net liters or otherwise) shall be corrected to and correspondingly adjusted by a reference temperature of 60 degrees Fahrenheit or 15.56 degrees Celsius.
- 2.7 <u>Benchmarking</u>. Kinergy shall furnish on a monthly basis a report benchmarking its performance in accordance with Exhibit D.

ARTICLE III PAYMENTS

3.1 Fees and Payments.

- (a) Within ten days after the date Project Company delivers Ethanol to Kinergy in accordance with Section 2.6(a). Kinergy shall pay to Project Company an amount equal to (i) the Purchase Price (Estimated) with respect to the Bilateral Transaction to which such delivery of Ethanol relates minus (ii) the aggregate amount of Transportation Costs (Estimated) with respect to such Bilateral Transaction minus (iii) the aggregate amount of the Incentive Fee (Estimated) with respect to such Bilateral Transaction (it being acknowledged that Kinergy shall retain for its own account the amount of such Transportation Costs (Estimated) and Incentive Fee (Estimated), and that such amount represents an estimate of the net amounts to be paid to Project Company in connection with such Bilateral Transaction). In connection with each such payment, Kinergy shall deliver to Project Company a statement detailing its calculations of the applicable Purchase Price (Estimated), the applicable Transportation Costs (Estimated) and the applicable Incentive Fee (Estimated).
- (b) Within the first five Business Days of each calendar month (each such date, a "<u>Payment Adjustment Date</u>"), the Parties shall reconcile and "true-up" the actual Purchase Price, Transportation Costs and Incentive Fees for all Bilateral Transactions entered into since the previous Payment Adjustment Date, with the intent of the Parties being that Kinergy shall make up the difference of any "under estimations" and Project Company shall refund any "over estimations". For example, if there are "under estimations" then Kinergy shall pay to Project Company an amount equal to:

- (i) (A) the Purchase Price with respect to such Bilateral Transaction *minus* (B) the Purchase Price (Estimated) with respect to such Bilateral Transaction (to the extent actually paid by Kinergy to Project Company pursuant to <u>Section 3.1(a)</u>), *minus*
- (ii) (A) the Transportation Costs with respect to each such Bilateral Transaction *minus* (B) the Transportation Costs (Estimated) with respect to such Bilateral Transaction, *minus*
- (iii) (A) the Incentive Fee with respect to each such Bilateral Transaction *minus* (B) the Incentive Fee (Estimated) with respect to such Bilateral Transaction.

Each such monthly reconciliation or "true-up" payment shall be paid by Kinergy or Project Company (as applicable) no later than five Business Days after the applicable Payment Adjustment Date. Each Party acknowledges that Kinergy (and not Project Company) bears the risk of non-payment by (a) a Third Party in connection with a Bilateral Transaction with respect to any Third Party (or an Affiliate thereof) that does not satisfy the Minimum Rating Criteria and (b) PEI or any Affiliate thereof that purchases Ethanol from Project Company; *provided* that Project Company and Kinergy acknowledge and agree that any risk of non-payment not borne by Kinergy shall be shared on a pro rata basis among Project Company and each other Person with respect to which sales are brokered by Kinergy (with such pro rata calculations based upon the amount (expressed in gallons) of Ethanol delivered by, or on behalf of, each such Person during the applicable period in respect of which nonpayment has occurred), all as reasonably determined by Kinergy.

- (c) Notwithstanding anything to the contrary in clause (a) or (b) above, if Project Company defaults in its obligation to provide Ethanol to Kinergy in accordance with the terms of this Agreement (including, without limitation, as contemplated by Section 2.3(c)), then Kinergy shall be entitled to set-off and deduct from current and/or future payments owed to Kinergy by Project Company (including the estimated payments pursuant to clause (a) above and the reconciliation and "true-up" payments pursuant to clause (b) above) an amount equal to, as applicable (i) the amount of damage payments owed by Kinergy to the applicable Third Party for failure to provide such Ethanol or (ii) the cost of any replacement Ethanol procured by Kinergy to satisfy the requirements of any Bilateral Transaction, each as a result of Project Company's failure to perform hereunder net of any revenues received in respect of such Bilateral Transaction.
- 3.2 Overdue Payments; Billing Dispute. If Project Company or Kinergy, in good faith, disputes the amount of any payment received by it or to be paid by it or set-off pursuant to Section 3.1 above, the disputing Party shall immediately notify the other Party of the basis for the dispute. The Parties will then meet and use their best efforts to resolve any such dispute. If any amount is ultimately determined to be due to or permitted to be set-off by Project Company or Kinergy (as the case may be), to the extent not previously paid or set-off, (a) Kinergy (or Project Company, as the case may be) shall pay such amount to Project Company (or Kinergy, as the case may be) within five Business Days of such determination or (b) Kinergy (or Project Company, as the case may be) may then set-off such amount (as the case may be). If any Party shall fail to make any payment when due hereunder, such overdue payment shall accrue interest at the Prime Rate plus 2% from the date originally due until the date paid.

Audit. Notwithstanding the payment of any amount pursuant to this Article III, Project Company shall remain entitled (upon reasonable prior notice, at reasonable times and at Kinergy's corporate offices) and the administrative agent under the Credit Agreement (and its consultants, as directed by the administrative agent) shall be entitled (upon reasonable prior notice, not more than once per calendar quarter and at Kinergy's corporate offices) to conduct a subsequent audit and review of (a) all Bilateral Transactions and related records to verify the amount of gross payments, Incentive Fees, Transportation Costs and damage payments and (b) the determination and calculation of the Purchase Price, in each case for a period of two years from and after the applicable Payment Adjustment Date. If, pursuant to such audit and review, it is determined that any amount previously paid by Kinergy to Project Company did not constitute all of the amounts which should have been paid to Project Company, Project Company shall advise Kinergy indicating such amount and the reason the amount should have been paid to Project Company and, subject to the next two sentences, Kinergy shall pay such amount to Project Company within five Business Days of such request along with interest accrued at the Prime Rate plus 2% from the date originally due until the date paid. If the Parties do not agree with respect to any item so noted, the Parties will then meet and use their best efforts to resolve the dispute. If Parties are not able to resolve issues raised by such an audit and review, any disputed items will be resolved in accordance with the provisions of Article X.

ARTICLE IV TERM: TERMINATION

- 4.1 Term. This Agreement shall be effective on the date hereof and, unless earlier terminated in accordance with its terms, shall continue in effect until and including June 30, 2012; *provided*, either Kinergy or Project Company may extend this Agreement for additional one year periods, in each case by written notice to the other (an "Extension Notice") delivered not less than ninety (90) days prior to the end of the original or renewal term, provided further that this Agreement shall nonetheless terminate if the recipient of any such Extension Notice rejects such Extension Notice not more than fifteen (15) days after receipt of the Extension Notice.
- 4.2 <u>Project Company Defaults and Kinergy Remedies.</u> Upon the occurrence of any of the following events, Kinergy may exercise such rights and remedies as may be available to it at law or in equity, including the right to terminate this Agreement, by written notice to Project Company, *provided*, that no such notice shall be required for a termination pursuant to clause (b) of this <u>Section 4.2</u>:
- (a) the failure by Project Company to make any payment, deposit or transfer required hereunder within thirty (30) Business Days after the date such payment, deposit or transfer is due, and such failure continues for fifteen (15) Business Days after receipt of written notice from Kinergy of such failure;

- (b) the occurrence of an Act of Insolvency with respect to Project Company; or
- (c) the failure of Project Company to perform any of its material obligations under this Agreement and such failure continues for 30 days after receipt of written notice from Kinergy of such failure; *provided*, that such 30-day period shall be extended for up to an aggregate of 90 days so long as Project Company is diligently attempting to cure such failure.
- 4.3 <u>Kinergy Defaults and Project Company Remedies</u>. Upon the occurrence of any of the following events, Project Company may exercise such rights and remedies as may be available to it at law or in equity, including the right to terminate this Agreement, by written notice to Kinergy, *provided*, that no such notice shall be required for a termination pursuant to clause (b) of this <u>Section 4.3</u>:
- (a) the failure by Kinergy to make any payment, deposit or transfer required hereunder within fifteen (15) Business Days after the date such payment, deposit or transfer is due, and such failure continues for fifteen (15) Business Days after receipt of written notice from Project Company of such failure;
 - (b) the occurrence of an Act of Insolvency with respect to Kinergy; or
- (c) the failure of Kinergy to perform any of its material obligations under this Agreement and such failure continues for 30 days after receipt of written notice from Project Company of such failure; *provided*, that such 30-day period shall be extended for up to an aggregate of 90 days so long as Kinergy is diligently attempting to cure such failure.

4.4 <u>Change of Control</u>.

- (a) This Agreement shall terminate 45 days after the occurrence of (i) any Change of Control with respect to Project Company or any transfer, assignment, sale or other disposition of more than a majority of the membership interests in Kinergy to any Person which is not an Affiliate of PEI or (ii) any transfer, assignment, sale or other disposition of all or substantially all of the assets comprising the Facility, unless in each case the Parties mutually agree to the contrary.
- (b) In the event that this Agreement (i) terminates pursuant to Section 4.4(a)(i) as a result of a Change of Control of Project Company or (ii) shall be terminated by Project Company pursuant to Section 4.3 above, Kinergy shall, if requested by Project Company, use its best efforts to assign to Project Company its interests in the rail car lease or leases (or the applicable portions thereof) to which Kinergy is a party and in respect of which are dedicated to the transportation of Ethanol purchased pursuant to this Agreement.
- 4.5 <u>Effect of Termination</u>. No termination under this <u>Article IV</u> shall release any of the Parties from any obligations arising hereunder prior to such termination, including payment and obligations under any Bilateral Transaction (or such Bilateral Transaction's corresponding back-to-back transaction arising under <u>Section 2.4</u>), that are not fully performed as of the date of such termination. The exercise of the right of a Party to terminate this Agreement, as provided herein, does not preclude such Party from exercising other remedies that are provided herein or are available at law or in equity; *provided*, *however*, that no Party shall have a right to terminate, revoke or treat this Agreement as repudiated other than in accordance with the other provisions of this Agreement; and *provided*, <u>further</u>, that the Parties' respective rights upon termination shall be subject to the liability limitations of <u>Article IV</u>. Except as otherwise set forth in this Agreement, remedies are cumulative, and the exercise of, or the failure to exercise, one or more remedies by a Party shall not, to the extent provided by Law, limit or preclude the exercise of, or constitute a waiver of, other remedies by such Party.

ARTICLE V INSURANCE

- 5.1 <u>Kinergy Insurance</u>. Without limiting any of the other obligations or liabilities of Kinergy under this Agreement, Kinergy shall at all times carry and maintain or cause to be carried and maintained, the minimum insurance coverage set forth in this Section:
- (a) Kinergy shall maintain or cause to be maintained (i) Workers' Compensation insurance in compliance with the workers' compensation laws of the states in which Kinergy provides services hereunder as extended by the Broad Form All States Endorsements, the United States Longshoreman's and Harbor Workers' Coverage Endorsements on an if-any-exposure basis and the Voluntary Compensation Coverage Endorsement, and (ii) Employer's Liability (including Occupational Disease) coverage with limits of not less than \$1,000,000, which shall cover all of Kinergy's employees engaged in providing services hereunder.
- (b) Kinergy shall maintain or cause to be maintained automobile liability insurance for owned (if any), non-owned and hired vehicles with combined single limits for bodily injury/property damage not less than \$1,000,000 per occurrence and containing appropriate no-fault insurance provisions wherever applicable.
- (c) Kinergy will maintain or cause to be maintained commercial general liability insurance with a limit for bodily injury/property damage of not less than \$1,000,000 per occurrence and \$2,000,000 in the annual aggregate. Such coverage shall include premises/operations, explosion, collapse and underground property damage, broad form contractual, independent contractors, products/completed operations (including operator errors and omissions), broad form property damage, personal injury and incidental professional liability (if not covered under product/completed operations and if commercially available).
- (d) Kinergy shall maintain or cause to be maintained umbrella liability insurance providing coverage limits in excess of those set forth in Section (a), (b) and (c) above. The limits of this umbrella coverage shall not be less than \$10,000,000 per occurrence and in the annual aggregate.
- (e) Kinergy shall maintain or cause to be maintained pollution legal liability for sudden and accidental pollution for physical damage and bodily injury to third parties in an amount of \$3,000,000 per occurrence and in the annual aggregate.

The terms and conditions of all insurance policies (including the amount, scope of coverage, deductibles, and self-insured retentions) shall be acceptable in all respects as of the effective date of this Agreement. All insurance carried pursuant to this Section shall conform to the relevant provisions of this Agreement and be with insurance companies which are rated "A-, X" or better by Best's Insurance Guide and Key Ratings, or other insurance companies of recognized responsibility satisfactory to Project Company. Project Company shall be furnished with satisfactory evidence that the foregoing insurance is in effect and Project Company shall be notified 30 calendar days prior to the cancellation or material change of any such coverage. Coverage for the insurance under Section (c) and (d) above shall be written on a claims made basis provided that if the policy is not renewed, Kinergy shall obtain for the benefit of Project Company an extended reporting period coverage or "tail" of at least three years past the final day of coverage of such policy. Kinergy shall provide Project Company with evidence that such extended reporting period coverage or "tail" has been obtained. Kinergy agrees to ensure that the insurance policies outlined in this Section require the insurer to waive subrogation against Project Company, the Financing Parties and their respective Affiliates together with their respective officers, directors, Affiliates and employees and all such Persons shall be an additional insured as their interests may appear with respect to all policies procured by Kinergy.

5.2 <u>Kinergy Insurance Premiums and Deductibles</u>. All premiums for insurance coverage procured by Kinergy pursuant to <u>Section 5.1</u> shall be reimbursed by Project Company upon demand. Kinergy shall be liable for the payment of all deductibles on insurance policies obtained pursuant to <u>Section 5.1</u>, which amounts shall not be reimbursed by Project Company, provided that, to the extent that a claim under a policy described in <u>Section 5.1</u> is attributable to Project Company's (including its employees' or agents') gross negligence or willful misconduct, Project Company shall be liable for the entire amount of such deductible. In no event shall any premiums, deductibles or any losses in excess of insurance coverage be reimbursed by Project Company hereunder.

ARTICLE VI LIMITATIONS ON LIABILITY

- 6.1 <u>No Consequential or Punitive Damages</u>. In no event shall either Party be liable to any other Party by way of indemnity or by reason of any breach of contract or of statutory duty or by reason of tort (including negligence or strict liability) or otherwise for any loss of profits, loss of revenue, loss of use, loss of production, loss of contracts or for any incidental, indirect, special or consequential or punitive damages of any other kind or nature whatsoever that may be suffered by such other Party, including any losses for which such other Party has insurance to the extent proceeds of insurance have been recovered for such losses.
- 6.2 <u>Breach.</u> Notwithstanding anything to the contrary herein, any failure, breach or default by Project Company under this Agreement that is caused by or is a result of a failure, breach or default by PEI under the Asset Management Agreement will not be a breach of this Agreement.

ARTICLE VII INDEMNIFICATION

7.1 <u>Project Company's Indemnity.</u> Project Company shall defend, indemnify and hold harmless Kinergy and its Affiliates (and each officer, director, employee, shareholder, partner, member or agent of Kinergy and its Affiliates) (each, a "<u>Project Company Indemnified Person</u>") from and against any and all third party claims, actions, damages, expenses (including reasonable and documented attorneys' fees and expenses), losses, settlements or liabilities (collectively, "<u>Liabilities</u>") incurred or asserted against any Project Company Indemnified Person (a) as a result of any failure on the part of Project Company to perform Project Company's obligations under this Agreement (including with respect to any back-to-back transaction under <u>Section 2.4</u>), or (b) arising out of or in any way connected with the grossly negligent acts or omissions of Project Company or its Affiliates.

7.2 <u>Kinergy's Indemnity</u>. Kinergy shall defend, indemnify and hold harmless Project Company and its Affiliates (and each officer, director, employee, shareholder, partner, member or agent of Project Company and their Affiliates) (each, a "<u>Kinergy Indemnified Person</u>") from and against any and all third party Liabilities incurred or asserted against any Kinergy Indemnified Person (a) as a result of any failure on the part of Kinergy to perform its obligations under this Agreement (including with respect to any Bilateral Transaction), or (b) arising out of or in any way connected with the grossly negligent acts or omissions of Kinergy or its Affiliates.

ARTICLE VIII REPRESENTATIONS AND WARRANTIES

Each Party represents that (i) it is duly organized under its jurisdiction of formation and in good standing in each jurisdiction where its failure to so qualify could have a material adverse effect on its ability to perform its obligations hereunder, (ii) it has all necessary power and authority to enter into this Agreement, (iii) it has duly authorized, executed and delivered this Agreement and (iv) this Agreement constitutes a legal, valid and binding obligation of such Party enforceable in accordance with its terms, subject to bankruptcy, reorganization, moratorium or other similar laws affecting the enforcement of the rights of creditors generally and subject to general principles of equity.

ARTICLE IX FORCE MAJEURE

9.1 <u>Definition</u>. As used herein, "<u>Force Majeure Event</u>" means any cause(s) which render(s) a Party wholly or partly unable to perform its obligations under this Agreement (other than obligations to make payments when due), and which are neither reasonably within the control of such Party nor the result of the fault or negligence of such Party, and which occur despite all reasonable attempts to avoid, mitigate or remedy, and shall include acts of God, war, riots, civil insurrections, cyclones, hurricanes, floods, fires, explosions, earthquakes, lightning, storms, chemical contamination, epidemics or plagues, acts or campaigns of terrorism or sabotage, blockades, embargoes, accidents or interruptions to transportation, trade restrictions, acts of any Governmental Authority after the date of this Agreement, strikes and other labor difficulties (other than with respect to its own employees), and other events or circumstances beyond the reasonable control of such Party. Mechanical breakdown (including a forced outage of the Facility) that continues for more than five consecutive days shall be deemed not to be "Force Majeure Event" unless such mechanical breakdown resulted from or was caused by a separate "Force Majeure Event."

- 9.2 <u>Effect.</u> A Party claiming relief as a result of a Force Majeure Event shall give the other Parties written notice within five Business Days of becoming aware of the occurrence of the Force Majeure Event, or as soon thereafter as practicable, describing the particulars of the Force Majeure Event, and will use reasonable efforts to remedy its inability to perform as soon as possible. If the Force Majeure Event (including the effects thereof) continues for fifteen consecutive days, the affected Party shall report to the other Parties the status of its efforts to resume performance and the estimated date thereof. If the Force Majeure Event (including the effects thereof) continues for 180 consecutive days, either Party may terminate this Agreement for convenience. If the affected Party was not able to resume performance prior to or at the time of the report to the other Party of the onset of the Force Majeure Event, then it will report in writing to the other Party when it is again able to perform. If a Party fails to give timely notice, the excuse for its non-performance shall not begin until notice is given.
- 9.3 <u>Limitations</u>. Any obligation(s) of a Party (other than an obligation to make payments when due) may be temporarily suspended during any period such Party is unable to perform such obligation(s) by reason of the occurrence of a Force Majeure Event, but only to the extent of such inability to perform, *provided*, that:
- (a) the suspension of performance is of no greater scope and of no longer duration than is reasonably required by the Force Majeure Event; and
 - (b) the Party claiming the occurrence of the Force Majeure Event bears the burden of proof.

ARTICLE X DISPUTE RESOLUTION

- Attempts to Settle. In the event that a Dispute between the Parties arises under, out of or in relation to, this Agreement, the Parties shall attempt in good faith to settle such Dispute by mutual discussions within fifteen Business Days after the date that an aggrieved Party gives written notice of the Dispute to the other Party. In the event that a Dispute is not resolved by discussion in accordance with the preceding sentence within the time period set forth therein, the Parties shall refer the Dispute to their respective senior officers for further consideration and attempted resolution within fifteen Business Days after the Dispute has been referred to such individuals (or such longer period as the Parties may agree).
- Resolution by Expert. If the Parties shall have failed to resolve the Dispute within fifteen Business Days after the date that the Parties referred the Dispute to their senior officers, then, provided the Parties shall so agree, the Dispute may be submitted for resolution by an Expert, such Expert to be appointed by the mutual agreement of the Parties. Proceedings before an Expert shall be held in Sacramento, California (or any other location agreed to by the Parties). The Expert shall apply to such proceedings the substantive law of the State of New York in effect at the time of such proceedings. The decision of the Expert shall be final and binding upon the Parties. In the event that (a) the Parties cannot agree on the appointment of an Expert within ten Business Days after the date that the Parties agreed to submit the Dispute for resolution by the Expert or (b) the Expert fails to resolve such Dispute within 60 days after the Parties have submitted such Dispute to the Expert, then any Party may file a demand for arbitration in writing in accordance with Section 10.3.

- Arbitration. Any Dispute that has not been resolved following the procedures set forth in Section 10.1 or 10.2 shall be settled by binding arbitration in Sacramento, California (or any other location agreed to by the Parties) before a panel of three arbitrators. Such arbitration shall be conducted in accordance with the Commercial Arbitration Rules of the American Arbitration Association as in effect on the date of execution of this Agreement. Such arbitration shall be governed by the laws of the State of New York. If arbitration proceedings have been initiated pursuant to this Section 10.3 and raise issues of fact or law which, in whole or in part, are substantially the same as issues of fact or law already pending in arbitration proceedings involving the applicable Parties, such issues shall be consolidated with the issues in the ongoing proceedings. THE PARTIES HEREBY AGREE THAT THE PROCEDURES SET FORTH IN THIS ARTICLE X SHALL BE THE EXCLUSIVE DISPUTE RESOLUTION PROCEDURES APPLICABLE TO ANY DISPUTE, CONTROVERSY OR CLAIM UNDER THIS AGREEMENT AND, EXCEPT AS SET FORTH IN SECTION 10.5, THE PARTIES HEREBY WAIVE ALL RIGHTS TO A COURT TRIAL OR TRIAL BY JURY WITH RESPECT TO ANY DISPUTE, CONTROVERSY OR CLAIM UNDER THIS AGREEMENT.
- 10.4 <u>Consequential and Punitive Damages.</u> Awards of Experts and arbitral panels shall be subject to the provisions of <u>Article VI.</u>
- 10.5 <u>Finality and Enforcement of Decision</u>. Any decision or award of an Expert or a majority of an arbitral panel, as applicable, shall be final and binding upon the Parties. Each of the Parties agrees that the arbitral award may be enforced against it or its assets wherever they may be found and that a judgment upon the arbitral award may be entered in any court having jurisdiction thereof.
- 10.6 <u>Costs</u>. The costs of submitting a Dispute to an Expert shall be shared equally among the Parties involved in the Dispute, unless the arbitral panel or the Expert determines otherwise. The costs of arbitration shall be paid in accordance with the decision of the arbitral panel pursuant to the Commercial Arbitration Rules of the American Arbitration Association as in effect on the date of execution of this Agreement.
- 10.7 <u>Continuing Performance Obligations</u>. While a Dispute is pending, each Party shall continue to perform its obligations under this Agreement, unless such Party is otherwise entitled to suspend its performance hereunder or terminate this Agreement in accordance with the terms hereof.

ARTICLE XI CONFIDENTIALITY

Each Party and its Affiliates shall treat as confidential the data and information in their possession regarding the Facility, the other Parties or any Affiliate of any other Party, unless: (a) the applicable other Party agrees in writing to the release of such data or information; (b) such data or information becomes publicly available other than through the wrongful actions of the disclosing Party or the disclosing Party's Affiliate; (c) such data or information was in the possession of the receiving Party or the receiving Party's Affiliate prior to receipt thereof from the disclosing Party with no corresponding confidentiality obligation; or (d) such data or information is required by Law to be disclosed. Notwithstanding the generality of the foregoing, any Party may disclose data and information to (i) the officers, directors, managers, partners, members, employees and Affiliates of such Party, but only to the extent such Persons have a need to know such information for purposes of permitting such Party to perform its obligations hereunder, (ii) any successors in interest and permitted assigns of such Party, (iii) any actual or potential Financing Parties or actual or potential lenders to PEI or any subsidiary thereof, and (iv) any potential equity investors in such Party or its Affiliates or acquirer or potential acquirer of the Project or all or any of the equity interests in Newco or any subsidiary thereof; *provided*, that any Person who receives confidential data and information pursuant to an exception contained in clauses (ii) through (iv) of this Article agrees to confidentiality provisions at least as restrictive as the provisions set forth herein.

ARTICLE XII ASSIGNMENT AND TRANSFER

No Party shall assign this Agreement or any of its rights or obligations hereunder without first obtaining the prior written consent of (a) in the case of Project Company, Kinergy, or (b) in the case of Kinergy, Project Company, *provided*, that any Party shall be entitled to assign its rights hereunder (as collateral security or otherwise) for financing purposes (including a collateral assignment to any Financing Parties) without the consent of any other Party.

ARTICLE XIII MISCELLANEOUS

- 13.1 <u>Entire Agreement</u>. This Agreement contains the entire agreement between the Parties with respect to the subject matter hereof and supersedes all prior agreements, negotiations and understandings among the Parties with respect to such subject matter. Nothing in this Agreement shall be construed as creating a partnership or joint venture between the Parties.
- 13.2 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts and each such counterpart shall be deemed to be an original instrument, but all such counterparts together shall constitute one and the same agreement.
- 13.3 <u>Survival</u>. Cancellation, expiration or earlier termination of this Agreement shall not relieve the Parties of obligations that by their nature should survive such cancellation, expiration or termination, including remedies, limitations on liability, promises of indemnity and payment, and confidentiality. Without limiting the generality of the foregoing, the following provisions of this Agreement shall survive: <u>Articles III, VI, VII, X</u> and XI, and <u>Sections 13.3, 13.4, 13.5, 13.6, 13.8</u> and <u>13.9</u>.
- 13.4 <u>Severability.</u> In the event any one or more of the provisions contained in this Agreement should be held invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein shall not in any way be affected or impaired thereby. The Parties shall endeavor in good-faith negotiations to replace the invalid, illegal or unenforceable provisions with valid provisions, the economic and practical effect of which comes as close as possible to that of the invalid, illegal or unenforceable provisions.

- 13.5 <u>Governing Law.</u> This Agreement shall be governed by and construed in accordance with the laws of the State of New York without regard to the principles of conflicts of laws thereof.
- Binding Effect. This Agreement shall be binding upon and shall inure to the benefit of the Parties hereto and their respective successors and permitted assigns. This Agreement is not made for the benefit of any Person or entity not a party hereto, and nothing in this Agreement shall be construed as giving any Person or entity, other than the Parties and their respective successors and permitted assigns, any right, remedy or claim under or in respect of this Agreement or any provision hereof.
- Notices. All notices or other communications which are required or permitted hereunder shall be in writing and shall be deemed sufficiently given (a) upon delivery, if delivered personally, (b) the day the notice is received, if it is delivered by overnight courier or certified or registered mail, postage prepaid, or (c) upon the effective receipt of electronic transmission, facsimile, telex or telegram (with effective receipt being deemed to occur upon the sender's receipt of confirmation of successful transmission of such notice or communication), to the addresses set forth below or such other address as the addressee may have specified in a notice duly given to sender as provided herein:

If to Kinergy:

Kinergy Marketing, LLC 400 Capitol Mall Suite 2060 Sacramento, California 95814 Attention: Neil Koehler Telephone: (916) 403-2123 Facsimile: (916) 446-3936

with a copy to:

Kinergy Marketing, LLC c/o Pacific Ethanol, Inc. 400 Capital Mall Suite 2060 Sacramento, CA 95814 Attn: General Counsel

Telephone: (916) 403-2130 Facsimile: (916) 403-2785

Pacific Ethanol [_____], LLC c/o JT Miller Group LLC 777 Campus Commons Road #200 Sacramento, CA 95825 Attn: John Miller

Telephone: (916) 565-7422 Facsimile: (916) 565-7423

- 13.8 <u>Amendment</u>. No Party hereto shall be bound by any termination, amendment, supplement, waiver or modification of any term hereof unless such Party shall have consented thereto in writing.
- 13.9 <u>No Implied Waiver</u>. No delay or failure on the part of any Party in exercising any rights hereunder, and no partial or single exercise thereof, shall constitute a waiver of such rights or of any other rights hereunder.
- 13.10 <u>Lines of Credit</u>. Kinergy shall promptly notify Project Company if at any time during the term of this Agreement it does not have lines of credit available to it in an amount of not less than \$5,000,000 or if an event of default shall occur and be continuing under any agreement evidencing such lines of credit.

[THE REMAINDER OF THIS PAGE HAS BEEN LEFT BLANK INTENTIONALLY]

	PACI	FIC ETHANOL [], LLC
	Ву:	Name: Title:
	KINE	RGY MARKETING, LLC
	Ву:	/s/ Neil M. Koehler Name: Neil M. Koehler Title: Chief Executive Officer
[Signature	Page to Ethano	ol Marketing Agreement]

IN WITNESS WHEREOF, this Amended and Restated Ethanol Marketing Agreement has been duly executed by the Parties hereto as

of the date first written above.

Exhibit A

Form of PEI Guaranty

[Please see attached.]

Form of Parent Guaranty (Amended and Restated Ethanol Marketing Agreement)

[DATE]
Pacific Ethanol [], LLC c/o JT Miller Group LLC 777 Campus Commons Road # 200 Sacramento, CA 95825 Attention: John Miller
Re: Guaranty of Affiliate Project Party Obligations under Amended and Restated Ethanol Marketing Agreement
Ladies and Gentlemen:
1. The Guaranty. For value received, Pacific Ethanol, Inc., a corporation organized under the laws of Delaware (the "Guarantor"), hereby unconditionally and absolutely guarantees the prompt and complete payment and performance when due, whether by acceleration or otherwise, of all obligations and liabilities, whether now in existence or hereafter arising (the "Obligations"), of Kinergy Marketing, LLC, an Oregon limited liability company ("Affiliate Project Party"), to Pacific Ethanol [], LLC, a Delaware limited liability company ("Project Owner"), under the Amended and Restated Ethanol Marketing Agreement, dated as of, 2011 (the "Agreement"); provided that the Guarantor's maximum aggregate liability under this Guaranty shall not exceed the maximum amount which Affiliate Project Party owes Project Owner under the Agreement.
Following any demand by Project Owner for payment hereunder, the Guarantor shall pay, or cause to be paid, such Obligations within five (5) business days of receipt of such demand. Such payments shall be made to: [Insert Revenue Account Wire Instructions]
The Guarantor's obligations hereunder are primary obligations of the Guarantor and are an absolute, unconditional, continuing and irrevocable guaranty of payment and performance and not of collectability, and are in no way conditioned on or contingent upon any attempt to enforce in whole or in part any liabilities and obligations of Affiliate Project Party or any other person. Each failure by the Guarantor to pay or perform, as the case may be, any amounts due or any obligations under this Guaranty shall give rise to a separate cause of action hereunder, and separate suits may be brought hereunder as each cause of action arises.
2. <u>Waivers</u> ; Absolute Obligations. The Guarantor hereby waives notice of acceptance of this Guaranty and notice of any obligation or liability to which it may apply, and waives presentment, demand for payment, protest, notice of dishonor or non-payment of any such obligation or liability, suit or the taking of other action by Project Owner against, and any other notice to, Affiliate Project Party, the Guarantor or others, notice of entry into the Agreement between Affiliate Project Party and Project Owner and of any amendments, supplements or modifications thereto; or any waiver of consent under the Agreement, including waivers of the payment and performance of the obligations thereunder; and any requirement that suit be brought against, or any other action by Project Owner be taken against, or any notice of default or other notice be

given to, or any demand be made on Project Owner or any other person, or that any other action be taken or not taken as a condition to the Guarantor's liability for the Obligations under this Guaranty or as a condition to the enforcement of this Guaranty against the Guarantor.

Project Owner may at any time and from time to time without notice to or consent of the Guarantor and without impairing or releasing the obligations of the Guarantor hereunder: (1) agree with Affiliate Project Party to make any change in the terms of any obligation or liability of Affiliate Project Party to Project Owner, (2) take or fail to take any action of any kind in respect of any security for any obligation or liability of Affiliate Project Party to Project Owner, (3) exercise or refrain from exercising any rights against Affiliate Project Party or others, or (4) compromise or subordinate any obligation or liability of Affiliate Project Party to Project Owner including any security therefor.

The liability of the Guarantor under this Guaranty shall be absolute and unconditional irrespective of:

- (a) any lack of validity or enforceability of or defect or deficiency applicable to Affiliate Project Party in the Agreement or any other documents executed in connection with the Agreement; or
- (b) any modification, extension or waiver of any of the terms of the Agreement; or
- (c) any change in the time, manner, terms or place of payment of or in any other term of, all or any of the Obligations, or any other amendment or waiver of or any consent to departure from the Agreement or any other agreement or instrument executed in connection therewith: or
- (d) failure, omission, delay, waiver or refusal by Project Owner to exercise, in whole or in part, any tight or remedy held by Project Owner with respect to the Agreement or any transaction under the Agreement; or
- (e) any change in the existence, structure or ownership of the Guarantor or Affiliate Project Party or Project Owner, or any insolvency, bankruptcy, reorganization or other similar proceeding affecting Affiliate Project Party or its assets; or
- any defense arising by reason of any bankruptcy, insolvency, reorganization, arrangement, readjustment of debt, liquidation or dissolution proceeding commenced by or against any person, including any discharge of, or bar or stay against collecting, all or any part of the amounts due under this Guaranty (or any interest on all or any part of the amounts due under this Guaranty) in or as a result of any such proceeding, any failure of Project Owner to file a claim in any such proceeding, or the occurrence of any of the following: (i) the election by Project Owner, in any bankruptcy proceeding of any person, of the application or non-application of Section 1111(b)(2) of Title 11 of the United States Code entitled "Bankruptcy" (together with any successor statute, and all rules promulgated thereunder, the "Bankruptcy Code"), (ii) any extension of credit or the grant of any lien or encumbrance under Section 364 of the Bankruptcy Code, (iii) any use of cash collateral under Section 363 of the Bankruptcy Code, or (iv) any agreement or stipulation with respect to the provision of adequate protection in any bankruptcy proceeding of any person; or

any duty on the part of Project Owner to disclose to the Guarantor any facts Project Owner may now or hereafter know about Affiliate Project Party, regardless of whether Project Owner has reason to believe that any such facts materially increase the risk beyond that which the Guarantor intends to assume, or have reason to believe that such facts are unknown to the Guarantor, or have a reasonable opportunity to communicate such facts to the Guarantor, since the Guarantor acknowledges that the Guarantor is fully responsible for being and keeping informed of the financial condition of Affiliate Project Party and of all circumstances beating on the risk of non-payment of any amounts due or non-performance of any obligations under this Guarantor.

Without limiting the-foregoing, the Guarantor hereby unconditionally and irrevocably waives and relinquishes, to the maximum extent permitted by applicable laws, all rights and remedies accorded to sureties or guarantors and agrees not to assert or take advantage of any such rights or remedies.

- 3. Bankruptcy Waivers. The Guarantor hereby irrevocably waives, to the extent it may do so under applicable laws, any protection to which it may be entitled under Sections 365(c)(1), 365(c)(2) and 365(e)(2) of the Bankruptcy Code or equivalent provisions of the laws or regulations of any other jurisdiction with respect to any proceedings, or any successor provision of law of similar import, in the event of any Act of Insolvency (as defined in the Agreement) with respect to Project Owner or Affiliate Project Party or any other guarantor or surety. Specifically, in the event that the trustee (or similar official) in an Act of Insolvency with respect to Affiliate Project Party or any other guarantor or surety or the debtor-in-possession takes any action (including the institution of any action, suit or other proceeding for the purpose of enforcing the rights of the relevant Borrower(s), or any other guarantor or surety under this Guaranty), the Guarantor shall, to the fullest extent it may do so under applicable law, not assert any defense, claim or counterclaim denying liability hereunder on the basis that this Guaranty is an executory contract or a "financial accommodation" that cannot be assumed, assigned or enforced or on any other theory directly or indirectly based on Sections 365(c)(1), 365(c)(2) or 365(e)(2) of the Bankruptcy Code, or equivalent provisions of the law or regulations of any other jurisdiction with respect to any proceedings or any successor provision of law of similar import. If an Act of Insolvency with respect to Affiliate Project Party or any other guarantor or surety shall occur, the Guarantor agrees, after the occurrence of such Act of Insolvency, to reconfirm in writing, to the extent permitted by applicable laws, its pre-petition waiver of any protection to which it may be entitled under Sections 365(c)(1), 365(c)(2) and 365(e)(2) of the Bankruptcy Code or equivalent provisions of the laws or regulations of any other jurisdiction with respect to proceedings and, to give effect to such waiver, the Guarantor consents, to the fullest extent it may do so under applicable law, to the assumption and enforcement of each provision of this Guaranty by the debtor-in-possession or the trustee in bankruptcy of Affiliate Project Party or of any other guarantor or surety, as the case may be.
- 4. <u>No Assignment.</u> The Guarantor may not assign its rights nor delegate its obligations under this Guaranty, in whole or in part, without prior written consent of Project Owner, and any purported assignment or delegation absent such consent is void, except for an assignment and delegation of all of the Guarantor's rights and obligations hereunder in whatever form the Guarantor determines may be appropriate to a partnership, corporation, trust or other organization in whatever form that succeeds to all or substantially all of the Guarantor's assets and business and that assumes such obligations by contract, operation of law or otherwise. Upon any such delegation and assumption of obligations, the Guarantor shall be relieved of and fully discharged from all obligations hereunder, whether such obligations arose before or after such delegation and assumption. Project Owner may, upon notice to the Guarantor, assign its rights hereunder (including any collateral assignment) without the consent of Guarantor to any person to which it assigns its interests in the Agreement.

- 5. Representations and Warranties. The Guarantor hereby represents and warrants for itself, as of the date hereof, that:
 - (a) it is duly organized and validly existing under the laws of its jurisdiction of incorporation, has the corporate power and has obtained all required governmental approvals to comply with and perform its respective obligations under and enter into this Guaranty;
 - (b) this Guaranty has been duly authorized and executed by it and constitutes its valid and legally binding obligation enforceable in accordance with its terms, except as enforceability hereof may be limited by bankruptcy, moratorium, insolvency or other similar laws affecting the enforcement of creditor's rights generally;
 - (c) neither the execution and delivery of this Guaranty nor the compliance with its terms will conflict with or result in a breach of any of the terms, conditions or provisions of, or constitute a default or require any consent which has not been obtained under, any indenture, mortgage, agreement or other instrument or arrangement to which it is a party or by which it or any of its properties or assets are bound, or violate any of the terms or provisions of its organizational documents (including its bylaws) or any governmental approval, judgment, decree or order or any other applicable law;
 - (d) it is, and after giving effect to the transactions contemplated under this Guaranty will be, solvent; and
 - (e) it is not executing this Guaranty with any intention to hinder, delay or defraud any of its present or future creditor or creditors.
- 6. <u>Subrogation</u>. Notwithstanding any payment or payments made by the Guarantor or the exercise by Project Owner of any of the remedies provided under this Guaranty or any set-off or application of funds of the Guarantor by Project Owner, the Guarantor hereby waives all rights of subrogation with respect to payments made under this Guaranty until all of the Obligations have been paid in full. Notwithstanding the foregoing, if any amount shall be paid to the Guarantor on account of such subrogation, such amount shall be held by the Guarantor in trust for Project Owner, segregated from other funds of the Guarantor, and shall be turned over to Project Owner, in the exact form received by the Guarantor (or duly endorsed by the Guarantor to Project Owner, if required) to be applied against such amounts in such order as Project Owner may elect.

7. <u>Notices</u>. All demands, notices and other communications provided for hereunder shall, unless otherwise specifically provided herein, (a) be in writing addressed to the party receiving the notice at the address set forth below or at such other address as may be designated by written notice, from time to time, to the other party, and (b) be effective upon receipt, when mailed by U.S. mail, registered or certified, return receipt requested, postage prepaid, facsimile or personally delivered. Notices shall be sent to the following addresses:

If to Project Owner:

Pacific Ethanol [_____], LLC c/o JT Miller Group LLC 777 Campus Commons Road # 200 Sacramento, CA 95825Attention: John Miller

If to Guarantor:

Pacific Ethanol, Inc. 400 Capitol Mall, Suite 2060 Sacramento, California 95814 Attn: Neil Koehler

Phone: (916) 403-2123 Fax: (916) 446-3937

with a copy to

Pacific Ethanol, Inc. 400 Capital Mall, Suite 2060 Sacramento, CA 95814 Attn: General Counsel Fax: (916) 446-3936

- 8. <u>Cumulative Remedies; Benefits</u>. No failure by Project Owner to exercise, and no delay by Project Owner in exercising, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law. The remedies herein provided are cumulative and not exclusive of any remedies provided by law. Nothing in this Guaranty, express or implied, shall give to any person, other than the parties hereto, and each of their successors and permitted assigns under this Guaranty, any benefit or any legal or equitable right or remedy under this Guaranty.
- 9. <u>Reinstatement.</u> This Guaranty shall continue to be effective or be reinstated, as the case may be, if at any time any payment of any of the Obligations are annulled, set aside, invalidated, declared to be fraudulent or preferential, rescinded or must otherwise be returned, refunded or repaid by Project Owner upon the insolvency, bankruptcy, dissolution, liquidation or reorganization of Affiliate Project Party or any other guarantor, or upon or as a result of the appointment of a receiver or conservator of, or trustee for Affiliate Project Party or any other guarantor or any substantial part of its property or otherwise, all as though such payment or payments had not been made.

- 10. <u>Governing Law, etc.</u> (A) THIS GUARANTY SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.
- (B) EACH PARTY HERETO IRREVOCABLY AND UNCONDITIONALLY SUBMITS, FOR ITSELF AND ITS PROPERTY, TO THE NON-EXCLUSIVE JURISDICTION OF THE COURTS OF THE STATE OF NEW YORK SITTING IN OR NEAR NEW YORK COUNTY AND OF THE UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF NEW YORK, AND ANY APPELLATE COURT FROM ANY THEREOF, IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS GUARANTY OR FOR RECOGNITION OR ENFORCEMENT OF ANY JUDGMENT, AND EACH OF THE PARTIES HERETO IRREVOCABLY AND UNCONDITIONALLY AGREES THAT ALL CLAIMS IN RESPECT OF ANY SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN SUCH STATE OR, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, IN SUCH FEDERAL COURT. EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW. NOTHING IN THIS GUARANTY SHALL AFFECT ANY RIGHT THAT ANY OF THE PARTIES HERETO MAY OTHERWISE HAVE TO BRING ANY ACTION OR PROCEEDING RELATING TO THIS GUARANTY IN THE COURTS OF ANY JURISDICTION.
- (C) EACH PARTY HERETO IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS GUARANTY. EACH PARTY HERETO IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, THE DEFENSE OF AN INCONVENIENT FORUM TO THE MAINTENANCE OF SUCH ACTION OR PROCEEDING IN ANY SUCH COURT.
- 11. <u>Termination</u>. This Guaranty shall terminate on the earlier of (i) date the Agreement terminates in accordance with the terms thereof and (ii) the date on which the Affiliate Project Party ceases to be an Affiliate of Guarantor.
- 12. <u>Survivability</u>. If any provision of this Guaranty is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Guaranty shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

13. <u>Counterparts.</u> This Guaranty may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Guaranty shall become effective when it has been executed by each of the parties hereto. Delivery of an executed counterpart of a signature page of this Guaranty by telecopy or
portable document format ("pdf") shall be effective as delivery of a manually executed counterpart of this Guaranty.
[Signature Page Follows]

Very truly yours,

PACIFIC ETHANOL, INC.

By:

Name: Neil M. Koehler
Title: Chief Executive Officer

Acknowledged and Accepted by:

PACIFIC ETHANOL [_____], LLC

By:

Name:
Title

Name:
Title

Exhibit B

Operating Protocol

Pacific Ethanol [], LLC ([]) will by the end of day Monday each week submit to Kinergy Marketing, LLC (Kinergy) a production forecast, (see attached example) that includes weekly production and scheduled maintenance downtime. [] will promptly Kinergy of any unscheduled downtime or other event or circumstance that will affect forecasted production.	
Kinergy will by end of day Tuesday each week submit to [] a 12 week sales forecast in a format similar to the production forecast. will by end of day Friday each week submit to [] a weekly Truck and Barge schedule for the following week. Changes to the week and barge schedule will be communicated to [] as the changes are identified.	٠.

Exhibit C

Certain Customers

Name	Address	Address 2	City	State	ZIP Code
ABI Motor Sports	P.O. Box 3151		Clovis	CA	93613
Archer Daniels Midland	P.O. Box 1470		Decatur	IL	62525
Co					
Arizona Petroleum	DBA: Colorado Petroleum	4080 Globeville Road	Denver	CO	80216
ArmorStruxx LLC	PO Box 2060		Lodi	CA	95241
Apex Oil Co Inc	23901 Calabasas Road Ste 2090		Calabasas	CA	91302
BP West Coast Products LLC	6 Centerpointe Drive		La Palma	CA	90623-2502
BioUrja Trading LLC	757 N. Eldridge Pkwy Ste 620		Houston	Tx	77079
Big West Oil LLC	P.O. Box 150310		Ogden	UT	84415
Stan Boyett and Son Inc	DBA: Boyett Petroleum	601 McHenry Ave	Modesto	CA	95350
CAF LLC	P.O. Box 1121		Maple Valley	WA	98038
Chevron Products	6000 Bollinger Canyon Rd		San Ramon	CA	94583
CHS Inc.	5500 Cenex Dr		Inner Grove Heights	MN	55077
Conoco Phillips	600 North Dairy Ashford		Houston	TX	77079
Coast Oil Company	4250 Williams Rd		San Jose	CA	95129
DAT's Trucking Inc	321 N Old Hwy 91		Hurricane	UT	84737
ECO Energy	730 Cool Springs Blvd, Ste 130		Franklin	TN	37067
ExxonMobil Oil Corporation	3225 Gallows Rd RM 3B1417		Fairfax	VA	22037-00004
Gavilon Group LLC - Conagra	11 ConAgra Dr Ste 5021		Omaha	NE	68102
Holly Refining and Marketing	Company, LLC	100 Crescent Court, Suite 1600	Dallas	TX	75201
Hunt & Sons, Inc.	5750 S Watt Ave		Sacramento	CA	95829
Idemitsu Apollo Corporation	1831 16th Street		Sacramento	CA	95814
IPC (USA) Inc.	20 Pacifica Ste 650		Irvine	CA	92618
Jaco Oil Co.	P O Box 82515		Bakersfield	CA	93380-2515
Kempler & Co Inc	233 Broadway Ste 2705		New York	New Yor	k 10279
Kern Oil & Refining Co.	Attn: Gloria Gutierrez	7724 East Panama Lane	Bakersfield	CA	93307
Lansing Ethanol Services		Ste 400	Overland Park	KS	66210
LiquidTitan LLC	139 South Marina Street		Prescott	AZ	86303
Maverik Inc	P.O. Box 8008	1014 S Washington	Afton	WY	83110-8008
Mieco Inc	301 East Ocean Blvd Ste 1100		Long Beach	CA	90802

Montana Refining 1900 10th St NE Great Falls MT 59404	Mansfield Oil Company of	Gainesville Inc	1025 Airport Pkwy SW	Gainesville	Georgia	30501-6813
Noble Americas Corp 333 Ludlow St Suite 1230 Stamford CT 6902 Nella Oil Co., LLC 2360 Lindbergh Street Auburn CA 95602 Navajo Refining 100 Crescent Court Suite 1600 Dallas TX 75201 Torpal Lic Torpa	Montana Refining	1900 10th St NE		Great Falls	MT	59404
Nella Oil Co., LLC 2360 Lindbergh Street Auburn CA 95602 Navajo Refining 100 Crescent Court Suite 1600 Dallas TX 75201 Company LLC New West Petroleum 1831 16Th Street Sacramento CA 95811 Propel Biofuels Inc 2317 Broadway Ste 350 Redwood City CA 94063 Petro Diamond Inc. 18401 Von Karman Ave Suite 300 Irvine Ca 92612 Pro Petroleum Inc. Attn Lisa Owens P O Box 10128 Lubbock TX 79416 QuikTrip Corp. PO Box 3475 Tulsa OK 74101 River City Petroleum P O Box 235 840 Delta Lane West Sacramento CA 95691 Rehel Oil 2200 South Highland Las Vegas NV 89102 Renova Energy LLC 950 W Bannock St Ste 500 Boise Idaho 83702 R.H. Smith Distributing 315 E. Wine Country Road Grandview WA 98930 Co ROBINSON Oil 955 Martin Avenue Santa Clara CA 95050 RPMG Inc 1157 Valley Park Dr #100 Shakopee MN 55379 Southern Counties Oil P O Box 4159 Orange CA 92863-4159 Co. Silver Eagle Refining Inc 2355 So 1100 West Woods Cross Utah 84087 Shell Oil Products U.S. 909 Fannin Plaza Level1 Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102 Sollet LLC DBA Deallet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. PO Box 7627 Boise ID 83707 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy Tower Energy Group 198 West 190th St. Suite 100 Torrance CA 90504	Murex N.A. LTD	5057 Keller Springs Rd	Suite 150	Addison	TX	75001
Navajo Refining 100 Crescent Court Suite 1600 Dallas TX 75201	Noble Americas Corp	333 Ludlow St Suite 1230		Stamford	CT	6902
Company LLC	Nella Oil Co., LLC	2360 Lindbergh Street		Auburn	CA	95602
New West Petroleum 1831 16Th Street Sacramento CA 95811	Navajo Refining	100 Crescent Court	Suite 1600	Dallas	TX	75201
Propel Biofuels Inc 2317 Broadway Ste 350 Redwood City CA 94063	Company LLC					
Petro Diamond Inc. 18401 Von Karman Ave Suite 300 Irvine Ca 92612	New West Petroleum	1831 16Th Street		Sacramento	CA	95811
Pro Petroleum Inc. Attn Lisa Owens P O Box 10128 Lubbock TX 79416 QuikTrip Corp. PO Box 3475 Tulsa OK 74101 River City Petroleum P O Box 235 840 Delta Lane West Sacramento CA 95691 Rebel Oil 2200 South Highland Las Vegas NV 89102 Renova Energy LLC 950 W Bannock St Ste 500 Boise Idaho 83702 R.H. Smith Distributing 315 E. Wine Country Road Grandview WA 98930 Co WA 98930 WA 98930 Co Santa Clara CA 95050 RPMG Inc 1157 Valley Park Dr #100 Shakopee MN 55379 Southern Counties Oil P O Box 4159 Orange CA 92863-4159 Co Woods Cross Utah 84087 Shell Oil Products U.S. 909 Fannin Plaza Level1 Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102	Propel Biofuels Inc	2317 Broadway Ste 350		Redwood City	CA	94063
QuikTrip Corp. PO Box 3475 Tulsa OK 74101 River City Petroleum P O Box 235 840 Delta Lane West Sacramento CA 95691 Rebel Oil 2200 South Highland Las Vegas NV 89102 Renova Energy LLC 950 W Bannock St Ste 500 Boise Idaho 83702 R.H. Smith Distributing 315 E. Wine Country Road Grandview WA 98930 Co Robinson Oil 955 Martin Avenue Santa Clara CA 95050 RPMG Inc 1157 Valley Park Dr #100 Shakopee MN 55379 Southern Counties Oil P O Box 4159 Orange CA 92863-4159 Co. Silver Eagle Refining Inc 2355 So 1100 West Woods Cross Utah 84087 Shell Oil Products U.S. 909 Fannin Plaza Level1 Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102 SolJet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 </td <td>Petro Diamond Inc.</td> <td>18401 Von Karman Ave</td> <td>Suite 300</td> <td>Irvine</td> <td>Ca</td> <td>92612</td>	Petro Diamond Inc.	18401 Von Karman Ave	Suite 300	Irvine	Ca	92612
River City Petroleum P O Box 235 840 Delta Lane West Sacramento CA 95691 Rebel Oil 2200 South Highland Las Vegas NV 89102 Renova Energy LLC 950 W Bannock St Ste 500 Boise Idaho 83702 R.H. Smith Distributing 315 E. Wine Country Road Grandview WA 98930 Co WA 98930 Santa Clara CA 95050 RPMG Inc 1157 Valley Park Dr #100 Shakopee MN 55379 Southern Counties Oil P O Box 4159 Orange CA 92863-4159 Co. Silver Eagle Refining Inc 2355 So 1100 West Woods Cross Utah 84087 Shell Oil Products U.S. 909 Fannin Plaza Level1 Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102 Sollet LLC DBA Deallet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 <tr< td=""><td>Pro Petroleum Inc.</td><td>Attn Lisa Owens</td><td>P O Box 10128</td><td>Lubbock</td><td>TX</td><td>79416</td></tr<>	Pro Petroleum Inc.	Attn Lisa Owens	P O Box 10128	Lubbock	TX	79416
Rebel Oil 2200 South Highland Las Vegas NV 89102 Renova Energy LLC 950 W Bannock St Ste 500 Boise Idaho 83702 R.H. Smith Distributing 315 E. Wine Country Road Grandview WA 98930 Co W C WA 98930 RPMG Inc 1157 Valley Park Dr #100 Santa Clara CA 95050 RPMG Inc 1157 Valley Park Dr #100 Shakopee MN 55379 Southern Counties Oil P O Box 4159 Orange CA 92863-4159 Co. Silver Eagle Refining Inc 2355 So 1100 West Woods Cross Utah 84087 Shell Oil Products U.S. 909 Fannin Plaza Level1 Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102 Sollet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 Self-Serve Petroleum 1045 Airport Blvd	QuikTrip Corp.	PO Box 3475		Tulsa	OK	74101
Renova Energy LLC 950 W Bannock St Ste 500 Boise Idaho 83702 R.H. Smith Distributing 315 E. Wine Country Road Grandview WA 98930 Co WA 98930 Robinson Oil 955 Martin Avenue Santa Clara CA 95050 RPMG Inc 1157 Valley Park Dr #100 Shakopee MN 55379 Southern Counties Oil P O Box 4159 Orange CA 92863-4159 Co. Silver Eagle Refining Inc 2355 So 1100 West Woods Cross Utah 84087 Shell Oil Products U.S. 909 Fannin Plaza Level1 Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102 Sollet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 84717 San Diego	River City Petroleum	P O Box 235	840 Delta Lane	West Sacramento	CA	95691
R.H. Smith Distributing 315 E. Wine Country Road Co Robinson Oil 955 Martin Avenue Santa Clara CA 95050 RPMG Inc 1157 Valley Park Dr #100 Shakopee MN 55379 Southern Counties Oil P O Box 4159 Co. Silver Eagle Refining Inc 2355 So 1100 West Woods Cross Utah 84087 Shell Oil Products U.S. 909 Fannin Plaza Level1 Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102 SolJet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 7627 Boise ID 83707 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Rebel Oil	2200 South Highland		Las Vegas	NV	89102
Co Robinson Oil 955 Martin Avenue Santa Clara CA 95050 RPMG Inc 1157 Valley Park Dr #100 Shakopee MN 55379 Southern Counties Oil P O Box 4159 Orange CA 92863-4159 Co. Silver Eagle Refining Inc 2355 So 1100 West Woods Cross Utah 84087 Shell Oil Products U.S. 909 Fannin Plaza Level1 Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102 SolJet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 7627 Boise ID 83707 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy San Antonio TX 78259 Marketing <	Renova Energy LLC	950 W Bannock St Ste 500		Boise	Idaho	83702
Robinson Oil 955 Martin Avenue Santa Clara CA 95050 RPMG Inc 1157 Valley Park Dr #100 Shakopee MN 55379 Southern Counties Oil P O Box 4159 Orange CA 92863-4159 Co. Silver Eagle Refining Inc 2355 So 1100 West Woods Cross Utah 84087 Shell Oil Products U.S. 909 Fannin Plaza Level Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102 SolJet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 7627 Boise ID 83707 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	R.H. Smith Distributing	315 E. Wine Country Road		Grandview	WA	98930
RPMG Inc 1157 Valley Park Dr #100 Shakopee MN 55379 Southern Counties Oil P O Box 4159 Orange CA 92863-4159 Co. Silver Eagle Refining Inc 2355 So 1100 West Woods Cross Utah 84087 Shell Oil Products U.S. 909 Fannin Plaza Level1 Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102 SolJet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 7627 Boise ID 83707 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Co					
Southern Counties Oil P O Box 4159 Co. Silver Eagle Refining Inc 2355 So 1100 West Shell Oil Products U.S. 909 Fannin Sinclair Oil Corporation 550 E South Temple SolJet LLC DBA DealJet DBA DealJet 125 N 53rd Ave Phoenix South San Francisco CA 92863-4159 Woods Cross Utah 84087 TX 77010 Salt Lake City UT 84102 SolJet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 7627 Boise ID 83707 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Robinson Oil	955 Martin Avenue		Santa Clara	CA	95050
Co. Silver Eagle Refining Inc 2355 So 1100 West Shell Oil Products U.S. 909 Fannin Plaza Level1 Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102 SolJet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 7627 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	RPMG Inc	1157 Valley Park Dr #100		Shakopee	MN	55379
Silver Eagle Refining Inc 2355 So 1100 West Shell Oil Products U.S. 909 Fannin Plaza Level1 Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102 SolJet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 7627 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Southern Counties Oil	P O Box 4159		Orange	CA	92863-4159
Shell Oil Products U.S. 909 Fannin Plaza Level1 Houston TX 77010 Sinclair Oil Corporation 550 E South Temple Salt Lake City UT 84102 SolJet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 7627 Boise ID 83707 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Co.					
Sinclair Oil Corporation 550 E South Temple SolJet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 7627 Boise ID 83707 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Silver Eagle Refining Inc	2355 So 1100 West		Woods Cross	Utah	84087
SolJet LLC DBA DealJet 125 N 53rd Ave Phoenix AZ 85043 Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 7627 Boise ID 83707 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy San Antonio TX 78259 Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Shell Oil Products U.S.	909 Fannin	Plaza Level1	Houston	TX	77010
Space Age Fuel Inc. P.O. Box 1429 Clackamas OR 97015 Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 7627 Boise ID 83707 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy San Antonio TX 78259 Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Sinclair Oil Corporation	550 E South Temple		Salt Lake City	UT	84102
Self-Serve Petroleum 1045 Airport Blvd South San Francisco CA 94080 Stinker Stores Inc. PO Box 7627 Boise ID 83707 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy San Antonio TX 78259 Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	SolJet LLC	DBA DealJet	125 N 53rd Ave	Phoenix	ΑZ	85043
Stinker Stores Inc. PO Box 7627 Boise ID 83707 Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy San Antonio TX 78259 Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Space Age Fuel Inc.	P.O. Box 1429		Clackamas	OR	97015
Supreme Oil Co. PO Box 84717 San Diego CA 92138 Tesoro Refining and 19100 Ridgewood Pkwy San Antonio TX 78259 Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Self-Serve Petroleum	1045 Airport Blvd		South San Francisco	CA	94080
Tesoro Refining and 19100 Ridgewood Pkwy Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Stinker Stores Inc.	PO Box 7627		Boise	ID	83707
Marketing Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Supreme Oil Co.	PO Box 84717		San Diego	CA	92138
Tower Energy Group 1983 West 190th St. Suite 100 Torrance CA 90504	Tesoro Refining and	19100 Ridgewood Pkwy		San Antonio	TX	78259
** *	Marketing					
Valero Marketing P O Box 696000 San Antonio TX 78269-6000	Tower Energy Group	1983 West 190th St.	Suite 100	Torrance	CA	90504
	Valero Marketing	P O Box 696000		San Antonio	TX	78269-6000
Vitol Incorporated 1100 Louisana Suite 5500 Houston Texas 77002	Vitol Incorporated	1100 Louisana Suite 5500		Houston	Texas	77002
Wilcox & Flegel Oil Co PO Box 69 Longview WA 98632		PO Box 69		Longview	WA	98632
Western Ethanol 1075 Yorba Place #101 Placentia CA 92870	Western Ethanol	1075 Yorba Place #101		Placentia	CA	92870
Company LLC						
Western Refining Co LP 123 W Mills Ave Ste 200 El Paso Texas 79901		123 W Mills Ave Ste 200		El Paso		79901
Western Energetix 2360 Lindbergh Street Auburn CA 95602	Western Energetix	2360 Lindbergh Street		Auburn	CA	95602

Exhibit D

Benchmarking

BENCHMARK

We propose the following example benchmarks and monitor criteria. Modifications to the benchmarks are expected to be made periodically as appropriate to reflect with changes in commodity market environments and in company purchase/sales contract pricing mechanisms.

•	ility PEHC actual data	Benchmark	Market Data Provider		Possible Discrepancy Reasons/Noises
ETHANOL					
PECOL	Netback ethanol	OPIS Pacific Northwest	Posting as provided by	a)	Fixed price contracts
	sales price		Oil Price	b)	Other Price index
PEMV	Netback ethanol	OPIS Chicago	Information Service	c)	Shipments distribution in a highly volatile market
	sales price		(OPIS)	d)	Other accounting adjustments items
PES	Netback ethanol sales price	OPIS LA CI-90			
CORN					
ALL	Corn CBOT	Delivered basis = FOB	Trade West report	a)	Fixed futures as a result of hedge
	equivalent	Midwest + freight	·	b)	Contracts based on corn futures other than nearby futures (Usually, this happens a couple of weeks before current future contract expires
				c)	Uneven distribution of corn consumptions
					(such as shut down) in a highly volatile market
WDG/Syrup					
ALL	Plant co-product	WDG Value as a % of	FC Stone Co-product	a)	Fixed price contracts
	netback	Plant delivered corn on	report	b)	Hedge activities
		DDG equivalent basis	^	c)	WDG freight

ETHANOL

	Plant Sales Netback		Basis (netback)	Benchmark	Actual vs. Benchmark
PECOL	reciback	OPIS PNW	(Hetback)		Denemiark
Jan-11	2.42	2.42	(0.04)	2.38	1.4%
Feb-11	2.45	2.50	(0.04)	2.46	-0.2%
Mar-11	2.60	2.63	(0.04)	2.59	0.5%
Apr-11	2.73	2.76	(0.04)	2.72	0.2%
•					
PEMV		OPIS Chicago			
Jan-11	2.38	2.32	0.01	2.33	2.1%
Feb-11	2.47	2.43	0.01	2.44	1.5%
Mar-11	2.54	2.50	0.01	2.51	1.2%
Apr-11	2.65	2.65	0.01	2.66	-0.2%
PES		OPISLA CI-90			
Jan-11	2.47	2.46	(0.01)	2.45	1.0%
Feb-11	2.57	2.53	(0.01)	2.52	2.1%
Mar-11	2.68	2.66	(0.01)	2.65	0.9%
Apr-11	2.80	2.80	(0.01)	2.79	0.1%

CORN

	Plant CBOT Equivalent	Plant basis	Plant Actual Corn dlvd \$/bu	Market	Actual vs. Market
PECOL					
Jan-11	6.27	0.563	6.83	7.18	(0.35)
Feb-11	6.56	0.595	7.15	7.74	(0.59)
Mar-11	6.68	0.519	7.20	7.64	(0.44)
Apr-11	7.22	0.496	7.71	8.35	(0.64)
PEMV					
Jan-11	6.34	0.650	6.99	7.04	(0.06)
Feb-11	6.87	0.650	7.52	7.60	(0.08)
Mar-11	6.90	0.563	7.46	7.50	(0.04)
Apr-11	7.38	0.639	8.02	8.20	(0.18)
PES					
Jan-11	6.22	0.656	6.88	7.18	(0.30)
Feb-11	6.76	0.663	7.42	7.74	(0.32)
Mar-11	6.74	0.619	7.36	7.65	(0.28)
Apr-11	7.21	0.687	7.90	8.36	(0.46)

CO PRODUCTS

	Plant Co-product \$/ton Netback	Plant Co-product DM	Plant Actual Corn dlvd \$/bu	Plant Actual Corn dlvd \$/ton	WDG Value % of Corn (1)
PECOL	ψ/toll I (teback	<i>D</i> 111	αινα φισα	αινα φ/τοιι	70 01 C0111 (1)
Jan-11	56.62	30%	6.83	243.91	69.2%
Feb-11	56.85	30%	7.15	255.37	66.4%
Mar-11	60.80	31%	7.20	257.06	68.0%
Apr-11	61.97	31%	7.71	275.44	65.3%
PEMV					
Jan-11	61.90	32%	6.99	249.48	70.5%
Feb-11	67.04	32%	7.52	268.48	71.0%
Mar-11	69.53	32%	7.46	266.45	74.3%
Apr-11	71.74	32%	8.02	286.40	70.5%
PES					
Jan-11	59.21	33%	6.88	245.68	65.4%
Feb-11	59.94	33%	7.42	264.98	61.4%
Mar-11	67.17	33%	7.36	262.95	70.3%
Apr-11	71.47	32%	7.90	282.00	70.5%

⁽¹⁾DDG equivalent, adjusted to standard 90% dry matter

[FORM	OF]

AMENDED AND RESTATED

[____] AGREEMENT

by and between

PACIFIC ETHANOL [____]

and

PACIFIC AG. PRODUCTS, LLC

Dated as of June 30, 2011

TABLE OF CONTENTS

ARTICLE 1	DEFINITIONS	1
1.1	Definitions	1
1.2	Interpretation	5
A DETICI E I	H. A CDEEMENT	e
	II AGREEMENT	5
2.1	Agreement	5
2.2	Entire Agreement	6
ARTICLE 1	III CORN PROCUREMENT	6
3.1	Appointment and Acceptance; Performance of Obligations.	6
3.2	Limitations on Authority.	6
3.3	Obligations of [Plant Owner]	7
3.4	Transaction Reports	7
3.5	Delivery Point; Operating Protocol; Nominations; Measurement.	7
3.6	Benchmarking	8
ARTICLE 1	IV COMPENSATION AND PAYMENT	8
4.1	Fees and Payments.	8
4.2	Overdue Payments; Indemnity Payments.	9
4.3	Billing Dispute	9
4.4	Audit	9
A DEICLE	N/ TEDM	10
ARTICLE ' 5.1		10
	Term	10
5.2	[Plant Owner] Defaults and PAP Remedies	10
5.3	PAP Defaults and [Plant Owner] Remedies	10
5.4	Change of Control	11
5.5	Replacement	11
ARTICLE '	VI INSURANCE	11
6.1	PAP Insurance	11
6.2	PAP Insurance Premiums and Deductibles	12
	VII INDEMNIFICATION	13
7.1	[Plant Owner]'s Indemnity	13
7.2	PAP's Indemnity	13
ARTICLE '	VIII LIABILITIES OF THE PARTIES	13
8.1	No Consequential or Punitive Damages	13
0.1	110 Consequential of Luminive Damages	13

i

ARTICLE IX	CONFIDENTIALITY CONFIDENTIALITY	13
ARTICLE X	FORCE MAJEURE	14
10.1	Events Constituting Force Majeure	14
10.2	Effect	14
10.3	Limitations	14
ADTICI E V	I DISPUTE RESOLUTION	15
11.1	Attempts to Settle	15
11.2	Resolution by Expert	15
11.3	Arbitration	15
11.4	Consequential and Punitive Damages	15
11.5	Finality and Enforcement of Decision	16
11.6	Costs	16
11.7	Continuing Performance Obligations	16
	II MISCELLANEOUS PROVISIONS	16
12.1	Assignment	16
12.2	Cooperation in Financing	16
12.3	Not for Benefit of Third Parties	16
12.4	Amendments	16
12.5	Survival	16
12.6	No Waiver	17
12.7	Notices	17
12.8	Representations and Warranties.	17
12.9	Counterparts and Execution	19
12.10	Governing Law	19
12.11	Severability	19
12.12	Successors and Assigns	19
12.13	Captions; Appendices	19
EXHIBIT		
Exhibit A:	Specifications	
Exhibit B:	Form of PEI Guaranty	
Exhibit C:	Operating Protocol	
Exhibit D:	Benchmarking	

This AMENDED AND RESTATED CORN PROCUREMENT AND HANDLING AGREEMENT (this " <u>Agreement</u> ") is made and entered into as of June 30, 2011 by and between Pacific Ethanol [], a Delaware limited liability company (" <u>Plant Owner</u>]"), and Pacific Ag. Products, LLC, a California limited liability company (" <u>PAP</u> "). [Plant Owner] and PAP are each individually referred to herein as a "Party", and collectively are referred to herein as the " <u>Parties</u> ".
RECITALS
A. The Parties hereto were previously party to that certain Corn Procurement and Handling Agreement, dated as of [], 2010 (the "Prior Agreement"), pursuant to which PAP provided certain services to [Plant Owner].
B. PAP provides grain services for denatured fuel ethanol production facilities.
C. [] owns an approximately [] million gallons-per-year denatured fuel ethanol production facility in [], (the "Facility") and has requested that PAP procure corn for the Facility.
D. The Parties desire to amend and restate in its entirety the Prior Agreement and enter into this Amended and Restated Corn Procurement and Handling Agreement pursuant to which PAP will provide such procurement and handling services.
AGREEMENT
NOW, THEREFORE, in consideration of the agreements and covenants hereinafter set forth, and intending to be legally bound, the Parties hereto covenant and agree as follows:
ARTICLE I DEFINITIONS
1.1 <u>Definitions</u> . The following terms shall have the meanings set forth below when used in this Agreement:
"Act of Insolvency" means, with respect to any Person, any of the following: (a) commencement by such Person of a voluntary proceeding under any jurisdiction's bankruptcy, insolvency or reorganization law; (b) the filing of an involuntary proceeding against such Person under any jurisdiction's bankruptcy, insolvency or reorganization law which is not vacated within 60 days after such filing; (c) the admission by such Person of the material allegations of any petition filed against it in any proceeding under any jurisdiction's bankruptcy, insolvency or reorganization law; (d) the adjudication of such Person as bankrupt or insolvent or the winding up or dissolution of such Person; (e) the making by such Person of a general assignment for the benefit of its creditors (assignments for a solvent financing excluded); (f) such Person fails or admits in writing its inability to pay its debts generally as they become due; (g) the appointment of a receiver or an administrator for all or a substantial portion of such Person's assets, which receiver or administrator, if appointed without the consent of such Person, is not discharged within 60 days after its appointment; or (h) the occurrence of any event analogous to any of the foregoing with respect to such Person occurring in any jurisdiction.
1

"Affiliate" of a specified Person means any corporation, partnership, sole proprietorship or other Person which directly or indirectly through one or more intermediaries controls, is controlled by or is under common control with the Person specified. The term "control" means the ownership, either direct or indirect, of twenty-five percent (25%) or more of the voting securities (or comparable equity interests) or other ownership interests of a Person, or the possession, either direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or any other means whatsoever.

"Agreement" means this Amended and Restated Corn Procurement and Handling Agreement, including all Appendices, as the same may be modified, supplemented or amended from time to time in accordance with the provisions hereof.

"Asset Management Agreement" means that certain Second Amended and Restated Asset Management Agreement by and among PEI, Pacific Ethanol Stockton, LLC, Pacific Ethanol Holding Co. LLC, Pacific Ethanol Madera LLC, Pacific Ethanol Columbia, LLC and Pacific Ethanol Magic Valley, LLC, dated as of June 30, 2011, as the same may be amended, supplemented or otherwise modified from time to time.

"Bilateral Transaction" means, with respect to corn purchased by PAP for delivery to [Plant Owner], a transaction entered into by PAP with one or more Third Parties consisting of one or more forward sales of corn.

"Business Day" means any day other than a Saturday, Sunday or a day on which commercial banks in Sacramento, California or New York, New York are required or authorized to be closed.

"Change of Control" has the meaning ascribed thereto in the Credit Agreement.

"Control" means, when used with respect to any Person, the power to direct or cause the direction of the management and policies of such Person, directly or indirectly, whether through the ownership or voting securities, by contract or otherwise.

"Corn Procurement and Handling Fee" means, as payment for the solicitation, proposal, negotiation and documentation of Bilateral Transactions and Grain Handling Services, \$0.045 per bushel of corn delivered to [Plant Owner] pursuant to Bilateral Transactions.

"Credit Agreement" means the Credit Agreement, dated as of June 25, 2010, by and among Pacific Ethanol Holding Co. LLC, Pacific Ethanol Madera LLC, Pacific Ethanol Columbia, LLC, Pacific Ethanol Magic Valley, LLC, and Pacific Ethanol Stockton, LLC, as Borrowers, Pacific Ethanol Holding Co. LLC, as Borrowers' Agent, WestLB AG, New York Branch, as the administrative agent and the collateral agent, and the lenders parties thereto from time to time, as the same may be amended, supplemented or otherwise modified from time to time.

"Dispute" means a dispute, controversy or claim.

"Effective Date" means June 30, 2011.

"Environmental Law" means any statute, law, regulation, ordinance, rule, judgment, order, decree, legally binding directive or requirement, or any similar form of decision of or determination by, or any interpretation or administration of any of the foregoing by a Governmental Authority, relating to the environment, health or safety as affected by the environment or any Hazardous Materials as now or hereinafter in effect.

"Expert" means an expert having sufficient technical expertise to address the matter subject to a Dispute.

"Extension Notice" has the meaning assigned to such term in Section 5.1.

"<u>Financing Documents</u>" means any and all loan agreements, credit agreements (including the Credit Agreement), reimbursement agreements, notes, indentures, bonds, security agreements, pledge agreements, mortgages, guarantee documents, intercreditor agreements, subscription agreements, equity contribution agreements and other agreements and instruments relating to the financing (or refinancing) of the operation, ownership and maintenance of the Facility.

"Financing Parties" means the banks, lenders, noteholders and/or other financial institutions (or an agent or trustee thereof) party to the Financing Documents.

"Force Majeure Event" has the meaning assigned to such term in Section 10.1.

"Good Industry Practice" means any of the practices, methods and acts engaged in or approved by a significant portion of the corn procurement and handling industry during the relevant time period, or any of the practices, methods and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision was made, could have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety and expedition. "Good Industry Practice" is not limited to a single set of optimum practices, methods or acts to the exclusion of others, but rather is intended to include acceptable practices, methods or acts generally accepted in the region.

"Governmental Authority" means any United States federal, state, municipal, local, territorial, or other governmental department, commission, board, bureau, agency, regulatory authority, instrumentality, judicial or administrative body.

"Grain Handling Services" means (a) receiving, unloading and conveying corn into the Storage Silos, (b) in the case of whole corn delivered to [Plant Owner], processing and hammering such whole corn and (c) conveying corn to the surge bin at the Facility.

"Hazardous Materials" means (a) any petroleum or petroleum products, radioactive materials, asbestos in any form that is or could become friable, urea formaldehyde foam insulation, and transformers or other equipment that contain dielectric fluid containing polychlorinated biphenyls; (b) any chemicals, materials or substances which are now or hereafter become defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous wastes," "toxic substances," "toxic pollutants," "pollution," "pollutants," "regulated substances," or works of similar import, under the Environmental Laws, including but not limited to the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended (42 U.S.C. § 9601 et seq.); the Hazardous Material Transportation Act, as amended (49 U.S.C. § 1801 et seq.); the Resource Conservation and Recovery Act, as amended (42 U.S.C. § 6901 et seq.); the Toxic Substances Control Act, as amended (42 U.S.C. § 7401 et seq.); the Clean Air Act, as amended (42 U.S.C. § 7401 et seq.); the Federal Water Pollution Control Act, as amended (33 U.S.C. § 1251 et seq.); or in the regulations promulgated pursuant to said laws; or (c) any other chemical, material, substance or waste declared to be hazardous, toxic, or polluting material by any Governmental Authority, exposure to which is now or hereafter prohibited, limited or regulated by any Governmental Authority.

"Late Payment Rate" means a rate of interest per annum equal to the Prime Rate plus two percent (2%).

"<u>Law</u>" means any law (including any Environmental Law), statute, act, legislation, bill, enactment, policy, treaty, international agreement, ordinance, judgment, injunction, award, decree, rule, regulation, interpretation, determination, requirement, writ or order of any Governmental Authority.

"Liabilities" has the meaning assigned to such term in Section 7.1.

"<u>Loading/Offloading Facilities</u>" means the rail spurs, barge and/or truck docks located at the Facility, and all grain and grain products loading and unloading equipment, including, but not limited to, all conveyors, lifts and elevators used in connection with movement of grain and grain products in and out of the Storage Silos.

"Monthly Date" means the last Business Day of each calendar month.

"NewCo" means New PE Holdco LLC, a Delaware limited liability company and the indirect owner on the date hereof of all the equity interests in [Plant Owner].

"PAP" has the meaning assigned to such term in the Preamble.

"PAP Indemnified Person" has the meaning assigned to such term in Section 7.2.

"PAP's Parties" means and includes, but is not limited to, employees, agents, contractors, subcontractors, invitees, and other Persons under PAP's Control or direction.

"Party" and "Parties" have the meanings assigned to such terms in the Preamble.

"PEI" means Pacific Ethanol, Inc., a Delaware corporation.

"<u>Permits</u>" means all permits, authorizations, registrations, consents, approvals, waivers, exceptions, variances, orders, judgments, written interpretations, decrees, licenses, exemptions, publications, filings, notices to and declarations of or with, or required by, any Governmental Authority, or required by any Law, and shall include all environmental and operating permits and licenses that are required for the full use, occupancy, zoning and operation of the Facility.

"Person" means any individual, partnership, corporation, association, business, trust, government or political subdivision thereof, governmental agency or other entity.

"[Plant Owner]" has the meaning assigned to such term in the Preamble.

"[Plant Owner] Indemnified Person" has the meaning assigned to such term in Section 7.1.

"[Plant Owner]'s Parties" means and includes, but is not limited to, employees, agents, contractors, subcontractors, invitees, and other Persons under [Plant Owner]'s Control or direction.

"Prime Rate" means the rate per annum listed as the "Prime Rate" in the "Money Rates" section of the Wall Street Journal from time to time.

"Prior Agreement" has the meaning assigned to such term in the Recitals.

"Services" has the meaning assigned to such term in Section 2.1.

"Storage Silos" means the grain storage silos located at the Facility.

"Third Party" means any Person (other than PEI or a subsidiary thereof) that enters into a Bilateral Transaction with PAP.

Interpretation. The following interpretations and rules of construction shall apply to this Agreement: (a) titles and headings are for convenience only and will not be deemed part of this Agreement for purposes of interpretation; (b) unless otherwise stated, references in this Agreement to "Sections," "Appendices" or "Articles" refer, respectively, to Sections, Appendices or Articles of this Agreement; (c) "including" means "including, but not limited to", and "include" or "includes" means "include, without limitation" or "includes, without limitation"; (d) "hereunder", "herein", "hereto" and "hereof", when used in this Agreement, refer to this Agreement as a whole and not to a particular Section or clause of this Agreement; (e) in the case of defined terms, the singular includes the plural and vice versa; (f) unless otherwise indicated, all accounting terms not specifically defined shall be construed in accordance with generally accepted accounting practices in the United States; (g) unless otherwise indicated, each reference to a particular Law is a reference to such Law as it may be amended, modified, extended, restated or supplemented from time to time, as well as to any successor Law thereto; (h) unless otherwise indicated, references to agreements shall be deemed to include all subsequent amendments, supplements and other modifications thereto; and (i) unless otherwise indicated, each reference to any Person shall include such Person's successors and permitted assigns.

ARTICLE II AGREEMENT

2.1 <u>Agreement.</u> [Plant Owner] has engaged PAP to (i) solicit, negotiate, enter into and administer, on behalf of [Plant Owner], corn supply arrangements necessary and sufficient to allow [Plant Owner] to procure corn necessary to operate the Facility and (ii) provide Grain Handling Services at the Facility as more specifically described herein (collectively, the "<u>Services</u>").

2.2 <u>Entire Agreement.</u> This Agreement contains the entire agreement between the Parties with respect to the subject matter hereof and supersedes all prior agreements, negotiations and understandings among the Parties with respect to such subject matter. Nothing in this Agreement shall be construed as creating a partnership or joint venture between the Parties.

ARTICLE III CORN PROCUREMENT

3.1 Appointment and Acceptance; Performance of Obligations.

- (a) Upon the terms and subject to the conditions of this Agreement and in furtherance of PAP's obligation to perform the Services, [Plant Owner] hereby appoints PAP as its agent, with effect from and after the Effective Date, having such authority as may be necessary for it to perform the Services including the authority to take actions and execute documents in the name of [Plant Owner], and PAP accepts such appointment and agrees to perform its duties under this Agreement. On and after the Effective Date, [Plant Owner] shall have the right by notice to PAP to revoke or rescind all or any part of the authority granted to PAP hereunder or otherwise reduce or restrict the scope of the Services.
- (b) PAP shall perform Services hereunder in all material respects in accordance with this Agreement, applicable Laws, applicable Permits and Good Industry Practice and with the intent to minimize the cost of corn to [Plant Owner].

3.2 <u>Limitations on Authority.</u>

- (a) PAP shall solicit, negotiate and administer Bilateral Transactions on general terms and conditions that have been established pursuant to the operating protocol established pursuant to Section 3.5(b) and subject to the limitations contained in this Section 3.2. PAP shall communicate to prospective corn suppliers that all proposals for Bilateral Transactions are subject to the conditions set forth in clause (c) below and any other conditions agreed by the Parties in the operating protocol.
- (b) PAP, as agent for [Plant Owner] or otherwise on behalf of [Plant Owner], shall not and shall cause each of its agents, employees and representatives not to enter into or execute any Bilateral Transaction or any other transaction in respect of the procurement of corn for the Facility having a term greater then one hundred eighty (180) days or otherwise involving forward purchases or sales of corn for more then one hundred eighty (180) days, in each case without the prior written consent of [Plant Owner], which consent may be withheld by [Plant Owner] in its discretion.
- (c) PAP, as agent for [Plant Owner] or otherwise on behalf of [Plant Owner], shall not and shall cause each of its agents, employees and representatives not to: (i) enter into, execute, suspend or terminate (or accept any termination of), (ii) amend, modify or supplement, (iii) give or accept waivers with respect to or (iv) take any action or omit to take any action with respect to, any Bilateral Transaction between [Plant Owner] and a Third Party, that would result in a violation of or a default under any Financing Document, unless the consent of the necessary Financing Party or Financing Parties, as the case may be, has theretofore been obtained by [Plant Owner].

- (d) Notwithstanding anything to the contrary herein, PAP shall not (i) solicit or propose any Bilateral Transaction with any Third Party that is the subject of an Act of Insolvency, (ii) solicit or propose any Bilateral Transaction which provides for the provision of corn in excess of the amount of corn required by [Plant Owner] (after giving effect to [Plant Owner]'s existing contractual obligations and the scheduling provisions set forth in Section 3.5(b) below), or (iii) enter into Bilateral Transactions in its name.
- (e) In the event of a breach or default by a Third Party under any Bilateral Transaction, PAP shall promptly notify [Plant Owner] of any such breach and default and provide [Plant Owner] from time to time with reasonably detailed information in respect of the same (including copies of all written communications in respect thereof). In the event of any such breach or default, PAP shall use commercially reasonable efforts to procure replacement corn.
- (f) [Plant Owner] and PAP expressly acknowledge and agree that: (i) this Agreement does not convey ownership or control over all or any part of the Facility from [Plant Owner] to PAP; and (ii) [Plant Owner] retains ultimate decision-making authority relating to the operation of the Facility, which authority may be exercised by [Plant Owner] in its discretion, *provided* that, except as otherwise specifically instructed by written direction from [Plant Owner] to PAP, PAP shall exercise its authority and fulfill its responsibility to perform the Services.
- 3.3 <u>Obligations of [Plant Owner]</u>. [Plant Owner] shall provide PAP with all information reasonably requested by PAP, and [Plant Owner] shall assist PAP as reasonably requested in connection with the Services provided hereunder. At the request of [Plant Owner], PAP will cause PEI to execute and deliver and maintain in full force and effect a guaranty in the form of Exhibit B hereto.
- 3.4 <u>Transaction Reports</u>. Within 30 days after each Monthly Date occurring after the Effective Date, PAP shall deliver to [Plant Owner] a written summary of the Bilateral Transactions which were entered into or performed, in whole or in part, during the month ending on such Monthly Date.

3.5 <u>Delivery Point; Operating Protocol; Nominations; Measurement.</u>

- (a) PAP shall arrange for corn purchased pursuant to Bilateral Transactions to be delivered to [Plant Owner] at the truck, barge and/or rail receiving basin at the Facility specified in the operating protocol described in <u>Section 3.5(b)</u>.
- (b) PAP and [Plant Owner] shall use the agreed operating protocol with respect to the mechanics, timing and process for (i) determining how much corn is required to be purchased on any particular day from Third Parties, (ii) guidelines for the transmittal of solicitations or proposal of Bilateral Transactions and [Plant Owner]'s response to such solicitation or proposal, (iii) [Plant Owner] to communicate to PAP its corn requirements on a monthly, weekly and daily basis, (iv) determining the quantity of corn to be stored in the Storage Silos, and (v) administering the corn purchases contemplated by this Agreement. By mutual agreement, such operating protocol shall be updated from time to time thereafter. A copy of such protocol is attached hereto as Exhibit C.

- (c) All corn supplied under this Agreement shall conform to the minimum specifications set forth on Exhibit A hereto. Notwithstanding anything to the contrary herein, if PAP supplies the Facility with corn that fails to meet the minimum specifications set forth on Exhibit A and such failure was not caused by PAP, such failure will not be a breach of this Agreement *provided* that (x) PAP shall pursue any and all damages, price reductions and price refunds from the applicable Third Party with respect to any such failure and pass on such damages, price reductions and price refunds to [Plant Owner], (y) in the event of any such failure, PAP shall use commercially reasonable efforts to procure replacement corn and (z) in any event, PAP will not enter into any Bilateral Transactions which provide for a baseline corn quality that is materially worse than the minimum specifications set forth on Exhibit A.
- (d) On or before the date that is fifteen days prior to the end of a calendar month, [Plant Owner] shall provide PAP with a forecast of its projected monthly corn requirements for the following month.
- (e) PAP agrees to use its commercially reasonable efforts to procure corn according to "first official grades" performed at time of loading, and PAP shall use its commercially reasonable efforts to ensure that the corn conforms to the standards of #2 Yellow Corn, as determined by the National Grain & Feed Association. PAP shall have the right to collect samples of each shipment of corn delivered to [Plant Owner] under Bilateral Transactions at the point of unloading. [Plant Owner] shall have the right, upon reasonable notice and at reasonable times and at its expense, to request copies of the "first official grades" documentation that accompanies every shipment to confirm that the corn delivered to it under Bilateral Transactions meets the requirements of such Bilateral Transactions and the requirements of Section 3.5(b).
 - 3.6 <u>Benchmarking.</u> PAP shall furnish on a monthly basis a report benchmarking its performance in accordance with Exhibit D.

ARTICLE IV COMPENSATION AND PAYMENT

As compensation to PAP for the performance of its services hereunder, [Plant Owner] shall pay PAP, in the manner and at the times specified in this <u>Article IV</u>, the Corn Procurement and Handling Fee, all as further described herein.

4.1 <u>Fees and Payments</u>.

(a) Following the last Business Day of each calendar month (each calendar month being referred to herein as a "<u>Payment Period</u>") and within 15 days following receipt of an invoice from PAP in respect of such Payment Period, [Plant Owner] shall pay to PAP an amount equal to the Corn Procurement and Handling Fee for such Payment Period; *provided*, *however*, that [Plant Owner] shall not be required to make payments to PAP in respect of any Bilateral Transactions for which [Plant Owner] has not received corn from the corresponding Third Party. In connection with each such invoice, PAP shall deliver to [Plant Owner] a statement detailing its calculations of the applicable Corn Procurement and Handling Fee.

(b) If PAP defaults in its obligation to provide Services in accordance with the terms of this Agreement, then [Plant Owner] shall be entitled to set-off and deduct from current and/or future payments owed to PAP by [Plant Owner] an amount equal to the amount of any damage payments owed by PAP to [Plant Owner] as a result of PAP's failure to perform hereunder.

4.2 Overdue Payments; Indemnity Payments.

- (a) If any Party shall fail to make any payment when due hereunder, such overdue payment shall accrue interest at the Late Payment Rate from the date originally due until the date paid.
- (b) Any indemnification payments received by PAP from a Third Party in respect of a Bilateral Transaction shall be paid to [Plant Owner] on or before the next payment date as determined in accordance with <u>Section 4.1</u>.
- 4.3 <u>Billing Dispute</u>. If [Plant Owner] or PAP, in good faith, disputes the amount of any payment received by it or to be paid by it or set-off pursuant to <u>Section 4.1</u> above, the disputing Party shall immediately notify the other Party of the basis for the dispute. The Parties will then meet and use their best efforts to resolve any such dispute. If any amount is ultimately determined to be due to or permitted to be set-off by [Plant Owner] or PAP (as the case may be), to the extent not previously paid or set-off, (a) PAP (or [Plant Owner], as the case may be) shall pay such amount to [Plant Owner] (or PAP, as the case may be) within five Business Days of such determination or (b) PAP (or [Plant Owner], as the case may be) may then set-off such amount (as the case may be).
- Audit. Notwithstanding the payment of any amount pursuant to this Article IV, [Plant Owner] shall remain entitled (upon reasonable prior notice, at reasonable times and at PAP's corporate offices) and the administrative agent under the Credit Agreement (and its consultants, as directed by the administrative agent) shall be entitled (upon reasonable prior notice, not more than once per calendar quarter and at PAP's corporate offices) to conduct a subsequent audit and review of (a) all Bilateral Transactions and related records to verify the amount of gross payments and damage payments and (b) the determination and calculation of the Corn Procurement and Handling Fee, for a period of two years from and after the end of the applicable Payment Period. If, pursuant to such audit and review, it is determined that any amount previously paid by [Plant Owner] to PAP was in excess of the amounts which should have been paid to PAP, [Plant Owner] shall advise PAP indicating such amount and the reason the amount should not have been paid by [Plant Owner] and, subject to the next two sentences, PAP shall pay such amount to [Plant Owner] within five Business Days of such request along with interest accrued at the Late Payment Rate from the date originally paid until the date repaid to [Plant Owner]. If the Parties do not agree with respect to any item so noted, the Parties will then meet and use their best efforts to resolve the dispute. If the Parties are not able to resolve issues raised by such an audit and review, any disputed items will be resolved in accordance with the provisions of Article XI.

ARTICLE V TERM

- 5.1 Term. This Agreement shall be effective as of the Effective Date and, unless earlier terminated in accordance with its terms, shall continue in effect until and including June 30, 2012; *provided*, that either [Plant Owner] or PAP may extend this Agreement for additional one year periods, in each case by written notice to the other (an "Extension Notice") delivered not less than ninety (90) days prior to the end of the original or renewal term, provided further that this Agreement shall nonetheless terminate if the recipient of any such Extension Notice rejects such Extension Notice not more than fifteen (15) days after receipt of the Extension Notice.
- 5.2 [Plant Owner] Defaults and PAP Remedies. Upon the occurrence of any of the following events, PAP may exercise such rights and remedies as may be available to it at law or in equity, including the right to terminate this Agreement, by written notice to [Plant Owner], provided, that no such notice shall be required for a termination pursuant to clause (c) of this Section 5.2:
- (a) the failure by [Plant Owner] to make any payment, deposit or transfer required hereunder within thirty (30) days after the date such payment, deposit or transfer is due, and such failure continues for fifteen (15) days after receipt of written notice from PAP of such failure:
- (b) the failure of any statement, representation or warranty made by [Plant Owner] in this Agreement to have been correct in any material respect when made if such failure could reasonably be expected to have a material adverse effect on [Plant Owner]'s ability to perform its obligations under this Agreement;
 - (c) the occurrence of an Act of Insolvency with respect to [Plant Owner]; or
- (d) the failure of [Plant Owner] to perform any of its material obligations under this Agreement and such failure continues for 30 days after receipt of written notice from PAP of such failure; *provided*, that such 30-day period shall be extended for up to an aggregate of 90 days so long as [Plant Owner] is diligently attempting to cure such failure.
- 5.3 <u>PAP Defaults and [Plant Owner] Remedies.</u> Upon the occurrence of any of the following events, [Plant Owner] may exercise such rights and remedies as may be available to it at law or in equity, including the right to terminate this Agreement, by written notice to PAP, provided, that no such notice shall be required for a termination pursuant to clause (c) of this Section 5.3:
- (a) the failure by PAP to make any payment, deposit or transfer required hereunder within thirty (30) days after the date such payment, deposit or transfer is due, and such failure continues for fifteen (15) days after receipt of written notice from [Plant Owner] of such failure;

- (b) the failure of any statement, representation or warranty made by PAP in this Agreement to have been correct in any material respect when made if such failure could reasonably be expected to have a material adverse effect on PAP's ability to perform its obligations under this Agreement;
 - (c) the occurrence of an Act of Insolvency with respect to PAP; or
- (d) the failure of PAP to perform any of its material obligations under this Agreement and such failure continues for 30 days after receipt of written notice from [Plant Owner] of such failure; *provided*, that such 30-day period shall be extended for up to an aggregate of 90 days so long as PAP is diligently attempting to cure such failure.
- 5.4 <u>Change of Control.</u> This Agreement shall terminate 45 days after the occurrence of (i) any Change of Control with respect to [Plant Owner] or any transfer, assignment, sale or other disposition of more than a majority of the membership interests in PAP to any Person that is not an Affiliate of PEI or (ii) any transfer, assignment, sale or other disposition of all or substantially all of the assets comprising the Facility, in each case unless the Parties mutually agree to the contrary.
- 5.5 Replacement. Notwithstanding any other provision of this Agreement, during the continuance of any default by PAP that would allow [Plant Owner] to terminate this Agreement pursuant to Section 5.3 or during the 30-day cure period provided in Section 5.3(b) (notwithstanding such cure period) if PAP fails to procure corn necessary to operate the Facility or during the continuance of any Force Majeure Event (including the effects thereof) that renders PAP unable to perform its obligations under this Agreement, then [Plant Owner] shall have the right to engage any other Person to provide the Services and PAP shall not be entitled to any compensation (including any Corn Procurement and Handling Fee) with respect to any replacement services provided by such other Person.

ARTICLE VI INSURANCE

- 6.1 <u>PAP Insurance</u>. Without limiting any of the other obligations or liabilities of PAP under this Agreement, PAP shall at all times carry and maintain or cause to be carried and maintained, the minimum insurance coverage set forth in this Section:
- (a) PAP shall maintain or cause to be maintained (i) Workers' Compensation insurance in compliance with the workers' compensation laws of the states in which PAP provides Services as extended by the Broad Form All States Endorsements, the United States Longshoreman's and Harbor Workers' Coverage Endorsements on an if-any-exposure basis and the Voluntary Compensation Coverage Endorsement, and (ii) Employer's Liability (including Occupational Disease) coverage with limits of not less than \$1,000,000, which shall cover all of PAP's employees engaged in providing services hereunder.

- (b) PAP shall maintain or cause to be maintained automobile liability insurance for owned (if any), non-owned and hired vehicles with combined single limits for bodily injury/property damage not less than \$1,000,000 per occurrence and containing appropriate no-fault insurance provisions wherever applicable.
- (c) PAP will maintain or cause to be maintained commercial general liability insurance with a limit for bodily injury/property damage of not less than \$1,000,000 per occurrence and \$2,000,000 in the annual aggregate. Such coverage shall include premises/operations, explosion, collapse and underground property damage, broad form contractual, independent contractors, products/completed operations (including operator errors and omissions), broad form property damage, personal injury and incidental professional liability (if not covered under product/completed operations and if commercially available).
- (d) PAP shall maintain or cause to be maintained umbrella liability insurance providing coverage limits in excess of those set forth in Section (a), (b) and (c) above. The limits of this umbrella coverage shall not be less than \$10,000,000 per occurrence and in the annual aggregate.
- (e) PAP shall maintain or cause to be maintained pollution legal liability for sudden and accidental pollution for physical damage and bodily injury to third parties in an amount of \$3,000,000 per occurrence and in the annual aggregate.

The terms and conditions of all insurance policies (including the amount, scope of coverage, deductibles, and self-insured retentions) shall be acceptable in all respects as of the Effective Date. All insurance carried pursuant to this Section shall conform to the relevant provisions of this Agreement and be with insurance companies which are rated "A-, X" or better by Best's Insurance Guide and Key Ratings, or other insurance companies of recognized responsibility satisfactory to [Plant Owner]. [Plant Owner] shall be furnished with satisfactory evidence that the foregoing insurance is in effect and [Plant Owner] shall be notified 30 calendar days prior to the cancellation or material change of any such coverage. Coverage for the insurance under Section (c) and (d) above shall be written on a claims made basis provided that if the policy is not renewed, PAP shall obtain for the benefit of [Plant Owner] an extended reporting period coverage or "tail" of at least three years past the final day of coverage of such policy. PAP shall provide [Plant Owner] with evidence that such extended reporting period coverage or "tail" has been obtained. PAP agrees to ensure that the insurance policies outlined in this Section require the insurer to waive subrogation against [Plant Owner], the Financing Parties and their respective Affiliates together with their respective officers, directors, Affiliates and employees and all such Persons shall be an additional insured as their interests may appear with respect to all policies procured by PAP.

6.2 <u>PAP Insurance Premiums and Deductibles.</u> All premiums for insurance coverage procured by PAP pursuant to <u>Section 6.1</u> shall be reimbursed by [Plant Owner] upon demand. PAP shall be liable for the payment of all deductibles on insurance policies obtained pursuant to <u>Section 6.1</u>, which amounts shall not be reimbursed by [Plant Owner], provided that, to the extent that a claim under a policy described in <u>Section 6.1</u> is attributable to [Plant Owner]'s (including its employees' or agents') gross negligence or willful misconduct, [Plant Owner] shall be liable for the entire amount of such deductible. In no event shall any premiums, deductibles or any losses in excess of insurance coverage be a reimbursable cost hereunder.

ARTICLE VII INDEMNIFICATION

- 7.1 [Plant Owner]'s Indemnity. [Plant Owner] shall defend, indemnify and hold harmless PAP and its Affiliates (and each officer, director, employee, shareholder, partner, member or agent of PAP and its Affiliates)(each, a "[Plant Owner] Indemnified Person") from and against any and all third party claims, actions, damages, expenses (including reasonable and documented attorneys' fees and expenses), losses, settlements or liabilities (collectively, "Liabilities") incurred or asserted against any [Plant Owner] Indemnified Person (a) as a result of any failure on the part of [Plant Owner] to perform [Plant Owner]'s obligations under this Agreement, or (b) arising out of or in any way connected with the grossly negligent acts or omissions of [Plant Owner] or its Affiliates.
- 7.2 PAP's Indemnity. PAP shall defend, indemnify and hold harmless [Plant Owner] and its Affiliates (and each officer, director, employee, shareholder, partner, member or agent of [Plant Owner] and their Affiliates)(each, a "PAP Indemnified Person") from and against any and all third party Liabilities incurred or asserted against any PAP Indemnified Person (a) as a result of any failure on the part of PAP to perform its obligations under this Agreement, or (b) arising out of or in any way connected with the grossly negligent acts or omissions of PAP or its Affiliates.

ARTICLE VIII LIABILITIES OF THE PARTIES

- 8.1 <u>No Consequential or Punitive Damages.</u> In no event shall either Party be liable to any other Party by way of indemnity or by reason of any breach of contract or of statutory duty or by reason of tort (including negligence or strict liability) or otherwise for any loss of profits, loss of revenue, loss of use, loss of production, loss of contracts or for any incidental, indirect, special or consequential or punitive damages of any other kind or nature whatsoever that may be suffered by such other Party, including any losses for which such other Party has insurance to the extent proceeds of insurance have been recovered for such losses.
- 8.2 <u>Breach</u>. Notwithstanding anything to the contrary herein, any failure, breach or default by [Plant Owner] under this Agreement that is caused by or is a result of a failure, breach or default by PEI under the Asset Management Agreement will not be a breach of this Agreement.

ARTICLE IX CONFIDENTIALITY

Each Party and its Affiliates shall treat as confidential the data and information in their possession regarding the Facility, the other Parties or any Affiliate of any other Party, unless: (a) the applicable other Party agrees in writing to the release of such data or information; (b) such data or information becomes publicly available other than through the wrongful actions of the disclosing Party or the disclosing Party's Affiliate; (c) such data or information was in the possession of the receiving Party or the receiving Party's Affiliate prior to receipt thereof from the disclosing Party with no corresponding confidentiality obligation; or (d) such data or information is required by Law to be disclosed. Notwithstanding the generality of the foregoing, any Party may disclose data and information to (i) the officers, directors, managers, partners, members, employees and Affiliates of such Party, but only to the extent such Persons have a need to know such information for purposes of permitting such Party to perform its obligations hereunder, (ii) any successors in interest and permitted assigns of such Party, (iii) any actual or potential Financing Parties or actual or potential lenders to PEI or any subsidiary thereof, and (iv) any potential equity investors in such Party or its Affiliates or acquirer or potential acquirer of the Project or all or any of the equity interests in Newco or any subsidiary thereof; *provided*, that any Person who receives confidential data and information pursuant to an exception contained in clauses (ii) through (iv) of this Article agrees to confidentiality provisions at least as restrictive as the provisions set forth herein.

ARTICLE X FORCE MAJEURE

- 10.1 Events Constituting Force Majeure. As used herein, "Force Majeure Event" means any cause(s) which render(s) a Party wholly or partly unable to perform its obligations under this Agreement (other than obligations to make payments when due), and which are neither reasonably within the control of such Party nor the result of the fault or negligence of such Party, and which occur despite all reasonable attempts to avoid, mitigate or remedy, and shall include acts of God, war, riots, civil insurrections, cyclones, hurricanes, floods, fires, explosions, earthquakes, lightning, storms, chemical contamination, epidemics or plagues, acts or campaigns of terrorism or sabotage, blockades, embargoes, accidents or interruptions to transportation not caused by PAP, trade restrictions, acts of any Governmental Authority after the date of this Agreement, strikes and other labor difficulties (other than with respect to its own employees) not caused by PAP, mechanical breakdowns, and other events or circumstances beyond the reasonable control of such Party.
- Effect. A Party claiming relief as a result of a Force Majeure Event shall give the other Party written notice within five Business Days of becoming aware of the occurrence of the Force Majeure Event, or as soon thereafter as practicable, describing the particulars of the Force Majeure Event, and will use reasonable efforts to remedy its inability to perform as soon as possible. If the Force Majeure Event (including the effects thereof) continues for fifteen consecutive days, the affected Party shall report to the other Party the status of its efforts to resume performance and the estimated date thereof. If the Force Majeure Event (including the effects thereof) continues for 180 consecutive days, either Party may terminate this Agreement for convenience. If the affected Party was not able to resume performance prior to or at the time of the report to the other Party of the onset of the Force Majeure Event, then it will report in writing to the other Party when it is again able to perform. If a Party fails to give timely notice, the excuse for its non-performance shall not begin until notice is given.
- 10.3 <u>Limitations</u>. Any obligation(s) of a Party (other than an obligation to make payments when due) may be temporarily suspended during any period such Party is unable to perform such obligation(s) by reason of the occurrence of a Force Majeure Event, but only to the extent of such inability to perform, *provided*, that:
- (a) the suspension of performance is of no greater scope and of no longer duration than is reasonably required by the Force Majeure Event; and
 - (b) the Party claiming the occurrence of the Force Majeure Event bears the burden of proof.

ARTICLE XI DISPUTE RESOLUTION

- Attempts to Settle. In the event that a Dispute between the Parties arises under, out of or in relation to, this Agreement, the Parties shall attempt in good faith to settle such Dispute by mutual discussions within fifteen Business Days after the date that an aggrieved Party gives written notice of the Dispute to the other Party. In the event that a Dispute is not resolved by discussion in accordance with the preceding sentence within the time period set forth therein, the Parties shall refer the Dispute to their respective senior officers for further consideration and attempted resolution within fifteen Business Days after the Dispute has been referred to such individuals (or such longer period as the Parties may agree).
- Resolution by Expert. If the Parties shall have failed to resolve the Dispute within fifteen Business Days after the date that the Parties referred the Dispute to their senior officers, then, provided the Parties shall so agree, the Dispute may be submitted for resolution by an Expert, such Expert to be appointed by the mutual agreement of the Parties. Proceedings before an Expert shall be held in Sacramento, California (or any other location agreed to by the Parties). The Expert shall apply to such proceedings the substantive law of the State of New York in effect at the time of such proceedings. The decision of the Expert shall be final and binding upon the Parties. In the event that (a) the Parties cannot agree on the appointment of an Expert within ten Business Days after the date that the Parties agreed to submit the Dispute for resolution by the Expert or (b) the Expert fails to resolve such Dispute within 60 days after the Parties have submitted such Dispute to the Expert, then any Party may file a demand for arbitration in writing in accordance with Section 11.3.
- Arbitration. Any Dispute that has not been resolved following the procedures set forth in Section 11.1 or 11.2 shall be settled by binding arbitration in Sacramento, California (or any other location agreed to by the Parties) before a panel of three arbitrators. Such arbitration shall be conducted in accordance with the Commercial Arbitration Rules of the American Arbitration Association as in effect on the date of execution of this Agreement. Such arbitration shall be governed by the laws of the State of New York. If arbitration proceedings have been initiated pursuant to this Section 11.3 and raise issues of fact or law which, in whole or in part, are substantially the same as issues of fact or law already pending in arbitration proceedings involving the applicable Parties, such issues shall be consolidated with the issues in the ongoing proceedings. THE PARTIES HEREBY AGREE THAT THE PROCEDURES SET FORTH IN THIS SECTION 11.3 SHALL BE THE EXCLUSIVE DISPUTE RESOLUTION PROCEDURES APPLICABLE TO ANY DISPUTE, CONTROVERSY OR CLAIM UNDER THIS AGREEMENT AND, EXCEPT AS SET FORTH IN SECTION 11.5, THE PARTIES HEREBY WAIVE ALL RIGHTS TO A COURT TRIAL OR TRIAL BY JURY WITH RESPECT TO ANY DISPUTE, CONTROVERSY OR CLAIM UNDER THIS AGREEMENT.
- 11.4 <u>Consequential and Punitive Damages.</u> Awards of Experts and arbitral panels shall be subject to the provisions of <u>Article VIII.</u>

- 11.5 <u>Finality and Enforcement of Decision</u>. Any decision or award of an Expert or a majority of an arbitral panel, as applicable, shall be final and binding upon the Parties. Each of the Parties agrees that the arbitral award may be enforced against it or its assets wherever they may be found and that a judgment upon the arbitral award may be entered in any court having jurisdiction thereof.
- 11.6 <u>Costs</u>. The costs of submitting a Dispute to an Expert shall be shared equally among the Parties, unless the arbitral panel or the Expert determines otherwise. The costs of arbitration shall be paid in accordance with the decision of the arbitral panel pursuant to the Commercial Arbitration Rules of the American Arbitration Association as in effect on the date of execution of this Agreement.
- 11.7 <u>Continuing Performance Obligations.</u> While a Dispute is pending, each Party shall continue to perform its obligations under this Agreement, unless such Party is otherwise entitled to suspend its performance hereunder or terminate this Agreement in accordance with the terms hereof.

ARTICLE XII MISCELLANEOUS PROVISIONS

- Assignment. No Party shall assign this Agreement or any of its rights or obligations hereunder without first obtaining the prior written consent of (a) in the case of [Plant Owner], PAP, or (b) in the case of PAP, [Plant Owner], provided, that either Party shall be entitled to assign its rights hereunder (as collateral security or otherwise) for financing purposes (including a collateral assignment by [Plant Owner] to any Financing Parties) without the consent of the other Party.
- 12.2 <u>Cooperation in Financing.</u> PAP shall use its reasonable efforts to execute, acknowledge and deliver any and all further documents and instruments, and to take any other actions, which may be necessary to satisfy the reasonable requests of any Financing Party or prospective Financing Party in connection with the financing of the Facility.
- 12.3 <u>Not for Benefit of Third Parties</u>. Except as otherwise expressly provided in this Agreement, each and every provision hereof is for the exclusive benefit of the Parties hereto and is not for the benefit of any third party.
- 12.4 <u>Amendments</u>. No Party hereto shall be bound by any termination, amendment, supplement, waiver or modification of any term hereof unless such Party shall have consented thereto in writing.
- 12.5 <u>Survival</u>. Cancellation, expiration or earlier termination of this Agreement shall not relieve the Parties of obligations that by their nature should survive such cancellation, expiration or termination, including remedies, limitations on liability, promises of payment, indemnity and confidentiality. Without limiting the generality of the foregoing, the following provisions of this Agreement shall survive: Articles IV, VII, VIII, IX, XI and XII.

- 12.6 <u>No Waiver</u>. No delay or failure on the part of any Party in exercising any rights hereunder, and no partial or single exercise thereof, shall constitute a waiver of such rights or of any other rights hereunder.
- Notices. All notices or other communications which are required or permitted hereunder shall be in writing and shall be deemed sufficiently given (a) upon delivery, if delivered personally, (b) the day the notice is received, if it is delivered by overnight courier or certified or registered mail, postage prepaid, or (c) upon the effective receipt of electronic transmission, facsimile, telex or telegram (with effective receipt being deemed to occur upon the sender's receipt of confirmation of successful transmission of such notice or communication), to the addresses set forth below or such other address as the addressee may have specified in a notice duly given to sender as provided herein:

If to PAP:

Pacific Ag. Products, LLC c/o Pacific Ethanol, Inc. 400 Capitol Mall Suite 2060 Sacramento, California 95814 Attention Neil Koehler Telephone: (916) 403-2123 Facsimile: (916) 446-3936

With a copy to:

Pacific Ethanol, Inc.
400 Capitol Mall, Suite 2060
Sacramento, California 95814
Attention General Counsel
Telephone: (916) 403-2130
Facsimile: (916) 403-2785

If to [Plant Owner]:

[Plant Owner]
c/o JT Miller Group LLC
777 Campus Commons Road # 200
Sacramento, California 95825
Attention John Miller
Telephone: (916) 565-7422
Facsimile: (916) 565-7423

12.8 Representations and Warranties.

12.8.1 PAP's Representations and Warranties. PAP represents and warrants to [Plant Owner], as of the date hereof, as follows: <u>Due Formation</u>. PAP (a) is a limited liability company duly formed and validly existing under the laws of the State of California, (b) has the requisite power and authority to own its properties and carry on its business as now being conducted and currently proposed to be conducted and to execute, deliver and perform its obligations under this Agreement, and (c) is qualified to do business in every jurisdiction in which failure so to qualify could be reasonably be expected to have a material adverse effect on PAP's ability to perform its obligations hereunder. 12.8.1.2 Authorization; Enforceability. PAP has taken all action necessary to authorize it to execute, deliver and perform its obligations under this Agreement. This Agreement constitutes a legal, valid and binding obligation of PAP enforceable in accordance with its terms, subject to bankruptcy, reorganization, moratorium or other similar laws affecting the enforcement of the rights of creditors generally and subject to general principles of equity. No Conflict. The execution, delivery and performance by PAP of this Agreement does not and will not (a) violate any Law applicable to PAP, (b) result in any breach of PAP's constituent documents or (c) conflict with, violate or result in a breach of or constitute a default under any agreement or instrument to which PAP or any of its properties or assets is bound or result in the imposition or creation of any lien or security interest in or with respect to any of PAP's property or assets, other than in each case any such violations, conflicts, breaches or impositions which could not be reasonably be expected to have a material adverse effect on PAP's ability to perform its obligations hereunder. 12.8.1.4 No Authorization. No authorization or approval or other action by, and no notice to or filing with, any Governmental Authority (other than those which have been obtained) is required for the due execution, delivery and performance by PAP of this Agreement, other than any such authorizations, approvals or actions the failure of which to obtain could not be reasonably be expected to have a material adverse effect on PAP's ability to perform its obligations hereunder. Litigation. PAP is not a party to any legal, administrative, arbitration or other proceeding, and, 12.8.1.5 to PAP's knowledge, no such proceeding is threatened, which could be reasonably be expected to have a material adverse effect on PAP's ability to perform its obligations hereunder. 12.8.2 [Plant Owner]'s Representations and Warranties. [Plant Owner] represents and warrants to PAP, as of the date hereof, as follows: 12.8.2.1 Due Formation. [Plant Owner] (a) is a limited liability company duly formed and validly existing under the laws of the State of Delaware, (b) has the requisite power and authority to own its properties and carry on its business as now being conducted and currently proposed to be conducted and to execute, deliver and perform its obligations under this Agreement, and (c) is

have a material adverse effect on [Plant Owner]'s ability to perform its obligations hereunder.

qualified to do business in the State of California and in every other jurisdiction in which failure so to qualify could be reasonably be expected to

- 12.8.2.2 <u>Authorization; Enforceability.</u> [Plant Owner] has taken all action necessary to authorize it to execute, deliver and perform its obligations under this Agreement. This Agreement constitutes a legal, valid and binding obligation of [Plant Owner] enforceable in accordance with its terms, subject to bankruptcy, reorganization, moratorium or other similar laws affecting the enforcement of the rights of creditors generally and subject to general principles of equity.
- 12.8.2.3 No Conflict. The execution, delivery and performance by [Plant Owner] of this Agreement does not and will not (a) violate any Law applicable to [Plant Owner], (b) result in any breach of [Plant Owner]'s constituent documents or (c) conflict with, violate or result in a breach of or constitute a default under any agreement or instrument to which [Plant Owner] or any of its properties or assets is bound or result in the imposition or creation of any lien or security interest in or with respect to any of [Plant Owner]'s property or assets, other than in each case any such violations, conflicts, breaches or impositions which could not be reasonably be expected to have a material adverse effect on [Plant Owner]'s ability to perform its obligations hereunder.
- 12.8.2.4 No authorization. No authorization or approval or other action by, and no notice to or filing with, any Governmental Authority (other than those which have been obtained) is required for the due execution, delivery and performance by [Plant Owner] of this Agreement, other than any such authorizations, approvals or actions the failure of which to obtain could not be reasonably be expected to have a material adverse effect on [Plant Owner]'s ability to perform its obligations hereunder.
- 12.8.2.5 <u>Litigation</u>. [Plant Owner] is not a party to any legal, administrative, arbitration or other proceeding, and, to [Plant Owner]'s knowledge, no such proceeding is threatened, which could be reasonably be expected to have a material adverse effect on [Plant Owner]'s ability to perform its obligations hereunder.
- 12.9 <u>Counterparts and Execution</u>. This Agreement may be executed in any number of counterparts and each such counterpart shall be deemed to be an original instrument, but all such counterparts together shall constitute one and the same agreement.
- 12.10 <u>Governing Law.</u> This Agreement shall be governed by and construed in accordance with the laws of the State of New York without regard to the principles of conflicts of laws thereof.
- 12.11 <u>Severability</u>. In the event any one or more of the provisions contained in this Agreement should be held invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein shall not in any way be affected or impaired thereby. The Parties shall endeavor in good-faith negotiations to replace the invalid, illegal or unenforceable provisions with valid provisions, the economic and practical effect of which comes as close as possible to that of the invalid, illegal or unenforceable provisions.

12.12	Successors and Assigns.	This Agreement shall be binding upon and shall inure to the benefit of the Parties hereto and their
respective successor	s and permitted assigns.	
	reference, and in no way	Titles or captions of sections contained in this Agreement are inserted only as a matter of define, limit, extend, describe or otherwise affect the scope of meaning of this Agreement or the attached hereto shall be considered a part hereof as though fully set forth herein.
	[THE REMAINDE	ER OF THIS PAGE HAS BEEN LEFT BLANK INTENTIONALLY]

IN WITNESS WHEREOF, this Amended and Restated Corn Properties hereto as of the date first written above.	rocurement and Handling Agreement has been duly executed by the
PAC	CIFIC ETHANOL []
By:	Name: Title:
PA	CIFIC AG. PRODUCTS, LLC
Ву:	/s/ Neil M. Koehler Name: Neil M. Koehler Title: Chief Executive Officer
[Signature Page to Corn Procurement	t Agreement — [Plant Owner]]

Exhibit A

Corn Specifications

A minimum feedstock specification of 54/lb bushel test weight corn containing a maximum of 15.0% moisture (by weight) and a maximum of 4% broken corn and foreign material (by weight).

National Grain & Feed Association rules for #2 Yellow Corn to apply.					

Exhibit B

Form of PEI Guaranty

[Please see attached.]

Form of Parent Guaranty (Amended and Restated Corn Procurement and Handling Agreement)

[DATE]
Pacific Ethanol [] c/o JT Miller Group LLC 777 Campus Commons Road # 200 Sacramento, California 95825 Attention: John Miller
Re: Guaranty of Affiliate Project Party Obligations under Amended and Restated Corn Procurement and Handling Agreement
Ladies and Gentlemen:
1. The Guaranty. For value received, Pacific Ethanol, Inc., a corporation organized under the laws of Delaware (the "Guarantor"), hereby unconditionally and absolutely guarantees the prompt and complete payment and performance when due, whether by acceleration or otherwise, of all obligations and liabilities, whether now in existence or hereafter arising (the "Obligations"), of Pacific Ag. Products, LLC, a California limited liability company (the "Affiliate Project Party"), to [], LLC, a Delaware limited liability company (the "Project Owner"), under that certain Amended and Restated Corn Procurement and Handling Agreement, dated as of _, 20_ (the "Agreement"); provided that the Guarantor's maximum aggregate liability under this Guaranty shall not exceed the maximum amount which the Affiliate Project Party owes the Project Owner under the Agreement.
Following any demand by the Project Owner for payment hereunder, the Guarantor shall pay, or cause to be paid, such Obligations within five (5) business days of receipt of such demand. Such payments shall be made to: [Insert Revenue Account Wire Instructions]
The Guarantor's obligations hereunder are primary obligations of the Guarantor and are an absolute, unconditional, continuing and irrevocable guaranty of payment and performance and not of collectibility, and are in no way conditioned on or contingent upon any attempt to enforce in whole or in part any liabilities and obligations of the Affiliate Project Party or any other person. Each failure by the Guarantor to pay or perform, as the case may be, any amounts due or any obligations under this Guaranty shall give rise to a separate cause of action hereunder, and separate suits may be brought hereunder as each cause of action arises.
Waivers; Absolute Obligations. The Guarantor hereby waives notice of acceptance of this Guaranty and notice of any obligation or liability to which it may apply, and waives presentment, demand for payment, protest, notice of dishonor or non-payment of any such obligation or liability, suit or the taking of other action by the Project Owner against, and any other notice to, the Affiliate Project Party, the Guarantor or others, notice of entry into the Agreement between Affiliate Project Party and Project Owner and of any amendments, supplements or modifications thereto; or any waiver of consent under the Agreement, including waivers of the payment and performance of the obligations thereunder; and any requirement that suit be brought against, or any other action by the Project Owner be taken against, or any notice of default or other notice be given to, or any demand be made on the Project Owner or any other person, or that any other action be taken or not taken as a condition to the Guarantor's liability for the Obligations under this Guaranty or as a condition to the enforcement of this Guaranty against the Guarantor.

The Project Owner may at any time and from time to time without notice to or consent of the Guarantor and without impairing or releasing the obligations of the Guarantor hereunder: (1) agree with the Affiliate Project Party to make any change in the terms of any obligation or liability of the Affiliate Project Party to the Project Owner, (2) take or fail to take any action of any kind in respect of any security for any obligation or liability of the Affiliate Project Party to the Project Owner, (3) exercise or refrain from exercising any rights against the Affiliate Project Party or others, or (4) compromise or subordinate any obligation or liability of the Affiliate Project Party to the Project Owner including any security therefor.

The liability of the Guarantor under this Guaranty shall be absolute and unconditional irrespective of:

- (a) any lack of validity or enforceability of or defect or deficiency applicable to the Affiliate Project Party in the Agreement or any other documents executed in connection with the Agreement; or
- (b) any modification, extension or waiver of any of the terms of the Agreement; or
- (c) any change in the time, manner, terms or place of payment of or in any other term of, all or any of the Obligations, or any other amendment or waiver of or any consent to departure from the Agreement or any other agreement or instrument executed in connection therewith; or
- (d) failure, omission, delay, waiver or refusal by the Project Owner to exercise, in whole or in part, any right or remedy held by the Project Owner with respect to the Agreement or any transaction under the Agreement; or
- (e) any change in the existence, structure or ownership of the Guarantor or the Affiliate Project Party or Project Owner, or any insolvency, bankruptcy, reorganization or other similar proceeding affecting the Affiliate Project Party or its assets; or
- (f) any defense arising by reason of any bankruptcy, insolvency, reorganization, arrangement, readjustment of debt, liquidation or dissolution proceeding commenced by or against any person, including any discharge of, or bar or stay against collecting, all or any part of the amounts due under this Guaranty (or any interest on all or any part of the amounts due under this Guaranty) in or as a result of any such proceeding, any failure of the Project Owner to file a claim in any such proceeding, or the occurrence of any of the following: (i) the election by the Project Owner, in any bankruptcy proceeding of any person, of the application or non-application of Section 1111(b)(2) of Title 11 of the United States Code entitled "Bankruptcy" (together with any successor statute, and all rules promulgated thereunder, the "Bankruptcy Code"), (ii) any extension of credit or the grant of any lien or encumbrance under Section 364 of the Bankruptcy Code, (iii) any use of cash collateral under Section 363 of the Bankruptcy Code, or (iv) any agreement or stipulation with respect to the provision of adequate protection in any bankruptcy proceeding of any person; or

(g) any duty on the part of the Project Owner to disclose to the Guarantor any facts the Project Owner may now or hereafter know about the Affiliate Project Party, regardless of whether the Project Owner has reason to believe that any such facts materially increase the risk beyond that which the Guarantor intends to assume, or have reason to believe that such facts are unknown to the Guarantor, or have a reasonable opportunity to communicate such facts to the Guarantor, since the Guarantor acknowledges that the Guarantor is fully responsible for being and keeping informed of the financial condition of the Affiliate Project Party and of all circumstances bearing on the risk of non-payment of any amounts due or non-performance of any obligations under this Guarantor.

Without limiting the foregoing, the Guarantor hereby unconditionally and irrevocably waives and relinquishes, to the maximum extent permitted by applicable laws, all rights and remedies accorded to sureties or guarantors and agrees not to assert or take advantage of any such rights or remedies.

3. Bankruptcy Waivers. The Guarantor hereby irrevocably waives, to the extent it may do so under applicable laws, any protection to which it may be entitled under Sections 365(c)(1), 365(c)(2) and 365(e)(2) of the Bankruptcy Code or equivalent provisions of the laws or regulations of any other jurisdiction with respect to any proceedings, or any successor provision of law of similar import, in the event of any Act of Insolvency (as defined in the Agreement) with respect to the Project Owner or the Affiliate Project Party or any other guarantor or surety. Specifically, in the event that the trustee (or similar official) in an Act of Insolvency with respect to the Affiliate Project Party or any other guarantor or surety or the debtor-in-possession takes any action (including the institution of any action, suit or other proceeding for the purpose of enforcing the rights of the relevant Borrower(s), or any other guarantor or surety under this Guaranty), the Guarantor shall, to the fullest extent it may do so under applicable law, not assert any defense, claim or counterclaim denying liability hereunder on the basis that this Guaranty is an executor contract or a "financial accommodation" that cannot be assumed, assigned or enforced or on any other theory directly or indirectly based on Sections 365(c)(1), 365(c)(2) or 365(e)(2) of the Bankruptcy Code, or equivalent provisions of the law or regulations of any other jurisdiction with respect to any proceedings or any successor provision of law of similar import. If an Act of Insolvency with respect to the Affiliate Project Party or any other guarantor or surety shall occur, the Guarantor agrees, after the occurrence of such Act of Insolvency, to reconfirm in writing, to the extent permitted by applicable laws, its pre-petition waiver of any protection to which it may be entitled under Sections 365(c)(1), 365(c)(2) and 365(e)(2) of the Bankruptcy Code or equivalent provisions of the laws or regulations of any other jurisdiction with respect to proceedings and, to give effect to such waiver, the Guarantor consents, to the fullest extent it may do so under applicable law, to the assumption and enforcement of each provision of this Guaranty by the debtor-in-possession or the trustee in bankruptcy of the Affiliate Project Party or of any other guarantor or surety, as the case may be.

- 4. <u>No Assignment.</u> The Guarantor may not assign its rights nor delegate its obligations under this Guaranty, in whole or in part, without prior written consent of the Project Owner, and any purported assignment or delegation absent such consent is void, except for an assignment and delegation of all of the Guarantor's rights and obligations hereunder in whatever form the Guarantor determines may be appropriate to a partnership, corporation, trust or other organization in whatever form that succeeds to all or substantially all of the Guarantor's assets and business and that assumes such obligations by contract, operation of law or otherwise. Upon any such delegation and assumption of obligations, the Guarantor shall be relieved of and fully discharged from all obligations hereunder, whether such obligations arose before or after such delegation and assumption. The Project Owner may, upon notice to the Guarantor, assign its rights hereunder (including any collateral assignment) without the consent of Guarantor to any person to which it assigns its interests in the Agreement.
- 5. Representations and Warranties. The Guarantor hereby represents and warrants for itself, as of the date hereof, that:
 - (a) it is duly organized and validly existing under the laws of its jurisdiction of incorporation, has the corporate power and has obtained all required governmental approvals to comply with and perform its respective obligations under and enter into this Guaranty;
 - (b) this Guaranty has been duly authorized and executed by it and constitutes its valid and legally binding obligation enforceable in accordance with its terms, except as enforceability hereof may be limited by bankruptcy, moratorium, insolvency or other similar laws affecting the enforcement of creditor's rights generally;
 - (c) neither the execution and delivery of this Guaranty nor the compliance with its terms will conflict with or result in a breach of any of the terms, conditions or provisions of, or constitute a default or require any consent which has not been obtained under, any indenture, mortgage, agreement or other instrument or arrangement to which it is a party or by which it or any of its properties or assets are bound, or violate any of the terms or provisions of its organizational documents (including its bylaws) or any governmental approval, judgment, decree or order or any other applicable law;
 - (d) it is, and after giving effect to the transactions contemplated under this Guaranty will be, solvent; and
 - (e) it is not executing this Guaranty with any intention to hinder, delay or defraud any of its present or future creditor or creditors.
- 6. <u>Subrogation</u>. Notwithstanding any payment or payments made by the Guarantor or the exercise by the Project Owner of any of the remedies provided under this Guaranty or any set-off or application of funds of the Guarantor by the Project Owner, the Guarantor hereby waives all rights of subrogation with respect to payments made under this Guaranty until all of the Obligations have been paid in full. Notwithstanding the foregoing, if any amount shall be paid to the Guarantor on account of such subrogation, such amount shall be held by the Guarantor in trust for the Project Owner, segregated from other funds of the Guarantor, and shall be turned over to the Project Owner, in the exact form received by the Guarantor (or duly endorsed by the Guarantor to the Project Owner, if required) to be applied against such amounts in such order as the Project Owner may elect.

7. <u>Notices</u>. All demands, notices and other communications provided for hereunder shall, unless otherwise specifically provided herein, (a) be in writing addressed to the party receiving the notice at the address set forth below or at such other address as may be designated by written notice, from time to time, to the other party, and (b) be effective upon receipt, when mailed by U.S. mail, registered or certified, return receipt requested, postage prepaid, facsimile or personally delivered. Notices shall be sent to the following addresses:

If to Project Owner:

Pacific Ethanol [____]
c/o JT Miller Group LLC
777 Campus Commons Road # 200
Sacramento, California 95825
Attention: John Miller

If to Guarantor:

Pacific Ethanol, Inc. 400 Capitol Mall, Suite 2060 Sacramento, California 95814 Attention: Neil Koehler Telephone: (530) 750-3017 Facsimile: (530) 309-4172

with a copy to

Pacific Ethanol, Inc. 4060 Capital Mall, Suite 2060 Sacramento, CA 95814 Attn: General Counsel

Phone: (916) 403-2130 Fax: (916) 446-3937

8. <u>Cumulative Remedies; Benefits.</u> No failure by the Project Owner to exercise, and no delay by the Project Owner in exercising, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law. The remedies herein provided are cumulative and not exclusive of any remedies provided by law. Nothing in this Guaranty, express or implied, shall give to any person, other than the parties hereto, and each of their successors and permitted assigns under this Guaranty, any benefit or any legal or equitable right or remedy under this Guaranty.

- 9. <u>Reinstatement</u>. This Guaranty shall continue to be effective or be reinstated, as the case may be, if at any time any payment of any of the Obligations are annulled, set aside, invalidated, declared to be fraudulent or preferential, rescinded or must otherwise be returned, refunded or repaid by Project Owner upon the insolvency, bankruptcy, dissolution, liquidation or reorganization of the Affiliate Project Party or any other guarantor, or upon or as a result of the appointment of a receiver or conservator of, or trustee for the Affiliate Project Party or any other guarantor or any substantial part of its property or otherwise, all as though such payment or payments had not been made.
- 10. <u>Governing Law, etc.</u> (A) THIS GUARANTY SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.
- (B) EACH PARTY HERETO IRREVOCABLY AND UNCONDITIONALLY SUBMITS, FOR ITSELF AND ITS PROPERTY, TO THE NON-EXCLUSIVE JURISDICTION OF THE COURTS OF THE STATE OF NEW YORK SITTING IN NEW YORK COUNTY AND OF THE UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF NEW YORK, AND ANY APPELLATE COURT FROM ANY THEREOF, IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS GUARANTY OR FOR RECOGNITION OR ENFORCEMENT OF ANY JUDGMENT, AND EACH OF THE PARTIES HERETO IRREVOCABLY AND UNCONDITIONALLY AGREES THAT ALL CLAIMS IN RESPECT OF ANY SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN SUCH STATE OR, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, IN SUCH FEDERAL COURT. EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW. NOTHING IN THIS GUARANTY SHALL AFFECT ANY RIGHT THAT ANY OF THE PARTIES HERETO MAY OTHERWISE HAVE TO BRING ANY ACTION OR PROCEEDING RELATING TO THIS GUARANTY IN THE COURTS OF ANY JURISDICTION.
- (C) EACH PARTY HERETO IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS GUARANTY. EACH PARTY HERETO IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, THE DEFENSE OF AN INCONVENIENT FORUM TO THE MAINTENANCE OF SUCH ACTION OR PROCEEDING IN ANY SUCH COURT.
- 11. <u>Termination</u>. This Guaranty shall terminate on the earlier of (i) date the Agreement terminates in accordance with the terms thereof and (ii) the date on which the Affiliate Project Party ceases to be an Affiliate of Guarantor.

12. <u>Survivability</u> . If any provision of this Guaranty is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability
of the remaining provisions of this Guaranty shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations
to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of
the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render
unenforceable such provision in any other jurisdiction.

13. <u>Counterparts</u> . This Guaranty may be executed in counterparts (and by different parties hereto in different counterparts), each of which
shall constitute an original, but all of which when taken together shall constitute a single contract. This Guaranty shall become effective when
has been executed by each of the parties hereto. Delivery of an executed counterpart of a signature page of this Guaranty by telecopy or portab
document format ("pdf") shall be effective as delivery of a manually executed counterpart of this Guaranty.

[Signature Page Follows]

Very truly yours,

PACIFIC ETHANOL, INC.

By: /s/ Neil Koehler

Name: Neil M. Koehler Title: Chief Executive Officer

Acknowledged and Accepted by:	
PACIFIC ETHANOL []	
By: Name: Title:	
	B-9

Exhibit C

Operating Protocol

acific Ethanol [], LLC ([PLANT OWNER]) will by the end of day Monday each week submit to Pacific Agricultural Products, LLC PAP) a twelve week corn consumption forecast, (see attached example) that includes weekly consumption and scheduled maintenance owntime. [PLANT OWNER] will promptly notify PAP of any unscheduled downtime or other event or circumstance that will affect forecasted onsumption.
AP will during the 1" week of each month submit to [PLANT OWNER] a monthly train/truck schedule for the next month. Changes to the nonthly train/truck schedule will be communicated to [PLANT OWNER] as the changes are identified.
EX C-1

Exhibit D

Benchmarking

[Please see attached.]

BENCHMARK

We propose the following example benchmarks and monitor criteria. Modifications to the benchmarks are expected to be made periodically as appropriate to reflect with changes in commodity market environments and in company purchase/sales contract pricing mechanisms.

Commodity/Facili ETHANOL	tyPEHC actual data	Benchmark	Market Data Provider		Possible Discrepancy Reasons/Noises
PECOL	Netback ethanol sales price	OPIS Pacific Northwest	Posting as provided by Oil Price	a) b)	Fixed price contracts Other Price index
PEMV	Netback ethanol	OPIS Chicago	Information Service	c)	Shipments distribution in a highly volatile market
	sales price		(OPIS)	d)	Other accounting adjustments items
PES	Netback ethanol sales price	OPIS LA CI-90			
CORN					
ALL	Corn CBOT equivalent	Delivered basis = FOB Midwest + freight	Trade West report	a) b)	Fixed futures as a result of hedge Contracts based on corn futures other than nearby futures (Usually, this happens a couple of weeks before current future contract expires Uneven distribution of corn consumptions (such as shut down) in a highly volatile market
WDG/Syrup					
ALL	Plant co-product netback	WDG Value as a % of Plant delivered corn on DDG equivalent basis	FC Stone Co-product report	a)b)c)	Fixed price contracts Hedge activities WDG freight

ETHANOL

	Plant Sales Netback		Basis (netback)	Benchmark	Actual vs. Benchmark
PECOL		OPIS PNW			
Jan-11	2.42	2.42	(0.04)	2.38	1.4%
Feb-11	2.45	2.50	(0.04)	2.46	-0.2%
Mar-11	2.60	2.63	(0.04)	2.59	0.5%
Apr-11	2.73	2.76	(0.04)	2.72	0.2%
PEMV		OPIS Chicago			
Jan-11	2.38	2.32	0.01	2.33	2.1%
Feb-11	2.47	2.43	0.01	2.44	1.5%
Mar-11	2.54	2.50	0.01	2.51	1.2%
Apr-11	2.65	2.65	0.01	2.66	-0.2%
PES		OPISLA CI-90			
Jan-11	2.47	2.46	(0.01)	2.45	1.0%
Feb-11	2.57	2.53	(0.01)	2.52	2.1%
Mar-11	2.68	2.66	(0.01)	2.65	0.9%
Apr-11	2.80	2.80	(0.01)	2.79	0.1%

CORN

	Plant CBOT Equivalent	Plant basis	Plant Actual Corn dlvd \$/bu	Market	Actual vs. Market
PECOL					
Jan-11	6.27	0.563	6.83	7.18	(0.35)
Feb-11	6.56	0.595	7.15	7.74	(0.59)
Mar-11	6.68	0.519	7.20	7.64	(0.44)
Apr-11	7.22	0.496	7.71	8.35	(0.64)
PEMV					
Jan-11	6.34	0.650	6.99	7.04	(0.06)
Feb-11	6.87	0.650	7.52	7.60	(0.08)
Mar-11	6.90	0.563	7.46	7.50	(0.04)
Apr-11	7.38	0.639	8.02	8.20	(0.18)
PES					
Jan-11	6.22	0.656	6.88	7.18	(0.30)
Feb-11	6.76	0.663	7.42	7.74	(0.32)
Mar-11	6.74	0.619	7.36	7.65	(0.28)
Apr-11	7.21	0.687	7.90	8.36	(0.46)

CO PRODUCTS

Plant Co-product \$/ton Netback	Plant Co-product DM	Plant Actual Corn dlvd \$/bu	Plant Actual Corn dlvd \$/ton	WDG Value % of Corn (1)
·		·	·	``
56.62	30%	6.83	243.91	69.2%
56.85	30%	7.15	255.37	66.4%
60.80	31%	7.20	257.06	68.0%
61.97	31%	7.71	275.44	65.3%
61.90	32%	6.99	249.48	70.5%
67.04	32%	7.52	268.48	71.0%
69.53	32%	7.46	266.45	74.3%
71.74	32%	8.02	286.40	70.5%
59.21	33%	6.88	245.68	65.4%
59.94	33%	7.42	264.98	61.4%
67.17	33%	7.36	262.95	70.3%
71.47	32%	7.90	282.00	70.5%
	\$/ton Netback 56.62 56.85 60.80 61.97 61.90 67.04 69.53 71.74 59.21 59.94 67.17	\$/ton Netback DM 56.62 30% 56.85 30% 60.80 31% 61.97 31% 61.90 32% 67.04 32% 69.53 32% 71.74 32% 59.21 33% 59.94 33% 67.17 33%	\$/ton Netback DM dlvd \$/bu 56.62 30% 6.83 56.85 30% 7.15 60.80 31% 7.20 61.97 31% 7.71 61.90 32% 6.99 67.04 32% 7.52 69.53 32% 7.46 71.74 32% 8.02 59.21 33% 6.88 59.94 33% 7.42 67.17 33% 7.36	\$/ton Netback DM dlvd \$/bu dlvd \$/ton 56.62 30% 6.83 243.91 56.85 30% 7.15 255.37 60.80 31% 7.20 257.06 61.97 31% 7.71 275.44 61.90 32% 6.99 249.48 67.04 32% 7.52 268.48 69.53 32% 7.46 266.45 71.74 32% 8.02 286.40 59.21 33% 6.88 245.68 59.94 33% 7.42 264.98 67.17 33% 7.36 262.95

⁽¹⁾DDG equivalent, adjusted to standard 90% dry matter

[FORM OF]

AMENDED AND RESTATED DISTILLERS GRAINS MARKETING AGREEMENT

([____] PROJECT)

by and between

PACIFIC ETHANOL [_____], LLC

and

PACIFIC AG. PRODUCTS, LLC

Dated as of June 30, 2011

TABLE OF CONTENTS

ARTICLE I	DEFINITIONS; INTERPRETATION	1
1.1	DEFINITIONS	1
1.2	INTERPRETATION	5
ARTICLE II	MARKETING ACTIVITIES	5
2.1	BILATERAL TRANSACTIONS.	5
2.2	STORAGE	6
2.3	OBLIGATIONS OF PROJECT COMPANY.	
2.4	BACK-TO-BACK TRANSACTIONS	
2.5	NETTING	
2.6	TITLE; DELIVERY POINT; NOMINATIONS; MEASUREMENT.	7
2.7	BENCHMARKING	8
A DTICLE III	PAYMENTS	8
AKTICLEIII	FAIMENIS	o
3.1	FEES AND PAYMENTS.	8
3.2	OVERDUE PAYMENTS; BILLING DISPUTE	9
3.3	AUDIT	9
ARTICLE IV	TERM; TERMINATION	10
4.1	TERM	10
4.2	PROJECT COMPANY DEFAULTS AND PAP REMEDIES	10
4.3	PAP DEFAULTS AND PROJECT COMPANY REMEDIES	10
4.4	CHANGE OF CONTROL	11
4.5	EFFECT OF TERMINATION	11
ARTICLE V	INSURANCE	11
		-
5.1	PAP INSURANCE	11
5.2	PAP INSURANCE PREMIUMS AND DEDUCTIBLES	12
A DETICAL EL VIII	LINATE TYONG ON LLADIUTY	10
ARTICLE VI	LIMITATIONS ON LIABILITY	13
6.1	NO CONSEQUENTIAL OR PUNITIVE DAMAGES	13
ARTICLE VII	INDEMNIFICATION	13
7.1	PROJECT COMPANY'S INDEMNITY	13
7.2	PAP'S INDEMNITY	13

ARTICLE VII	IREPRESENTATIONS AND WARRANTIES	13
ARTICLE IX	FORCE MAJEURE	14
9.1	DEFINITION	14
9.2	EFFECT	14
9.3	LIMITATIONS	14
ARTICLE X	DISPUTE RESOLUTION	15
10.1	ATTEMPTS TO SETTLE	15
10.2	RESOLUTION BY EXPERT	
10.3	ARBITRATION	
10.4	CONSEQUENTIAL AND PUNITIVE DAMAGES	
10.5	FINALITY AND ENFORCEMENT OF DECISION	
10.6	COSTS	16
10.7	CONTINUING PERFORMANCE OBLIGATIONS	16
ARTICLE XI	CONFIDENTIALITY	16
ARTICLE XII	ASSIGNMENT AND TRANSFER	16
ARTICLE XII	IMISCELLANEOUS	16
13.1	ENTIRE AGREEMENT	16
13.2	COUNTERPARTS	17
13.3	SURVIVAL	17
13.4	SEVERABILITY	17
13.5	GOVERNING LAW	17
13.6	BINDING EFFECT	17
13.7	NOTICES	17
13.8	AMENDMENT	18
13.9	NO IMPLIED WAIVER	18
EXHIBIT		
Exhibit A:	Form of Guaranty	
Exhibit B:	Operating Protocol	
Exhibit C:	Benchmarking	

This AMENDED AND RESTATED DISTILLERS GRAINS MARKETING AGREEMENT (as amended, amended and restated, supplemented or otherwise modified from time to time, this " <u>Agreement</u> ") is entered into by and between PACIFIC ETHANOL [], LLC, a Delaware limited liability company (" <u>Project Company</u> "), and PACIFIC AG. PRODUCTS, LLC, a California limited liability company (" <u>PAP</u> "), as of June 30, 2011. Project Company and PAP are each individually referred to herein as a " <u>Party</u> ", and collectively are referred to herein as the " <u>Parties</u> ".
RECITALS
A. The Parties hereto were previously party to that certain Distillers Grains Marketing Agreement, dated as of [], 2010 (the "Prior Agreement"), pursuant to which PAP provided certain services to the Project Company.
B. PAP provides marketing services for Distillers Grains (as defined below) from denatured fuel ethanol production facilities.
C. Project Company owns an approximately [] million gallons-per-year denatured fuel ethanol production facility in [], [] (the "Facility") and Project Company has requested that PAP provide Distillers Grains marketing services for the Facility.
D. The Parties desire to amend and restate in its entirety the Prior Agreement and enter into this Amended and Restated Distillers Grains Marketing Agreement pursuant to which PAP will provide such marketing services.
AGREEMENT
NOW, THEREFORE, in consideration of the agreements and covenants hereinafter set forth, and intending to be legally bound, the Parties hereto covenant and agree as follows:
ARTICLE I DEFINITIONS; INTERPRETATION
1.1 <u>Definitions</u> . The following terms shall have the meanings set forth below when used in this Agreement:
"Act of Insolvency" means, with respect to any Person, any of the following: (a) commencement by such Person of a voluntary proceeding under any jurisdiction's bankruptcy, insolvency or reorganization law; (b) the filing of an involuntary proceeding against such Person under any jurisdiction's bankruptcy, insolvency or reorganization law which is not vacated within 60 days after such filing; (c) the admission by such Person of the material allegations of any petition filed against it in any proceeding under any jurisdiction's bankruptcy, insolvency or reorganization law; (d) the adjudication of such Person as bankrupt or insolvent or the winding up or dissolution of such Person; (e) the making by such Person of a general assignment for the benefit of its creditors (assignments for a solvent financing excluded); (f) such Person fails or admits in writing its inability to pay its debts generally as they become due; (g) the appointment of a receiver or an administrator for all or a substantial portion of such Person's assets, which receiver or administrator, if appointed without the consent of such Person, is not discharged within 60 days after its appointment; or (h) the occurrence of any event analogous to any of the foregoing with respect to such Person occurring in any jurisdiction.
1

"Affiliate" of a specified Person means any corporation, partnership, sole proprietorship or other Person which directly or indirectly through one or more intermediaries controls, is controlled by or is under common control with the Person specified. The term "control" means the ownership, either direct or indirect, of twenty-five percent (25%) or more of the voting securities (or comparable equity interests) or other ownership interests of a Person, or the possession, either direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or any other means whatsoever.

"Agreement" has the meaning given to such term in the preamble hereto.

"Asset Management Agreement" means that certain Second Amended and Restated Asset Management Agreement by and among PEI, Pacific Ethanol Stockton, LLC, Pacific Ethanol Holding Co. LLC, Pacific Ethanol Madera LLC, Pacific Ethanol Columbia, LLC and Pacific Ethanol Magic Valley, LLC, dated as of June 30, 2011, as the same may be amended, supplemented or otherwise modified from time to time.

"Bilateral Transaction" means, with respect to each sale of Distillers Grains produced at the Facility by Project Company, a transaction entered into by PAP with one or more Third Parties consisting of one or more forward sales of Distillers Grains.

"Business Day" means any day other than a Saturday, Sunday or a day on which commercial banks in Sacramento, California or New York, New York are required or authorized to be closed.

"Change of Control" has the meaning ascribed thereto in the Credit Agreement.

"Credit Agreement" means the Credit Agreement, dated as of June 25, 2010, by and among Pacific Ethanol Holding Co. LLC, Pacific Ethanol Madera LLC, Pacific Ethanol Stockton, LLC, Pacific Ethanol Magic Valley, LLC, and Pacific Ethanol Columbia, LLC, as Borrowers, Pacific Ethanol Holding Co. LLC, as Borrowers' Agent, WestLB AG, New York Branch, as the administrative agent and the collateral agent, and the lenders parties thereto from time to time, as the same may be amended, supplemented or otherwise modified from time to time.

"DDG" means dried distillers grains produced by Project Company at the Facility.

"Dispute" means a dispute, controversy or claim.

"Distillers Grains" means DDG, WDG and any other form of distillers grain products produced by Project Company at the Facility from time to time.

"Expert" means an expert having sufficient technical expertise to address the matter subject to a Dispute.

"Extension Notice" has the meaning assigned to such term in Section 4.1.

"Facility" has the meaning given to such term in the recitals hereto.

"<u>Financing Documents</u>" means any and all loan agreements, credit agreements (including the Credit Agreement), reimbursement agreements, notes, indentures, bonds, security agreements, pledge agreements, mortgages, guarantee documents, intercreditor agreements, subscription agreements, equity contribution agreements and other agreements and instruments relating to the financing (or refinancing) of the ownership, operation and maintenance of the Facility.

"Financing Parties" means the banks, lenders, noteholders and/or other financial institutions (or an agent or trustee thereof) party to the Financing Documents.

"Force Majeure Event" has the meaning set forth in Section 9.1.

"Good Industry Practice" means any of the practices, methods and acts engaged in or approved by a significant portion of the distillers grains production or marketing (as the case may be) industry during the relevant time period, or any of the practices, methods and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision was made, could have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety and expedition. "Good Industry Practice" is not limited to a single, optimum practice, method or act to the exclusion of others, but rather is intended to include acceptable practices, methods or acts generally accepted in the region.

"Governmental Authority" means any United States federal, state, municipal, local, territorial, or other governmental department, commission, board, bureau, agency, regulatory authority, instrumentality, judicial or administrative body.

"Incentive Fee" means, for each Bilateral Transaction, the product of 5% *multiplied by* the aggregate amount of the Purchase Price for such Bilateral Transaction; *provided, however*, that in no event shall the Incentive Fee for any Bilateral Transaction be less than \$2.00 per ton of Distillers Grains nor greater than \$3.50 per ton of Distillers Grains.

"Incentive Fee (Estimated)" means, for each Bilateral Transaction, the product of 5% multiplied by the aggregate amount of the Purchase Price (Estimated) for such Bilateral Transaction; provided, however, that in no event shall the Incentive Fee (Estimated) for any Bilateral Transaction be less than \$2.00 per ton of Distillers Grains nor greater than \$3.50 per ton of Distillers Grains.

"<u>Law</u>" means any law, statute, act, legislation, bill, enactment, policy, treaty, international agreement, ordinance, judgment, injunction, award, decree, rule, regulation, interpretation, determination, requirement, writ or order of any Governmental Authority.

"Liabilities" has the meaning given to such term in Section 7.1.

"Monthly Date" means the last Business Day of each calendar month.

"NewCo" means New PE Holdco LLC, a Delaware limited liability company and the indirect owner on the date hereof of all the equity interests in Project Company.

- "PAP" has the meaning given to such term in the preamble hereto.
- "PAP Indemnified Person" has the meaning given to such term in Section 7.2.
- "Party" or "Parties" has the meaning given to such term in the <u>preamble</u> hereto.
- "Payment Adjustment Date" has the meaning given to such term in Section 3.1(b).
- "PEI" means Pacific Ethanol, Inc., a Delaware corporation.
- "Person" means and includes natural persons, corporations, limited liability companies, limited partnerships, general partnerships, joint stock companies, joint ventures, associations, companies, trusts, banks, trust companies and other organizations, whether or not legal entities, Governmental Authorities and any other entity.
- "Prime Rate" means the rate per annum listed as the "Prime Rate" in the "Money Rates" section of *The Wall Street Journal* from time to time.
 - "Prior Agreement" has the meaning assigned to such term in the Recitals.
 - "Project Company" has the meaning given to such term in the preamble hereto.
 - "Project Company Indemnified Person" has the meaning given to such term in Section 7.1.
- "<u>Purchase Price</u>" means, with respect to each Bilateral Transaction, the aggregate gross payments received by PAP (or, if the applicable Third Party defaults in its payment obligations to PAP in respect of such Bilateral Transaction, the aggregate amount of gross payments which PAP was entitled to receive) for such Bilateral Transaction from the applicable Third Party.
- "Purchase Price (Estimated)" means, with respect to each Bilateral Transaction, the aggregate amount of gross payments anticipated to be received by PAP for such Bilateral Transaction from the applicable Third Party (as reasonably determined by PAP).
 - "Syrup" means corn condensed distiller's solubles produced by Project Company at the Facility.
- "Tonnage Fees" means all documented fees or taxes payable to any Governmental Authority in connection with the tonnage of Distillers Grains or Syrup produced or marketed within a given jurisdiction.
- "Tonnage Fees (Estimated)" means all estimated fees or taxes payable to any Governmental Authority in connection with the tonnage of Distillers Grains or Syrup produced or marketed within a given jurisdiction.
 - "Third Party" means any Person (other than PEI or a subsidiary thereof) that enters into a Bilateral Transaction with PAP.

"<u>Transportation Costs</u>" means, for each Bilateral Transaction, all actual, out-of-pocket and documented costs and other expenses incurred by or on behalf of PAP in connection with the transportation of Distillers Grains to the applicable Third Party, including truck, rail, barge and/or terminal costs.

"<u>Transportation Costs (Estimated)</u>" means, for each Bilateral Transaction, the aggregate amount of Transportation Costs anticipated to be incurred by PAP in connection with such Bilateral Transaction (as reasonably determined by PAP).

"WDG" means wet distillers grains produced by Project Company at the Facility.

1.2 <u>Interpretation.</u> The following interpretations and rules of construction shall apply to this Agreement: (a) titles and headings are for convenience only and will not be deemed part of this Agreement for purposes of interpretation; (b) unless otherwise stated, references in this Agreement to "Sections" or "Articles" refer, respectively, to Sections or Articles of this Agreement; (c) "including" means "including, but not limited to", and "include" or "includes" means "include, without limitation" or "includes, without limitation"; (d) "hereunder", "herein", "hereto" and "hereof", when used in this Agreement, refer to this Agreement as a whole and not to a particular Section or clause of this Agreement; (e) in the case of defined terms, the singular includes the plural and vice versa; (f) unless otherwise indicated, each reference to a particular Law is a reference to such Law as it may be amended, modified, extended, restated or supplemented from time to time, as well as to any successor Law thereto; (g) unless otherwise indicated, references to agreements shall be deemed to include all subsequent amendments, supplements and other modifications thereto; and (h) unless otherwise indicated, each reference to any Person shall include such Person's successors and permitted assigns.

ARTICLE II MARKETING ACTIVITIES

2.1 <u>Bilateral Transactions</u>.

Subject to the terms hereof, Project Company hereby grants PAP the exclusive right to market, purchase and sell all (a) of Project Company's Distillers Grains (which, as of the date hereof, is approximately 533,000 tons-per-year) commencing June 30, 2011 and continuing through the expiration or early termination of this Agreement, provided, that during the continuance of any default by PAP that would allow Project Company to terminate this Agreement pursuant to Section 4.3 or during the 30-day cure period provided in Section 4.3(c) (notwithstanding such cure period), if PAP is not performing its obligations with respect to marketing the Project Company's Distillers Grains or during the continuance of any Force Majeure Event (including the effects thereof) that renders PAP unable to perform its obligations under this Agreement, then Project Company shall have the right to engage any other Person to market, purchase and sell the Project Company's Distillers Grains and PAP shall not be entitled to any compensation (including Incentive Fees) with respect to any replacement services provided by such Person. PAP shall use its reasonable commercial efforts to solicit, negotiate and enter into, and PAP shall perform, Bilateral Transactions with Third Parties. PAP shall have absolute discretion in the solicitation, negotiation, administration (including the collection of payments), enforcement and execution of Bilateral Transactions and all sales of Distillers Grains produced by the Facility shall be effectuated by Bilateral Transactions. PAP shall not enter into any transaction in respect of the Project Company's Distillers Grains that (i) is not a Bilateral Transaction or (ii) requires deliveries of WDG or DDG more than one hundred (180) days after the date of execution of such transaction, without the prior written consent of the Project Company, which consent may be withheld by Project Company in its discretion. Project Company hereby grants PAP the power and authority necessary to perform its obligations and exercise its rights hereunder.

- (b) As further described in <u>Sections 2.3</u>, <u>2.4</u> and <u>2.6</u> below and except as otherwise provided herein, Project Company shall provide Distillers Grains to PAP free and clear of all liens and encumbrances.
- (c) PAP shall perform its obligations hereunder and under Bilateral Transactions in accordance with this Agreement, applicable Laws and Good Industry Practice and shall use commercially reasonable efforts to maximize the proceeds generated from the sale of Distillers Grains.
- 2.2 <u>Storage</u>. PAP acknowledges that Project Company has only limited storage capacity and PAP agrees that it shall take any Distillers Grains requested by PAP within two days (or such longer period of time as may reasonably be agreed by Project Company) of the time that Project Company has made such Distillers Grains available to PAP.

2.3 Obligations of Project Company.

- (a) Project Company shall provide PAP with all information reasonably requested by PAP, and Project Company shall assist PAP as reasonably requested in the solicitation, negotiation and performance of Bilateral Transactions.
- (b) Notwithstanding anything to the contrary herein, Project Company shall not be responsible for the delivery of any Distillers Grains to PAP during any periods of scheduled Facility maintenance (unless and to the extent the applicable Distillers Grains is available to be delivered to PAP from Project Company's storage facilities); *provided*, that, at any time that PEI or one of its Affiliates is not the asset manager pursuant to the Asset Management Agreement (or any successor agreement), PAP shall have received at least ten Business Days prior notice of such scheduled maintenance (it being acknowledged and agreed that if PAP does not receive at least ten Business Days prior notice, then such maintenance activity shall be deemed to be a mechanical breakdown and covered by clause (c) below for purposes hereof).
- (c) If on any day Project Company is unable to perform its obligations to deliver Distillers Grains under this Agreement due to a mechanical breakdown (including a forced outage of the Facility) that is not a Force Majeure Event and such mechanical breakdown has continued for more than three consecutive days, PAP shall, at Project Company's option and at Project Company's expense, and provided that, at any time that PEI or one of its Affiliates is not the asset manager pursuant to the Asset Management Agreement (or any successor agreement), Project Company provides PAP with prompt notice of its intent to exercise such option, use commercially reasonable efforts to identify and procure replacement distillers grains to be delivered to the Third Party under the applicable Bilateral Transaction. In such event, if and only if the Parties reach agreement as to an alternative delivery point, PAP shall acquire and deliver replacement distillers grains in a quantity sufficient to meet the contract quantity of such Bilateral Transaction at such alternate point (and Project Company shall be responsible for all transportation costs associated therewith). In all other instances, Project Company shall be responsible for any damages incurred by PAP in connection with PAP's failure to perform under the applicable Bilateral Transaction as a result of such mechanical breakdown (it being acknowledged and agreed that PAP shall use commercially reasonable efforts to mitigate the effects of any such mechanical breakdown and Project Company's resulting inability to deliver Distillers Grains.

- (d) At the request of the Project Company, PAP will cause PEI to execute and deliver and maintain in full force and effect a guaranty in the form of Exhibit A hereto.
- 2.4 <u>Back-to-Back Transactions</u>. Each Bilateral Transaction undertaken by PAP shall immediately and automatically, without necessity of further documentation or any action whatsoever by any of the Parties, create and cause to be undertaken according to the terms of this Agreement an equivalent transaction in terms of the obligation to deliver Distillers Grains, the quantity of Distillers Grains sold and the timing for the delivery of such Distillers Grains by Project Company with PAP (as if PAP were the Third Party).
- 2.5 Netting. Netting of amounts due in respect of Bilateral Transactions between PAP and a Third Party may arise in circumstances in which PAP owes amounts to such Third Party in respect of Bilateral Transactions and, at the same time, such Third Party owes amounts to PAP in respect of Bilateral Transactions. In such circumstances, the party owing the greater amount may pay such amount to the other party as reduced by the amount owed to it and both parties will be deemed to have satisfied their obligations thereby. When such netting occurs, for purposes of this Agreement, for all Bilateral Transactions that have been subject to such netting arrangements, PAP shall be deemed to have paid amounts owed by it and to have received amounts owed to it.

2.6 Title; Delivery Point; Nominations; Measurement.

- (a) Project Company shall deliver Distillers Grains to PAP in respect of Bilateral Transactions (or corresponding back-to-back transactions under Section 2.4) via bucket-loader into a receiving truck that will remove such Distillers Grains from the Facility. Title to, risk of loss with respect to and the obligation to transport such Distillers Grains shall pass from Project Company to PAP at the point that such Distillers Grains drop into the applicable receiving truck. The Parties acknowledge that the quality and quantity of Distillers Grains may degrade or shrink after such Distillers Grains is delivered by Project Company to PAP at such delivery point, and the Parties acknowledge that the risk of such degradation or shrinkage and all other risk of loss shall be borne by PAP.
- (b) PAP and Project Company shall use the previously agreed upon operating protocol with respect to the mechanics, timing and process for (i) PAP to communicate to Project Company its Distillers Grains requirements on a monthly, weekly and daily basis, (ii) determining the quantity of Distillers Grains to be stored by Project Company in its storage facilities, and (iii) implementing the Distillers Grains sales contemplated by this Agreement. By mutual agreement, such operating protocol shall be updated from time to time thereafter. A copy of such protocol is attached hereto as Exhibit B.

2.7 <u>Benchmarking.</u> PAP shall furnish on a monthly basis a report benchmarking its performance in accordance with Exhibit C.

ARTICLE III PAYMENTS

3.1 Fees and Payments.

- (a) Within ten days after the date Project Company delivers Distillers Grains to PAP in accordance with <u>Section 2.6(a)</u>, PAP shall pay to Project Company an amount equal to (i) the Purchase Price (Estimated) with respect to the Bilateral Transaction to which such delivery of Distillers Grains relates *minus* (ii) the aggregate amount of Transportation Costs (Estimated) with respect to such Bilateral Transaction *minus* Tonnage Fees (Estimated) with respect to such Bilateral Transaction (it being acknowledged that PAP shall retain for its own account the amount of such Transportation Costs (Estimated), Incentive Fee (Estimated) and Tonnage Fees (Estimated), and that such amount represents an estimate of the net amounts to be paid to Project Company in connection with such Bilateral Transaction). In connection with each such payment, PAP shall deliver to Project Company a statement detailing its calculations of the applicable Purchase Price (Estimated), the applicable Transportation Costs (Estimated), the applicable Incentive Fee (Estimated) and the applicable Tonnage Fees (Estimated).
- (b) Within the first five Business Days of each calendar month (each such date, a "Payment Adjustment Date"), the Parties shall reconcile and "true-up" the actual Purchase Price, Transportation Costs, Incentive Fees and Tonnage Fees for all Bilateral Transactions entered into since the previous Payment Adjustment Date, with the intent of the Parties being that PAP shall make up the difference of any "under estimations" and Project Company shall refund any "over estimations". For example, if there are "under estimations" then PAP shall pay to Project Company an amount equal to:
 - (i) (A) the Purchase Price with respect to such Bilateral Transaction *minus* (B) the Purchase Price (Estimated) with respect to such Bilateral Transaction (to the extent actually paid by PAP to Project Company pursuant to <u>Section 3.1(a)</u>), *minus*
 - (ii) (A) the Transportation Costs with respect to each such Bilateral Transaction *minus* (B) the Transportation Costs (Estimated) with respect to such Bilateral Transaction, *minus*
 - (iii) (A) the Incentive Fee with respect to each such Bilateral Transaction *minus* (B) the Incentive Fee (Estimated) with respect to such Bilateral Transaction, *minus*
 - (iv) (A) the Tonnage Fees with respect to each such Bilateral Transaction *minus* (B) the Tonnage Fees (Estimated) with respect to such Bilateral Transaction.

Each such monthly reconciliation or "true-up" payment shall be paid by PAP or Project Company (as applicable) no later than five Business Days after the applicable Payment Adjustment Date. Each Party acknowledges that Project Company (and not PAP) bears the risk of non-payment by a Third Party in connection with a Bilateral Transaction.

- (c) Notwithstanding anything to the contrary in clause (a) or (b) above, if Project Company defaults in its obligation to provide Distillers Grains to PAP in accordance with the terms of this Agreement (including, without limitation, as contemplated by Section 2.3(c)), then PAP shall be entitled to set-off and deduct from current and/or future payments owed to PAP by Project Company (including the estimated payments pursuant to clause (a) above and the reconciliation and "true-up" payments pursuant to clause (b) above) an amount equal to, as applicable, (i) the amount of damage payments owed by PAP to the applicable Third Party for failure to provide such Distillers Grains and (ii) the cost of any replacement Distillers Grains procured by PAP to satisfy the requirements of any Bilateral Transaction, each as a result of Project Company's failure to perform hereunder net of any revenue received in respect of such Bilateral Transaction.
- 3.2 Overdue Payments; Billing Dispute. If Project Company or PAP, in good faith, disputes the amount of any payment received by it or to be paid by it or set-off pursuant to Section 3.1 above, the disputing Party shall immediately notify the other Party of the basis for the dispute. The Parties will then meet and use their best efforts to resolve any such dispute. If any amount is ultimately determined to be due to or permitted to be set-off by Project Company or PAP (as the case may be), to the extent not previously paid or set-off, (a) PAP (or the Project Company as the case may be) shall pay such amount to Project Company (or PAP, as the case may be) within five Business Days of such determination or (b) PAP (or the Project Company as the case may be) may then set-off such amount (as the case may be). If any Party shall fail to make any payment when due hereunder, such overdue payment shall accrue interest at the Prime Rate plus 2% from the date originally due until the date paid.
- Audit. Notwithstanding the payment of any amount pursuant to this Article III, Project Company shall remain entitled (upon reasonable prior notice, at reasonable times and at PAP's corporate offices) and the administrative agent under the Credit Agreement (and its consultants, as directed by the administrative agent) shall be entitled (upon reasonable prior notice, not more than once per calendar quarter and at PAP's corporate offices) to conduct a subsequent audit and review of (a) all Bilateral Transactions and related records to verify the amount of gross payments, Incentive Fees, Transportation Costs and damage payments and (b) the determination and calculation of the Purchase Price, in each case for a period of two years from and after the applicable Payment Adjustment Date. If, pursuant to such audit and review, it is determined that any amount previously paid by PAP to Project Company did not constitute all of the amounts which should have been paid to Project Company, Project Company shall advise PAP indicating such amount and the reason the amount should have been paid to Project Company and, subject to the next two sentences, PAP shall pay such amount to Project Company within five Business Days of such request along with interest accrued at the Prime Rate plus 5% from the date originally due until the date paid. If the Parties do not agree with respect to any item so noted, the Parties will then meet and use their best efforts to resolve the dispute. If Parties are not able to resolve issues raised by such an audit and review, any disputed items will be resolved in accordance with the provisions of Article X.

ARTICLE IV TERM; TERMINATION

- 4.1 Term. This Agreement shall be effective on the date hereof and, unless earlier terminated in accordance with its terms, shall continue in effect until and including June 30, 2012; provided, that either Project Company or PAP may extend this Agreement for additional one year periods, in each case by written notice to the other (an "Extension Notice") delivered not less than ninety (90) days prior to the end of the original or renewal term, provided further that this Agreement shall nonetheless terminate if the recipient of any such Extension Notice rejects such Extension Notice not more than fifteen (15) days after receipt of the Extension Notice.
- 4.2 <u>Project Company Defaults and PAP Remedies</u>. Upon the occurrence of any of the following events, PAP may exercise such rights and remedies as may be available to it at law or in equity, including the right to terminate this Agreement, by written notice to Project Company, *provided*, that no such notice shall be required for a termination pursuant to clause (b) of this <u>Section 4.2</u>:
- (a) the failure by Project Company to make any payment, deposit or transfer required hereunder within thirty (30) Business Days after the date such payment, deposit or transfer is due, and such failure continues for fifteen (15) Business Days after receipt of written notice from PAP of such failure;
 - (b) the occurrence of an Act of Insolvency with respect to Project Company; or
- (c) the failure of Project Company to perform any of its material obligations under this Agreement and such failure continues for 30 days after receipt of written notice from PAP of such failure; *provided*, that such 30-day period shall be extended for up to an aggregate of 90 days so long as Project Company is diligently attempting to cure such failure.
- 4.3 <u>PAP Defaults and Project Company Remedies</u>. Upon the occurrence of any of the following events, Project Company may exercise such rights and remedies as may be available to it at law or in equity, including the right to terminate this Agreement, by written notice to PAP, *provided*, that no such notice shall be required for a termination pursuant to clause (b) of this Section 4.3:
- (a) the failure by PAP to make any payment, deposit or transfer required hereunder within fifteen (15) Business Days after the date such payment, deposit or transfer is due, and such failure continues for fifteen (15) Business Days after receipt of written notice from Project Company of such failure;
 - (b) the occurrence of an Act of Insolvency with respect to PAP; or

- (c) the failure of PAP to perform any of its material obligations under this Agreement and such failure continues for 30 days after receipt of written notice from Project Company of such failure; *provided*, that such 30-day period shall be extended for up to an aggregate of 90 days so long as PAP is diligently attempting to cure such failure.
- 4.4 <u>Change of Control.</u> This Agreement shall terminate 45 days after the occurrence of (i) any Change of Control with respect to Project Company or any transfer, assignment, sale or other disposition of more than a majority of the membership interests in PAP to any Person that is not an Affiliate of PEI or (ii) any transfer, assignment, sale or other disposition of all or substantially all of the assets comprising the Facility, unless in each case the Parties mutually agree to the contrary.
- 4.5 <u>Effect of Termination</u>. No termination under this <u>Article IV</u> shall release any of the Parties from any obligations arising hereunder prior to such termination, including payment and obligations under any Bilateral Transaction (or such Bilateral Transaction's corresponding back-to-back transaction arising under <u>Section 2.4</u>), that are not fully performed as of the date of such termination. The exercise of the right of a Party to terminate this Agreement, as provided herein, does not preclude such Party from exercising other remedies that are provided herein or are available at law or in equity; *provided*, *however*, that no Party shall have a right to terminate, revoke or treat this Agreement as repudiated other than in accordance with the other provisions of this Agreement; and *provided*, *further*, that the Parties' respective rights upon termination shall be subject to the liability limitations of <u>Article VI</u>. Except as otherwise set forth in this Agreement, remedies are cumulative, and the exercise of, or the failure to exercise, one or more remedies by a Party shall not, to the extent provided by Law, limit or preclude the exercise of, or constitute a waiver of, other remedies by such Party.

ARTICLE V INSURANCE

- 5.1 <u>PAP Insurance</u>. Without limiting any of the other obligations or liabilities of PAP under this Agreement, PAP shall at all times carry and maintain or cause to be carried and maintained, the minimum insurance coverage set forth in this Section:
- (a) PAP shall maintain or cause to be maintained (i) Workers' Compensation insurance in compliance with the workers' compensation laws of the states in which PAP provides services hereunder as extended by the Broad Form All States Endorsements, the United States Longshoreman's and Harbor Workers' Coverage Endorsements on an if-any-exposure basis and the Voluntary Compensation Coverage Endorsement, and (ii) Employer's Liability (including Occupational Disease) coverage with limits of not less than \$1,000,000, which shall cover all of PAP's employees engaged in providing services hereunder.
- (b) PAP shall maintain or cause to be maintained automobile liability insurance for owned (if any), non-owned and hired vehicles with combined single limits for bodily injury/property damage not less than \$1,000,000 per occurrence and containing appropriate no-fault insurance provisions wherever applicable.

- (c) PAP will maintain or cause to be maintained commercial general liability insurance with a limit for bodily injury/property damage of not less than \$1,000,000 per occurrence and \$2,000,000 in the annual aggregate. Such coverage shall include premises/operations, explosion, collapse and underground property damage, broad form contractual, independent contractors, products/completed operations (including operator errors and omissions), broad form property damage, personal injury and incidental professional liability (if not covered under product/completed operations and if commercially available).
- (d) PAP shall maintain or cause to be maintained umbrella liability insurance providing coverage limits in excess of those set forth in Section (a), (b) and (c) above. The limits of this umbrella coverage shall not be less than \$10,000,000 per occurrence and in the annual aggregate.
- (e) PAP shall maintain or cause to be maintained pollution legal liability for sudden and accidental pollution for physical damage and bodily injury to third parties in an amount of \$3,000,000 per occurrence and in the annual aggregate.

The terms and conditions of all insurance policies (including the amount, scope of coverage, deductibles, and self-insured retentions) shall be acceptable in all respects as of the effective date of this Agreement. All insurance carried pursuant to this Section shall conform to the relevant provisions of this Agreement and be with insurance companies which are rated "A-, X" or better by Best's Insurance Guide and Key Ratings, or other insurance companies of recognized responsibility satisfactory to Project Company. Project Company shall be furnished with satisfactory evidence that the foregoing insurance is in effect and Project Company shall be notified 30 calendar days prior to the cancellation or material change of any such coverage. Coverage for the insurance under Section (c) and (d) above shall be written on a claims made basis provided that if the policy is not renewed, PAP shall obtain for the benefit of Project Company an extended reporting period coverage or "tail" of at least three years past the final day of coverage of such policy. PAP shall provide Project Company with evidence that such extended reporting period coverage or "tail" has been obtained. PAP agrees to ensure that the insurance policies outlined in this Section require the insurer to waive subrogation against Project Company, the Financing Parties and their respective Affiliates together with their respective officers, directors, Affiliates and employees and all such Persons shall be an additional insured as their interests may appear with respect to all policies procured by PAP.

5.2 <u>PAP Insurance Premiums and Deductibles.</u> All premiums for insurance coverage procured by PAP pursuant to <u>Section 5.1</u> shall be reimbursed by Project Company upon demand. PAP shall be liable for the payment of all deductibles on insurance policies obtained pursuant to <u>Section 5.1</u>, which amounts shall not be reimbursed by Project Company, provided that, to the extent that a claim under a policy described in <u>Section 5.1</u> is attributable to Project Company's (including its employees' or agents') gross negligence or willful misconduct, Project Company shall be liable for the entire amount of such deductible. In no event shall any premiums, deductibles or any losses in excess of insurance coverage be reimbursed by Project Company hereunder.

ARTICLE VI LIMITATIONS ON LIABILITY

- 6.1 <u>No Consequential or Punitive Damages.</u> In no event shall either Party be liable to any other Party by way of indemnity or by reason of any breach of contract or of statutory duty or by reason of tort (including negligence or strict liability) or otherwise for any loss of profits, loss of revenue, loss of use, loss of production, loss of contracts or for any incidental, indirect, special or consequential or punitive damages of any other kind or nature whatsoever that may be suffered by such other Party, including any losses for which such other Party has insurance to the extent proceeds of insurance have been recovered for such losses.
- 6.2 <u>Breach.</u> Notwithstanding anything to the contrary herein, any failure, breach or default by Project Company under this Agreement that is caused by or is a result of a failure, breach or default by PEI under the Asset Management Agreement will not be a breach of this Agreement.

ARTICLE VII INDEMNIFICATION

- 7.1 <u>Project Company's Indemnity.</u> Project Company shall defend, indemnify and hold harmless PAP and its Affiliates (and each officer, director, employee, shareholder, partner, member or agent of PAP and its Affiliates) (each, a "<u>Project Company Indemnified Person</u>") from and against any and all third party claims, actions, damages, expenses (including reasonable and documented attorneys' fees and expenses), losses, settlements or liabilities (collectively, "<u>Liabilities</u>") incurred or asserted against any Project Company Indemnified Person (a) as a result of any failure on the part of Project Company to perform Project Company's obligations under this Agreement (including with respect to any back-to-back transaction under <u>Section 2.4</u>), or (b) arising out of or in any way connected with the grossly negligent acts or omissions of Project Company or its Affiliates.
- 7.2 <u>PAP's Indemnity.</u> PAP shall defend, indemnify and hold harmless Project Company and its Affiliates (and each officer, director, employee, shareholder, partner, member or agent of Project Company and their Affiliates) (each, a "<u>PAP Indemnified Person</u>") from and against any and all third party Liabilities incurred or asserted against any PAP Indemnified Person (a) as a result of any failure on the part of PAP to perform its obligations under this Agreement (including with respect to any Bilateral Transaction), or (b) arising out of or in any way connected with the grossly negligent acts or omissions of PAP or its Affiliates.

ARTICLE VIII REPRESENTATIONS AND WARRANTIES

Each Party represents that (i) it is duly organized under its jurisdiction of formation and in good standing in each jurisdiction where its failure to so qualify could have a material adverse affect on its ability to perform its obligations hereunder, (ii) it has all necessary power and authority to enter into this Agreement, (iii) it has duly authorized, executed and delivered this Agreement and (iv) this Agreement constitutes a legal, valid and binding obligation of such Party enforceable in accordance with its terms, subject to bankruptcy, reorganization, moratorium or other similar laws affecting the enforcement of the rights of creditors generally and subject to general principles of equity.

ARTICLE IX FORCE MAJEURE

- 9.1 <u>Definition</u>. As used herein, "<u>Force Majeure Event</u>" means any cause(s) which render(s) a Party wholly or partly unable to perform its obligations under this Agreement (other than obligations to make payments when due), and which are neither reasonably within the control of such Party nor the result of the fault or negligence of such Party, and which occur despite all reasonable attempts to avoid, mitigate or remedy, and shall include acts of God, war, riots, civil insurrections, cyclones, hurricanes, floods, fires, explosions, earthquakes, lightning, storms, chemical contamination, epidemics or plagues, acts or campaigns of terrorism or sabotage, blockades, embargoes, accidents or interruptions to transportation, trade restrictions, acts of any Governmental Authority after the date of this Agreement, strikes and other labor difficulties (other than with respect to its own employees), and other events or circumstances beyond the reasonable control of such Party. Mechanical breakdown (including a forced outage of the Facility) that continues for more than five consecutive days shall be deemed not to be "Force Majeure Event" unless such mechanical breakdown resulted from or was caused by a separate "Force Majeure Event."
- 9.2 Effect. A Party claiming relief as a result of a Force Majeure Event shall give the other Parties written notice within five Business Days of becoming aware of the occurrence of the Force Majeure Event, or as soon thereafter as practicable, describing the particulars of the Force Majeure Event, and will use reasonable efforts to remedy its inability to perform as soon as possible. If the Force Majeure Event (including the effects thereof) continues for fifteen consecutive days, the affected Party shall report to the other Parties the status of its efforts to resume performance and the estimated date thereof. If the Force Majeure Event (including the effects thereof) continues for 180 consecutive days, either Party may terminate this Agreement for convenience. If the affected Party was not able to resume performance prior to or at the time of the report to the other Party of the onset of the Force Majeure Event, then it will report in writing to the other Party when it is again able to perform. If a Party fails to give timely notice, the excuse for its non-performance shall not begin until notice is given.
- 9.3 <u>Limitations</u>. Any obligation(s) of a Party (other than an obligation to make payments when due) may be temporarily suspended during any period such Party is unable to perform such obligation(s) by reason of the occurrence of a Force Majeure Event, but only to the extent of such inability to perform, *provided*, that:
- (a) the suspension of performance is of no greater scope and of no longer duration than is reasonably required by the Force Majeure Event; and
 - (b) the Party claiming the occurrence of the Force Majeure Event bears the burden of proof.

ARTICLE X DISPUTE RESOLUTION

- Attempts to Settle. In the event that a Dispute between the Parties arises under, out of or in relation to, this Agreement, the Parties shall attempt in good faith to settle such Dispute by mutual discussions within fifteen Business Days after the date that an aggrieved Party gives written notice of the Dispute to the other Party. In the event that a Dispute is not resolved by discussion in accordance with the preceding sentence within the time period set forth therein, the Parties shall refer the Dispute to their respective senior officers for further consideration and attempted resolution within fifteen Business Days after the Dispute has been referred to such individuals (or such longer period as the Parties may agree).
- Resolution by Expert. If the Parties shall have failed to resolve the Dispute within fifteen Business Days after the date that the Parties referred the Dispute to their senior officers, then, provided the Parties shall so agree, the Dispute may be submitted for resolution by an Expert, such Expert to be appointed by the mutual agreement of the Parties. Proceedings before an Expert shall be held in Sacramento, California (or any other location agreed to by the Parties). The Expert shall apply to such proceedings the substantive law of the State of New York in effect at the time of such proceedings. The decision of the Expert shall be final and binding upon the Parties. In the event that (a) the Parties cannot agree on the appointment of an Expert within ten Business Days after the date that the Parties agreed to submit the Dispute for resolution by the Expert or (b) the Expert fails to resolve such Dispute within 60 days after the Parties have submitted such Dispute to the Expert, then any Party may file a demand for arbitration in writing in accordance with Section 10.3.
- Arbitration. Any Dispute that has not been resolved following the procedures set forth in Section 10.1 or 10.2 shall be settled by binding arbitration in Sacramento, California (or any other location agreed to by the Parties) before a panel of three arbitrators. Such arbitration shall be conducted in accordance with the Commercial Arbitration Rules of the American Arbitration Association as in effect on the date of execution of this Agreement. Such arbitration shall be governed by the laws of the State of New York. If arbitration proceedings have been initiated pursuant to this Section 10.3 and raise issues of fact or law which, in whole or in part, are substantially the same as issues of fact or law already pending in arbitration proceedings involving the applicable Parties, such issues shall be consolidated with the issues in the ongoing proceedings. THE PARTIES HEREBY AGREE THAT THE PROCEDURES SET FORTH IN THIS Article X SHALL BE THE EXCLUSIVE DISPUTE RESOLUTION PROCEDURES APPLICABLE TO ANY DISPUTE, CONTROVERSY OR CLAIM UNDER THIS AGREEMENT AND, EXCEPT AS SET FORTH IN SECTION 10.5, THE PARTIES HEREBY WAIVE ALL RIGHTS TO A COURT TRIAL OR TRIAL BY JURY WITH RESPECT TO ANY DISPUTE, CONTROVERSY OR CLAIM UNDER THIS AGREEMENT.
 - 10.4 Consequential and Punitive Damages. Awards of Experts and arbitral panels shall be subject to the provisions of Article VI.
- 10.5 <u>Finality and Enforcement of Decision</u>. Any decision or award of an Expert or a majority of an arbitral panel, as applicable, shall be final and binding upon the Parties. Each of the Parties agrees that the arbitral award may be enforced against it or its assets wherever they may be found and that a judgment upon the arbitral award may be entered in any court having jurisdiction thereof.

- 10.6 <u>Costs</u>. The costs of submitting a Dispute to an Expert shall be shared equally among the Parties involved in the Dispute, unless the arbitral panel or the Expert determines otherwise. The costs of arbitration shall be paid in accordance with the decision of the arbitral panel pursuant to the Commercial Arbitration Rules of the American Arbitration Association as in effect on the date of execution of this Agreement.
- 10.7 <u>Continuing Performance Obligations</u>. While a Dispute is pending, each Party shall continue to perform its obligations under this Agreement, unless such Party is otherwise entitled to suspend its performance hereunder or terminate this Agreement in accordance with the terms hereof.

ARTICLE XI CONFIDENTIALITY

Each Party and its Affiliates shall treat as confidential the data and information in their possession regarding the Facility, the other Parties or any Affiliate of any other Party, unless: (a) the applicable other Party agrees in writing to the release of such data or information; (b) such data or information becomes publicly available other than through the wrongful actions of the disclosing Party or the disclosing Party's Affiliate; (c) such data or information was in the possession of the receiving Party or the receiving Party's Affiliate prior to receipt thereof from the disclosing Party with no corresponding confidentiality obligation; or (d) such data or information is required by Law to be disclosed. Notwithstanding the generality of the foregoing, any Party may disclose data and information to (i) the officers, directors, managers, partners, members, employees and Affiliates of such Party, but only to the extent such Persons have a need to know such information for purposes of permitting such Party to perform its obligations hereunder, (ii) any successors in interest and permitted assigns of such Party, (iii) any actual or potential Financing Parties or actual or potential lenders to PEI or any subsidiary thereof, and (iv) any potential equity investors in such Party or its Affiliates or acquirer or potential acquirer of the Project or all or any of the equity interests in Newco or any subsidiary thereof; *provided*, that any Person who receives confidential data and information pursuant to an exception contained in clauses (ii) through (iv) of this Article agrees to confidentiality provisions at least as restrictive as the provisions set forth herein.

ARTICLE XII ASSIGNMENT AND TRANSFER

No Party shall assign this Agreement or any of its rights or obligations hereunder without first obtaining the prior written consent of (a) in the case of Project Company, PAP, or (b) in the case of PAP, Project Company, *provided*, that any Party shall be entitled to assign its rights hereunder (as collateral security or otherwise) for financing purposes (including a collateral assignment to any Financing Parties) without the consent of any other Party.

ARTICLE XIII MISCELLANEOUS

13.1 <u>Entire Agreement</u>. This Agreement contains the entire agreement between the Parties with respect to the subject matter hereof and supersedes all prior agreements, negotiations and understandings among the Parties with respect to such subject matter. Nothing in this Agreement shall be construed as creating a partnership or joint venture between the Parties.

- 13.2 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts and each such counterpart shall be deemed to be an original instrument, but all such counterparts together shall constitute one and the same agreement.
- 13.3 <u>Survival</u>. Cancellation, expiration or earlier termination of this Agreement shall not relieve the Parties of obligations that by their nature should survive such cancellation, expiration or termination, including remedies, limitations on liability, promises of indemnity and payment, and confidentiality. Without limiting the generality of the foregoing, the following provisions of this Agreement shall survive: Ariticles <u>III, IV, VII, X</u> and <u>XI</u> and <u>Section 13.3, 13.4, 13.5, 13.6, 13.8</u> and <u>13.9</u>.
- 13.4 <u>Severability.</u> In the event any one or more of the provisions contained in this Agreement should be held invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein shall not in any way be affected or impaired thereby. The Parties shall endeavor in good-faith negotiations to replace the invalid, illegal or unenforceable provisions with valid provisions, the economic and practical effect of which comes as close as possible to that of the invalid, illegal or unenforceable provisions.
- 13.5 Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of New York without regard to the principles of conflicts of laws thereof.
- Binding Effect. This Agreement shall be binding upon and shall inure to the benefit of the Parties hereto and their respective successors and permitted assigns. This Agreement is not made for the benefit of any Person or entity not a party hereto, and nothing in this Agreement shall be construed as giving any Person or entity, other than the Parties and their respective successors and permitted assigns, any right, remedy or claim under or in respect of this Agreement or any provision hereof.
- Notices. All notices or other communications which are required or permitted hereunder shall be in writing and shall be deemed sufficiently given (a) upon delivery, if delivered personally, (b) the day the notice is received, if it is delivered by overnight courier or certified or registered mail, postage prepaid, or (c) upon the effective receipt of electronic transmission, facsimile, telex or telegram (with effective receipt being deemed to occur upon the sender's receipt of confirmation of successful transmission of such notice or communication), to the addresses set forth below or such other address as the addressee may have specified in a notice duly given to sender as provided herein:

If to PAP:

Pacific Ag. Products, LLC c/o Pacific Ethanol, Inc. 400 Capitol Mall Suite 2060

Sacramento, California 95814 Attention Neil Koehler Telephone: (916) 403-2123 Facsimile: (916) 446-3936

With a copy to:

Pacific Ethanol, Inc.
400 Capitol Mall, Suite 2060
Sacramento, California 95814
Attention General Counsel
Telephone: (916) 403-2130
Facsimile: (916) 403-2785

If to Project Company:

Pacific Ethanol [_____], LLC c/o JT Miller Group LLC 777 Campus Commons Road # 200 Sacramento, California 95825 Attention John Miller Telephone: (916) 565-7422 Facsimile: (916) 565-7423

- 13.8 <u>Amendment</u>. No Party hereto shall be bound by any termination, amendment, supplement, waiver or modification of any term hereof unless such Party shall have consented thereto in writing.
- 13.9 <u>No Implied Waiver</u>. No delay or failure on the part of any Party in exercising any rights hereunder, and no partial or single exercise thereof, shall constitute a waiver of such rights or of any other rights hereunder.

[THE REMAINDER OF THIS PAGE HAS BEEN LEFT BLANK INTENTIONALLY]

IN WITNESS WHEREOF, this Amended and Restated Distille hereto as of the date first written above.	ers Grains Marketing Agreement has been duly executed by the Parties
PA	ACIFIC ETHANOL [], LLC
Ву	7: <u>/s/</u>
	Name: Title:
P	ACIFIC AG. PRODUCTS, LLC
В	y: /s/ Neil M. Koehler
	Name: Neil M. Koehler Title: Chief Executive Officer
[Signature Page to Distillers Grains Ma	arketing Agreement – []]

Exhibit A

[Form of PEI Guaranty]

[Please see attached.]

Form of Parent Guaranty (Amended and Restated Distillers Grains Marketing Agreement)

Pacific Ethanol [], LLC c/o JT Miller Group LLC 777 Campus Commons Road # 200 Sacramento, California, 95825 Attention: John Miller
Re: Guaranty of Affiliate Project Party Obligations under Amended and Restated Distillers Grains Marketing Agreement
Ladies and Gentlemen:
1. The Guaranty. For value received, Pacific Ethanol, Inc., a corporation organized under the laws of Delaware (the "Guarantor"), hereby unconditionally and absolutely guarantees the prompt and complete payment and performance when due, whether by acceleration or otherwise, of all obligations and liabilities, whether now in existence or hereafter arising (the "Obligations"), of Pacific Ag. Products, LLC, a California limited liability company (the "Affiliate Project Party"), to Pacific Ethanol [], LLC, a Delaware limited liability company (the "Project Owner"), under that certain Amended and Restated Distillers Grains Marketing Agreement, dated as of,, 2011 (the "Agreement"); provided that the Guarantor's maximum aggregate liability under this Guaranty shall not exceed the maximum amount which the Affiliate Project Party owes the Project Owner under the Agreement.
Following any demand by the Project Owner for payment hereunder, the Guarantor shall pay, or cause to be paid, such Obligations within five (5) business days of receipt of such demand. Such payments shall be made to: [Insert Revenue Account Wire Instructions]
The Guarantor's obligations hereunder are primary obligations of the Guarantor and are an absolute, unconditional, continuing and irrevocable guaranty of payment and performance and not of collectibility, and are in no way conditioned on or contingent upon any attempt to enforce in whole or in part any liabilities and obligations of the Affiliate Project Party or any other person. Each failure by the Guarantor to payor perform, as the case may be, any amounts due or any obligations under this Guaranty shall give rise to a separate cause of action hereunder, and separate suits may be brought hereunder as each cause of action arises.
Waivers; Absolute Obligations. The Guarantor hereby waives notice of acceptance of this Guaranty and notice of any obligation or liability to which it may apply, and waives presentment, demand for payment, protest, notice of dishonor or non-payment of any such obligation or liability, suit or the taking of other action by the Project Owner against, and any other notice to, the Affiliate Project Party, the Guarantor or others, notice of entry into the Agreement between Affiliate Project Party and Project Owner and of any amendments, supplements or modifications thereto; or any waiver of consent under the Agreement, including waivers of the payment and performance of the obligations thereunder; and any requirement that suit be brought against, or any other action by the Project Owner be taken against, or any notice of default or other notice be given to, or any demand be made on the Project Owner or any other person, or that any other action be taken or not taken as a condition to the Guarantor's liability for the Obligations under this Guaranty or as a condition to the enforcement of this Guaranty against the Guarantor.
A-2

The Project Owner may at any time and from time to time without notice to or consent of the Guarantor and without impairing or releasing the obligations of the Guarantor hereunder: (1) agree with the Affiliate Project Party to make any change in the terms of any obligation or liability of the Affiliate Project Party to the Project Owner, (2) take or fail to take any action of any kind in respect of any security for any obligation or liability of the Affiliate Project Party to the Project Owner, (3) exercise or refrain from exercising any rights against the Affiliate Project Party or others, or (4) compromise or subordinate any obligation or liability of the Affiliate Project Party to the Project Owner including any security therefor.

The liability of the Guarantor under this Guaranty shall be absolute and unconditional irrespective of:

- (a) any lack of validity or enforceability of or defect or deficiency applicable to the Affiliate Project Party in the Agreement or any other documents executed in connection with the Agreement; or
- (b) any modification, extension or waiver of any of the terms of the Agreement; or
- (c) any change in the time, manner, terms or place of payment of or in any other term of, all or any of the Obligations, or any other amendment or waiver of or any consent to departure from the Agreement or any other agreement or instrument executed in connection therewith; or
- (d) failure, omission, delay, waiver or refusal by the Project Owner to exercise, in whole or in part, any right or remedy held by the Project Owner with respect to the Agreement or any transaction under the Agreement; or
- (e) any change in the existence, structure or ownership of the Guarantor or the Affiliate Project Party or Project Owner, or any insolvency, bankruptcy, reorganization or other similar proceeding affecting the Affiliate Project Party or its assets; or
- (f) any defense arising by reason of any bankruptcy, insolvency, reorganization, arrangement, readjustment of debt, liquidation or dissolution proceeding commenced by or against any person, including any discharge of, or bar or stay against collecting, all or any part of the amounts due under this Guaranty (or any interest on all or any part of the amounts due under this Guaranty) in or as a result of any such proceeding, any failure of the Project Owner to file a claim in any such proceeding, or the occurrence of any of the following: (i) the election by the Project Owner, in any bankruptcy proceeding of any person, of the application or non-application of Section 1111(b)(2) of Title 11 of the United States Code entitled "Bankruptcy" (together with any successor statute, and all rules promulgated thereunder, the "Bankruptcy Code"), (ii) any extension of credit or the grant of any lien or encumbrance under Section 364 of the Bankruptcy Code, (iii) any use of cash collateral under Section 363 of the Bankruptcy Code, or (iv) any agreement or stipulation with respect to the provision of adequate protection in any bankruptcy proceeding of any person; or

(g) any duty on the part of the Project Owner to disclose to the Guarantor any facts the Project Owner may now or hereafter know about the Affiliate Project Party, regardless of whether the Project Owner has reason to believe that any such facts materially increase the risk beyond that which the Guarantor intends to assume, or have reason to believe that such facts are unknown to the Guarantor, or have a reasonable opportunity to communicate such facts to the Guarantor, since the Guarantor acknowledges that the Guarantor is fully responsible for being and keeping informed of the financial condition of the Affiliate Project Party and of all circumstances bearing on the risk of non-payment of any amounts due or non-performance of any obligations under this Guarantor.

Without limiting the foregoing, the Guarantor hereby unconditionally and irrevocably waives and relinquishes, to the maximum extent permitted by applicable laws, all rights and remedies accorded to sureties or guarantors and agrees not to assert or take advantage of any such rights or remedies.

3. Bankruptcy Waivers. The Guarantor hereby irrevocably waives, to the extent it may do so under applicable laws, any protection to which it may be entitled under Sections 365(c)(1), 365(c)(2) and 365(e)(2) of the Bankruptcy Code or equivalent provisions of the laws or regulations of any other jurisdiction with respect to any proceedings, or any successor provision of law of similar import, in the event of any Act of Insolvency (as defined in the Agreement) with respect to the Project Company or the Affiliate Project Party or any other guarantor or surety. Specifically, in the event that the trustee (or similar official) in an Act of Insolvency with respect to the Affiliate Project Party or any other guarantor or surety or the debtor-in-possession takes any action (including the institution of any action, suit or other proceeding for the purpose of enforcing the rights of the relevant Borrower(s), or any other guarantor or surety under this Guaranty), the Guarantor shall, to the fullest extent it may do so under applicable law, not assert any defense, claim or counterclaim denying liability hereunder on the basis that this Guaranty is an executor contract or a "financial accommodation" that cannot be assumed, assigned or enforced or on any other theory directly or indirectly based on Sections 365(c)(1), 365(c)(2) or 365(e)(2) of the Bankruptcy Code, or equivalent provisions of the law or regulations of any other jurisdiction with respect to any proceedings or any successor provision of law of similar import. If an Act of Insolvency with respect to the Affiliate Project Party or any other guarantor or surety shall occur, the Guarantor agrees, after the occurrence of such Act of Insolvency, to reconfirm in writing, to the extent permitted by applicable laws, its pre-petition waiver of any protection to which it may be entitled under Sections 365(c)(1), 365(c)(2) and 365(e)(2) of the Bankruptcy Code or equivalent provisions of the laws or regulations of any other jurisdiction with respect to proceedings and, to give effect to such waiver, the Guarantor consents, to the fullest extent it may do so under applicable law, to the assumption and enforcement of each provision of this Guaranty by the debtor-in-possession or the trustee in bankruptcy of the Affiliate Project Party or of any other guarantor or surety, as the case may be.

- 4. <u>No Assignment.</u> The Guarantor may not assign its rights nor delegate its obligations under this Guaranty, in whole or in part, without prior written consent of the Project Owner, and any purported assignment or delegation absent such consent is void, except for an assignment and delegation of all of the Guarantor's rights and obligations hereunder in whatever form the Guarantor determines may be appropriate to a partnership, corporation, trust or other organization in whatever form that succeeds to all or substantially all of the Guarantor's assets and business and that assumes such obligations by contract, operation of law or otherwise. Upon any such delegation and assumption of obligations, the Guarantor shall be relieved of and fully discharged from all obligations hereunder, whether such obligations arose before or after such delegation and assumption. The Project Owner may, upon notice to the Guarantor, assign its rights hereunder (including any collateral assignment) without the consent of Guarantor to any person to which it assigns its interests in the Agreement.
- 5. Representations and Warranties. The Guarantor hereby represents and warrants for itself, as of the date hereof, that:
 - (a) it is duly organized and validly existing under the laws of its jurisdiction of incorporation, has the corporate power and has obtained all required governmental approvals to comply with and perform its respective obligations under and enter into this Guaranty;
 - (b) this Guaranty has been duly authorized and executed by it and constitutes its valid and legally binding obligation enforceable in accordance with its terms, except as enforceability hereof may be limited by bankruptcy, moratorium, insolvency or other similar laws affecting the enforcement of creditor's rights generally;
 - (c) neither the execution and delivery of this Guaranty nor the compliance with its terms will conflict with or result in a breach of any of the terms, conditions or provisions of, or constitute a default or require any consent which has not been obtained under, any indenture, mortgage, agreement or other instrument or arrangement to which it is a party or by which it or any of its properties or assets are bound, or violate any of the terms or provisions of its organizational documents (including its bylaws) or any governmental approval, judgment, decree or order or any other applicable law;
 - (d) it is, and after giving effect to the transactions contemplated under this Guaranty will be, solvent; and
 - (e) it is not executing this Guaranty with any intention to hinder, delay or defraud any of its present or future creditor or creditors.
- 6. <u>Subrogation</u>. Notwithstanding any payment or payments made by the Guarantor or the exercise by the Project Owner of any of the remedies provided under this Guaranty or any set-off or application of funds of the Guarantor by the Project Owner, the Guarantor hereby waives all rights of subrogation with respect to payments made under this Guaranty until all of the Obligations have been paid in full. Notwithstanding the foregoing, if any amount shall be paid to the Guarantor on account of such subrogation, such amount shall be held by the Guarantor in trust for the Project Owner, segregated from other funds of the Guarantor, and shall be turned over to the Project Owner, in the exact form received by the Guarantor (or duly endorsed by the Guarantor to the Project Owner, if required) to be applied against such amounts in such order as the Project Owner may elect.

7. <u>Notices</u>. All demands, notices and other communications provided for hereunder shall, unless otherwise specifically provided herein, (a) be in writing addressed to the party receiving the notice at the address set forth below or at such other address as may be designated by written notice, from time to time, to the other party, and (b) be effective upon receipt, when mailed by U.S. mail, registered or certified, return receipt requested, postage prepaid, facsimile or personally delivered. Notices shall be sent to the following addresses:

If to Project Owner:

Pacific Ethanol [_____], LLC c/o JT Miller Group LLC 777 Campus Commons Road # 200 Sacramento, California, 95825 Attention: John Miller

If to Guarantor:

Pacific Ethanol, Inc. 400 Capitol Mall, Suite 2060 Sacramento, California 95814 Attention: Neil Koehler Telephone: (530) 750-3017 Facsimile: (530) 309-4172

- 8. <u>Cumulative Remedies; Benefits.</u> No failure by the Project Owner to exercise, and no delay by the Project Owner in exercising, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law. Nothing in this Guaranty, express or implied, shall give to any person, other than the parties hereto, and each of their successors and permitted assigns under this Guaranty, any benefit or any legal or equitable right or remedy under this Guaranty.
- 9. <u>Reinstatement</u>. This Guaranty shall continue to be effective or be reinstated, as the case may be, if at any time any payment of any of the Obligations are annulled, set aside, invalidated, declared to be fraudulent or preferential, rescinded or must otherwise be returned, refunded or repaid by Project Owner upon the insolvency, bankruptcy, dissolution, liquidation or reorganization of the Affiliate Project Party or any other guarantor, or upon or as a result of the appointment of a receiver or conservator of, or trustee for the Affiliate Project Party or any other guarantor or any substantial part of its property or otherwise, all as though such payment or payments had not been made.

- 10. <u>Governing Law, etc.</u> (A) THIS GUARANTY SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.
- (B) EACH PARTY HERETO IRREVOCABLY AND UNCONDITIONALLY SUBMITS, FOR ITSELF AND ITS PROPERTY, TO THE NON-EXCLUSIVE JURISDICTION OF THE COURTS OF THE STATE OF NEW YORK SITTING IN NEW YORK COUNTY AND OF THE UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF NEW YORK, AND ANY APPELLATE COURT FROM ANY THEREOF, IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS GUARANTY OR FOR RECOGNITION OR ENFORCEMENT OF ANY JUDGMENT, AND EACH OF THE PARTIES HERETO IRREVOCABLY AND UNCONDITIONALLY AGREES THAT ALL CLAIMS IN RESPECT OF ANY SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN SUCH STATE OR, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, IN SUCH FEDERAL COURT. EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW. NOTHING IN THIS GUARANTY SHALL AFFECT ANY RIGHT THAT ANY OF THE PARTIES HERETO MAY OTHERWISE HAVE TO BRING ANY ACTION OR PROCEEDING RELATING TO THIS GUARANTY IN THE COURTS OF ANY JURISDICTION.
- (C) EACH PARTY HERETO IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS GUARANTY. EACH PARTY HERETO IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, THE DEFENSE OF AN INCONVENIENT FORUM TO THE MAINTENANCE OF SUCH ACTION OR PROCEEDING IN ANY SUCH COURT.
- 11. <u>Termination</u>. This Guaranty shall terminate on the earlier of (i) date the Agreement terminates in accordance with the terms thereof and (ii) the date on which the Affiliate Project Party ceases to be an Affiliate of Guarantor.

12. <u>Survivability</u> . If any provision of this Guaranty is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability
of the remaining provisions of this Guaranty shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations
to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of
the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render
unenforceable such provision in any other jurisdiction.

13. <u>Counterparts</u>. This Guaranty may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Guaranty shall become effective when it has been executed by each of the parties hereto. Delivery of an executed counterpart of a signature page of this Guaranty by telecopy or portable document format ("pdf") shall be effective as delivery of a manually executed counterpart of this Guaranty.

[Signature Page Follows]

Very truly yours,

PACIFIC ETHANOL, INC.

By: /s/ Neil M. Koehler

Name: Neil M. Koehler Title: Chief Executive Officer

Acknowledged and Accepted by:	
PACIFIC ETHANOL [], LLC	
By: Name: Title:	
A-9	

Exhibit B

Operating Protocol

Pacific Ethanol [], LLC ([]) will by the end of day Monday each week submit to Pacific Agricultural Products, LLC (PAP) a 12 week production forecast, (see attached example) that includes weekly production and scheduled maintenance downtime. [] will promptly notify PAP of any unscheduled downtime or other event or circumstance that will affect forecasted production.
PAP will by 12 noon on Friday each week submit to [] a weekly truck schedule for the following week. Changes to the weekly truck schedule will be communicated to [] as the changes are identified.
B-1

Exhibit C

Benchmarking

[Please see attached.]

Exhibit D

Benchmarking

BENCHMARK

We propose the following example benchmarks and monitor criteria. Modifications to the benchmarks are expected to be made periodically as appropriate to reflect with changes in commodity market environments and in company purchase/sales contract pricing mechanisms.

Commodity/Facility PEHC actual data F		Benchmark Market Data Provider			Possible Discrepancy Reasons/Noises	
ETHANOL						
PECOL	Netback ethanol	OPIS Pacific Northwest	Posting as provided by	a)	Fixed price contracts	
	sales price		Oil Price	b)	Other Price index	
PEMV	Netback ethanol	OPIS Chicago	Information Service	c)	Shipments distribution in a highly volatile market	
	sales price		(OPIS)	d)	Other accounting adjustments items	
PES	Netback ethanol sales price	OPIS LA CI-90	_			
CORN	•					
ALL	Corn CBOT equivalent	Delivered basis = FOB Midwest + freight	Trade West report	a) b)	Fixed futures as a result of hedge Contracts based on corn futures other than nearby futures (Usually, this happens a couple of weeks before current future contract expires Uneven distribution of corn consumptions (such as shut down) in a highly volatile market	
WDG/Syrup						
ALL	Plant co-product	WDG Value as a % of	FC Stone Co-product	a)	Fixed price contracts	
	netback	Plant delivered corn on	report	b)	Hedge activities	
		DDG equivalent basis		c)	WDG freight	

ETHANOL

	Plant Sales Netback		Basis (netback)	Benchmark	Actual vs. Benchmark
PECOL		OPIS PNW			
Jan-11	2.42	2.42	(0.04)	2.38	1.4%
Feb-11	2.45	2.50	(0.04)	2.46	-0.2%
Mar-11	2.60	2.63	(0.04)	2.59	0.5%
Apr-11	2.73	2.76	(0.04)	2.72	0.2%
PEMV		OPIS Chicago			
Jan-11	2.38	2.32	0.01	2.33	2.1%
Feb-11	2.47	2.43	0.01	2.44	1.5%
Mar-11	2.54	2.50	0.01	2.51	1.2%
Apr-11	2.65	2.65	0.01	2.66	-0.2%
PES		OPISLA CI-90			
Jan-11	2.47	2.46	(0.01)	2.45	1.0%
Feb-11	2.57	2.53	(0.01)	2.52	2.1%
Mar-11	2.68	2.66	(0.01)	2.65	0.9%
Apr-11	2.80	2.80	(0.01)	2.79	0.1%
_					

CORN

	Plant CBOT Equivalent	Plant basis	Plant Actual Corn dlvd \$/bu	Market	Actual vs. Market
PECOL					
Jan-11	6.27	0.563	6.83	7.18	(0.35)
Feb-11	6.56	0.595	7.15	7.74	(0.59)
Mar-11	6.68	0.519	7.20	7.64	(0.44)
Apr-11	7.22	0.496	7.71	8.35	(0.64)
PEMV					
Jan-11	6.34	0.650	6.99	7.04	(0.06)
Feb-11	6.87	0.650	7.52	7.60	(0.08)
Mar-11	6.90	0.563	7.46	7.50	(0.04)
Apr-11	7.38	0.639	8.02	8.20	(0.18)
PES					
Jan-11	6.22	0.656	6.88	7.18	(0.30)
Feb-11	6.76	0.663	7.42	7.74	(0.32)
Mar-11	6.74	0.619	7.36	7.65	(0.28)
Apr-11	7.21	0.687	7.90	8.36	(0.46)

CO PRODUCTS

	Plant Co-product \$/ton Netback	Plant Co-product DM	Plant Actual Corn dlvd \$/bu	Plant Actual Corn dlvd \$/ton	WDG Value % of Corn (1)
PECOL	·		·	·	
Jan-11	56.62	30%	6.83	243.91	69.2%
Feb-11	56.85	30%	7.15	255.37	66.4%
Mar-11	60.80	31%	7.20	257.06	68.0%
Apr-11	61.97	31%	7.71	275.44	65.3%
PEMV					
Jan-11	61.90	32%	6.99	249.48	70.5%
Feb-11	67.04	32%	7.52	268.48	71.0%
Mar-11	69.53	32%	7.46	266.45	74.3%
Apr-11	71.74	32%	8.02	286.40	70.5%
PES					
Jan-11	59.21	33%	6.88	245.68	65.4%
Feb-11	59.94	33%	7.42	264.98	61.4%
Mar-11	67.17	33%	7.36	262.95	70.3%
Apr-11	71.47	32%	7.90	282.00	70.5%

⁽¹⁾DDG equivalent, adjusted to standard 90% dry matter